Twisted Stable Homotopy Theory

by

Christopher Lee Douglas

Bachelor of Science, Massachusetts Institute of Technology, June 1999 Master of Science, Oxford University, June 2001

Submitted to the Department of Mathematics in partial fulfillment of the requirements for the degree of

Doctor of Philosophy

at the

ILIN 2 8 2005

ARCHIVES

MASSACHUSETTS INSTITUTE OF TECHNOLOGY LIBRARIES

June 2005

© Christopher L. Douglas, 2005. All rights reserved.

The author hereby grants to MIT permission to reproduce and distribute publicly paper and electronic copies of this thesis document in whole or in part.

Author	····· le · · · · · · · · · · · · · · · ·	
	Department of Mathematics	
	April 29,72005	
Certified by	6	
	Michael JXHopkins	
	Professor of Mathematics	
	Thesis Supervisor	
	1	
Accepted by	·····	
	Pavel Etingof	
Chairman, Department Com	imittee on Graduate Students	

Twisted Stable Homotopy Theory

by Christopher Lee Douglas

Submitted to the Department of Mathematics on April 29, 2005, in partial fulfillment of the requirements for the degree of Doctor of Philosophy

Abstract

There are two natural interpretations of a twist of stable homotopy theory. The first interpretation of a twist is as a nontrivial bundle whose fibre is the stable homotopy category. This kind of radical global twist forms the basis for twisted parametrized stable homotopy theory, which is introduced and explored in Part I of this thesis. The second interpretation of a twist is as a nontrivial bundle whose fibre is a particular element in the stable homotopy category. This milder notion of twisting leads to twisted generalized homology and cohomology and is central to the well established field of parametrized stable homotopy theory. Part II of this thesis concerns a computational problem in parametrized stable homotopy, namely the determination of the twisted K-homology of the simple Lie groups. In more detail, the contents of the two parts of the thesis are as follows.

Part I: I describe a general framework for twisted forms of parametrized stable homotopy theory. An ordinary parametrized spectrum over a space X is a map from X into the category *Spec* of spectra; in other words, it is a section of the trivial *Spec*bundle over X. A twisted parametrized spectrum over X is a section of an arbitrary bundle whose fibre is the category of spectra. I present various ways of characterizing and classifying these twisted parametrized spectra in terms of invertible sheaves and local systems of categories of spectra. I then define homotopy-theoretic invariants of twisted parametrized spectra and describe a spectral sequence for computing these invariants. In a more geometric vein, I show how a polarized infinite-dimensional manifold gives rise to a twisted form of parametrized stable homotopy, and I discuss how this association should be realized explicitly in terms of semi-infinitely indexed spectra. This connection with polarized manifolds provides a foundation for applications of twisted parametrized stable homotopy to problems in symplectic Floer and Seiberg-Witten-Floer homotopy theory.

Part II: I prove that the twisted K-homology of a simply connected simple Lie group G of rank n is an exterior algebra on n-1 generators tensor a cyclic group. I give a detailed description of the order of this cyclic group in terms of the dimensions of irreducible representations of G and show that the congruences determining this cyclic order lift along the twisted index map to relations in the twisted $Spin^c$ bordism group of G.

Thesis Supervisor: Michael J. Hopkins Title: Professor of Mathematics

Acknowledgments

I feel blessed to have had such inspiring and caring teachers and family and friends, and I hope that they do not need to read it in my thesis acknowledgments to know how grateful I am and how much I have treasured their presence in my life.

I would not be a mathematician today if not for Mike Hopkins. I met him during the spring of my junior year of college—after a disillusioning freshman year I had taken two years off from mathematics and I was on my way toward graduate work in philosophy when a departmental requirement landed me in Mike's seminar on the fundamental group. I found Mike to be a model of what it is possible to be as a mathematician: he radiated the vibrancy and creativity and humanity of mathematics at its best. I was taken and reinvigorated and I thought again about whether mathematics might fit into who I wanted to be. In the years since, Mike has been a generous source of mathematical guidance, and his gripping insights and perspective have molded every aspect of my understanding of topology. He has been an inspiration to me, and I will be forever grateful for his energy and his brilliance, for his humor and care and concern.

My graduate school years were also happily and productively shaped by the instruction, guidance, advice, and ideas of Tom Mrowka, Haynes Miller, Graeme Segal, and Ulrike Tillmann, for which I am most grateful. Along the way my spirits were buoyed by many vibrant, encouraging, and informative conversations with Bill Dwyer, Stephan Stolz, Ciprian Manolescu, Ralph Cohen, and Jacob Lurie. I am especially thankful to my friends and colleagues at MIT, Andre Henriques, Elizabeth Mann, Mike Hill, Mark Behrens, Tyler Lawson, Daniel Biss, and Max Lieblich, for their curiosity, their support, their willingness to shoot the breeze on dreary and sunny afternoons alike. Special thanks go to Marco Gualtieri for ever engaging causeries (too often by email) and for making me laugh more and with less restraint than anyone else ever has. To Teena Gerhardt I am enormously grateful for her love and support along the way, even at moments I didn't deserve it, for keeping me from working too hard and for focusing my work when it needed it, for keeping me from starving (in most delectable style) and for teaching me, a little, how to cook for myself, but most of all for being ever close by, as family.

My parents, Michael Hanau and Richard Douglas, deserve most of the credit here: they were, quite simply, exceptional parents—they were energetic and engaged, loving and attentive, trusting enough to give me a freedom and sense of self-determination that few children ever experience; they were models of curiosity and care and of a perfect melding of doing what you love and bettering the world at the same time they were my first teachers and friends and I will always consider it an honor that they are the heart of my family. For my grandparents I am also grateful: Richard Hanau instilled in me a practical inquisitiveness that keeps me alive to the excitement of, well, everything, from gravitational waves to the design of a flue; I learned from Laia Hanau the pleasures of debate and something ineffable about the interlacing of history and society and religion and class; she passed on to me crucial ideas about clarity and structure in writing and speaking, and what academic success I've had owes much to that influence; Judy Douglas has been a beacon of cheer and lightheartedness and a source of bursting and infectious optimism and energy; my love of intricate puzzles and family games traces straight back to hers. David Douglas' staggering, inspired devotions are matched only by his humane and measured wisdom—he renews my happiness and faith, my vivid longing and unrestrained affection for the world.

I have had so many teachers who inspired me, who piqued my curiosity, who were exemplars of what it might mean to verge on the life of the mind: to John Stilgoe, Stanley Cavell, Peter Sacks, Melissa Barry; John Talbot, John Carlevale, Ethan Hoblitzelle; Lee Perlman, Travis Merritt, Ross Lippert; Michael Freedman, David Greenberg, Sandra Bennett, and Audrey Newton I am singularly grateful.

Contents

Ι	Τv	visted Parametrized Stable Homotopy Theory	14
1	Intr	roduction	15
2	Inv	ertible Sheaves of Categories of Spectra	17
	2.1	The Structure Stack of Parametrized Spectra	18
	2.2	Modules over the Structure Stack	23
		2.2.1 Haunts	23
		2.2.2 Specters	27
3	Loc	al Systems of Categories of Spectra	35
	3.1	The Homotopy Theory of Homotopy Theories	36
	3.2	Diagrams of Simplicial Categories	38
		3.2.1 Haunts	38
		3.2.2 Specters	41
	3.3	A_∞ Thom Spectra on Loop Spaces $\hfill\hfilt$	46
		3.3.1 Automorphisms of the Category of Spectra	47
		3.3.2 Specters as Module Spectra	50
4	Inv	ariants of Specters	53
	4.1	<i>R</i> -Haunts and <i>R</i> -Specters	54
	4.2	<i>R</i> -Homology of Specters	57
5	Pol	arized Hilbert Manifolds and Semi-Infinite Spectra	65
	5.1	The Homotopy Theory of Polarized Bundles	66

		5.1.1	Polarizations of Hilbert Space	66
		5.1.2	Symplectic Polarizations and Haunts	71
		5.1.3	Unitary Polarizations and Specter Invariants	73
	5.2	Semi-I	nfinitely Indexed Spectra	75
II	0	n the	Twisted K-Homology of Simple Lie Groups	80
6	Intr	oducti	on	81
	6.1	Six Int	erpretations of Twisted K-Theory	81
		6.1.1	1-Dimensional Elements in Elliptic Cohomology	81
		6.1.2	Projective Hilbert Space Bundles	83
		6.1.3	$K(\mathbb{Z}, 2)$ -Equivariant K-Theory	83
		6.1.4	Perfect Complexes of α -Twisted Sheaves	84
		6.1.5	Central Extensions of Loop Groups	84
		6.1.6	B-Fields and D-Branes	85
	6.2	Result	S	85
	6.3	Techni	ques and Overview	87
7	Twi	sted K	-Theory and the Rothenberg-Steenrod Spectral Sequence	89
	7.1	Twiste	d Homology Theories	89
	7.2	The T	wisted Rothenberg-Steenrod Spectral Sequence	93
8	Tate	e Resol	lutions and $\operatorname{Tor}^{K,\Omega G}(\mathbb{Z},\mathbb{Z}_{\tau})$ for $G \neq Spin(n)$	97
	8.1	Tor for	SU(n+1) and $Sp(n)$	98
	8.2	Tor for	r the Exceptional Groups	99
	8.3	Proof	of Theorem 6.2.1	105
9	Gen	erating	g Varieties, the Cyclic Order of $K^{\tau}G$, and $\operatorname{Tor}^{K.\Omega Spin(n)}(\mathbb{Z},\mathbb{Z}_{\tau})$	107
	9.1	Genera	ating Varieties and Holomorphic Induction	108
		9.1.1	Bott Generating Varieties	108
		9.1.2	The Twisting Map via Holomorphic Induction	111

	9.2	Subvarieties of $\Omega SU(n+1)$ and ΩG_2	114
	9.3	Generating Varieties for $\Omega Sp(n)$	117
	9.4	The Tor Calculation for $Spin(2n+1)$	121
	9.5	Poincare-Dual Bases and the Cyclic Order of $K^{\tau}G$	126
10	Twi	sted $Spin^c$ Bordism and the Twisted Index	131
10	Twi 10.1	sted Spin ^c Bordism and the Twisted IndexA Cocycle Model for Twisted Spin ^c Bordism	131 132
10	Twi 10.1 10.2	sted Spin ^c Bordism and the Twisted Index A Cocycle Model for Twisted Spin ^c Bordism Twisted Nullbordism and the Geometry of the Cyclic Order	131 132 134

List of Tables

2.1	The analogy with basic algebraic geometry	19
4.1	The homotopy groups of the classifying spaces for haunts \ldots .	60
5.1	The homotopy types of the classifying spaces for polarizations	71

List of Figures

2-1	A bundle over the circle with fibre the category of spectra	25
2-2	A twisted parametrized spectrum over the circle	28
9-1	Generating Coweights for Rank 2 Lie Groups	109
9-2	The Weight Lattice of G_2	128

Part I

Twisted Parametrized Stable Homotopy Theory

Chapter 1

Introduction

Despite its widespread use and compelling application to problems in symplectic topology and gauge theory, Floer homology remains rather a mystery. The very existence of the Floer homology of an infinite-dimensional manifold depends on delicate and haphazard properties of the manifold and of the flow associated to a Floer function; moreover, it is completely unknown how the Floer homology depends on the choice of Floer function. Confronted with this situation, Cohen, Jones, and Segal [8] asked whether it would be possible to build a Floer homotopy type encoding the relevant data from the manifold and the function in such a way that a homology functor would recover Floer homology. Besides elucidating the structure of Floer theory and clarifying its dependence on the Floer function, such a Floer homotopy type would immediately provide other invariants such as Floer K-theory and Floer bordism. Cohen, Jones, and Segal suggested that prospectra might encode some of the Floer data; though this thought proved useful, in retrospect it is clear that prospectra can only account for the Floer homotopy types of trivially polarized manifolds-this restriction on the polarization accounts for the difficulty Cohen, Jones, and Segal had finding examples of Floer prospectra.

The purpose of this paper is not to answer the Floer homotopy question, but to introduce a framework, namely twisted parametrized stable homotopy theory, that is a necessarily component of any description of Floer homotopy. That some twisted form of homotopy theory was needed to account for nontrivial polarizations was first realized by Furuta [21], and it will turn out that the twisted space he wrote down (as a conjectural model for the Seiberg-Witten-Floer homotopy type of T^3) is a very specialized example of our twisted parametrized spectra. A twisted parametrized spectrum is a section of a bundle whose fibre is the category of spectra, and as such it has the same relationship to an ordinary parametrized spectrum that a section of a line bundle has to a function. A polarized infinite-dimensional manifold has a naturally associated bundle of categories of spectra, and the fundamental ansatz is that geometric information about such a manifold (its Floer homotopy type, for example) involves this bundle and its sections. In this paper, we present the theory of twisted parametrized spectra, including various definitions, characterizations, and classifications of them, a thorough description of their homotopy-theoretic invariants, and an overview of their relationship to infinite-dimensional polarized manifolds. The specific association of a twisted parametrized homotopy type to a manifold with Floer function is the subject of ongoing work with Mike Hopkins and Ciprian Manolescu and will appear elsewhere [9, 10].

I would like to thank especially Mike Hopkins for insightful and inspiring questions and indispensable pointers, Bill Dwyer for fruitful suggestions and encouraging words, and Jacob Lurie for technical help and much headache-saving advice.

Chapter 2

Invertible Sheaves of Categories of Spectra

Let X be an algebraic variety over a ring R. The structure sheaf \mathcal{O}_X can be described as the sheaf of R-valued functions on X. The most fundamental \mathcal{O}_X -modules are the invertible or locally free rank-one modules. These modules, which we will often think of as line bundles, are classified by the first cohomology group of X with coefficients in the units of R; denote by $\mathcal{L}(c)$ the line bundle associated to the cohomology class $c \in H^1(X; \mathbb{R}^{\times})$. A global section of $\mathcal{L}(c)$ is determined by a 0-cochain on X whose coboundary is c, that is by an element $f \in C^0(X; \mathbb{R})$ such that $\delta f = c$.

Twisted parametrized stable homotopy theory is a precise analog of these algebraic concepts: the ring R is replaced by the category Sp of spectra, a "categorical semi-ring" under the wedge and smash products. The set of Sp-valued functions on a space X is naturally interpreted as the category of parametrized spectra on X, and the structure "sheaf" is therefore the structure stack \mathcal{O}_X of parametrized spectra. There is a notion of locally free rank-one module over the structure stack of parametrized spectra and we refer to such modules, briefly, as *haunts*. Haunts are classified by the first cohomology group of X with coefficients in the so-called Picard category $\operatorname{Pic}(S^0)$ of units of Sp. The fundamental objects of twisted parametrized stable homotopy theory are the global sections of a haunt; these global sections are the twisted parametrized spectra or *specters* for short. Thus, a specter has the same relationship to a parametrized spectrum as a section of a line bundle has to a function. Moreover, a specter for the haunt $\mathcal{L}(c)$ associated to a class $c \in H^1(X; \operatorname{Pic}(S^0))$ is determined by a 0-cochain with coboundary c, that is by an element $f \in C^0(X; Sp)$ together with an identification $\delta f \cong c$.

This fundamental analogy is summarized in table 2.1 and is explained in detail in the following sections.

2.1 The Structure Stack of Parametrized Spectra

We begin by describing the category of parametrized spectra and its associated homotopy theory. A spectrum E is, most naively, a series E_i of based spaces equipped with structure maps $\Sigma E_i \to E_{i+1}$ from the suspension of one space to the next. Similarly, we can describe a parametrized spectrum over X by giving a series E_i of based spaces over X together with structure maps $\Sigma_X E_i \to E_{i+1}$ from the fibrewise suspension of one space to the next. (A based space over X is a space together with a projection map to X and a section of this projection.) This naive viewpoint is sufficient for many purposes, including taking the homology and cohomology of a base space X with coefficients in a parametrized spectrum, but it fails to provide a foundation for a good smash product on the category of parametrized spectra. As we are interested in considering this category to be a semi-ring, it is essential that we have a highly associative and commutative smash product. We therefore work with the category of orthogonal parametrized spectra. An orthogonal parametrized spectrum on X is a diagram spectrum in the category of based spaces over X, where the diagram category is finite dimensional inner product spaces and their isometries. See Mandell, et al. [34] for a description of diagram spectra and May-Sigurdsson [37] for an extensive discussion of orthogonal parametrized spectra. In keeping with our sheaf-theoretic philosophy, will we let $\mathcal{O}_X(U)$ denote the category of (orthogonal) parametrized spectra on the open set $U \subset X$.

We have selected a model for parametrized spectra because we need to ensure that we have a well behaved smash product, but we are not of course interested

	In Algebraic Geometry	In Twisted Parametrized Stable Homotopy Theory
The basic ring R is	an ordinary ring	the category Sp of spectra
The structure sheaf \mathcal{O}_X is given by	$\mathcal{O}_X(U) = R$ -valued functions on U	$\mathcal{O}_X(U) = $ the category of parametrized spectra on U
A line bundle ${\cal L}$ is	a sheaf that is an invertible \mathcal{O}_X -module	a stack that is a locally free rank-one \mathcal{O}_X -module
The set of units in the ring R is	the multiplicative group R^{x}	the monoidal category $\operatorname{Pic}(S^0)$ of invertible S^0 -modules
Line bundles $\mathcal{L}(c)$ are classified by	elements $c \in H^1(X; R^{\times})$	elements $c \in H^1(X; \operatorname{Pic}(S^0))$ or maps $X \to B \operatorname{Pic}(S^0)$
A global section of a line bundle is	an ordinary section	a twisted parametrized spectrum
Such a section is presented by	an element $f \in C^0(X; R)$ such that $\delta f = c$	an element $f \in C^0(X; Sp)$ together with $\delta f \cong c$

Table 2.1: The analogy with basic algebraic geometry

in the point-set-level peculiarities of this particular model. As such, we need to keep in mind a homotopy theory, that is a model structure or at least a notion of weak equivalences, on the category $\mathcal{O}_X(U)$ of parametrized spectra; we should consider a category \mathcal{C} equivalent to $\mathcal{O}_X(U)$ not if there is an equivalence of categories between them, but if there is a functor inducing an equivalence of homotopy theories. We will focus on the homotopy theory on $\mathcal{O}_X(U)$ associated to the stable model structure defined in [37]. Suffice it to say that the fibrant objects in the stable model structure are, in particular, quasi-fibrations of spectra over X, and the weak equivalences between fibrant objects are maps that induce weak equivalences on each fibre. We can think of this as a homotopy theory of quasi-fibrations of spectra, rather than of all parametrized spectra; this reduction will be important in our consideration of local systems of categories of spectra in section 3. As an aside, we note that there may be other interesting model structures on parametrized spectra, for example ones in which there is a much larger class of fibrant objects; the formulation of twisted parametrized spectra in section 2.2 will work for these alternate model structures, producing a very different and perhaps even more intriguing theory.

Remark 2.1.1. We have fixed a notion of homotopy theory on the category of parametrized spectra, namely the one coming from the stable model structure. The concept of an ∞ -category conveniently encodes the notion of a category together with an associated homotopy theory. An ∞ -category is, roughly speaking, a category together with 2-morphisms, 3-morphisms, and so on, such that all the *n*-morphisms are invertible for n > 1. Though ∞ -categories are as yet little utilized, many familiar structures, including simplicial categories, Segal categories, Segal spaces, and quasi-categories, give models for ∞ -categories; see Lurie [31, 32] for a thorough treatment. The pairing of a category and a notion of homotopy theory will be so pervasive that in this section and in section 2.2 we will frequently use "category" to mean " ∞ -category" and implicitly take associated notions, such as equivalence, monoidal structure, module, and so forth, to refer to their ∞ -categorical analogs. The reader who is bothered by the resulting inexplicitness should defer to section 3 where haunts and specters are recharacterized in more traditional terms. (Model-theoretically inclined readers may

want to take "category" to mean "model category" and this will be perfectly suitable by way of understanding, but we will be utilizing categories of categories (read ∞ -categories of ∞ -categories) and the category of model categories is not known to have a model structure.)

 $\sim \sim \sim \sim \sim$

The association to an open set $U \subset X$ of the category $\mathcal{O}_X(U)$ of parametrized spectra over U is meant to function as a "sheaf of rings" analogous to the structure sheaf of R-valued functions on an algebraic variety. First we consider the ring-like structure on $\mathcal{O}_X(U)$ and then proceed to the sheaf- or stack-like properties of \mathcal{O}_X .

The category $\mathcal{O}_X(U)$ of parametrized spectra is a symmetric bimonoidal category in the sense of Laplaza [28]; that is, it comes equipped with two symmetric monoidal functors (wedge and smash) and natural distributivity isomorphisms satisfying various coherence relations. In fact, the category is better behaved that the average symmetric bimonoidal category because the wedge product is the categorical coproduct; the additive associativity isomorphisms and the distributivity isomorphisms are therefore canonically defined. Of course, there is a rigidification functor [18, 12] that replaces a symmetric bimonoidal category with an equivalent bipermutative category (where the associativity isomorphisms are identity transformations). We will frequently and implicitly use the bipermutative category associated to $\mathcal{O}_X(U)$, particularly when discussing modules in the next section. This symmetric bimonoidal or bipermutative structure makes $\mathcal{O}_X(U)$, for all intents and purposes, into a semi-ring.

Philosophically, a stack C is a presheaf of categories satisfying descent up to equivalence of categories; (see, for example, Moerdijk's treatment in [40]). That is, the category C(U) living over a large open set U is determined, up to equivalence, by the categories C(V) living over small open subsets $V \subset U$, in the same way that the value of a sheaf is determined by its local behavior. Note that these categories C(U) need not be groupoids (as is usually assumed) and that both "presheaf" and "equivalence" can be freely interpreted. For example, "presheaf" might mean literal presheaf, presheaf up to coherent natural isomorphism, or presheaf up to coherent natural homotopy equivalence; similarly, "equivalence of categories" might mean ordinary equivalence of categories, Quillen equivalence of model categories, homotopical equivalence of ∞ -categories, or something analogous. In general, we take presheaf to mean presheaf up to coherent natural isomorphism, and equivalence of categories to mean homotopical equivalence of ∞ -categories.

As the notation suggests, we have restriction functors $i^* : \mathcal{O}_X(U) \to \mathcal{O}_X(V)$ associated to inclusions $i : V \subset U$. Because these restrictions of parametrized spectra boil down to literal restriction functors in categories of topological spaces over the base X, these functors give \mathcal{O}_X the structure of a presheaf of categories on X. This presheaf \mathcal{O}_X is a stack. Though it is a stack in the usual, literal sense that it satisfies descent up to equivalence of categories (as can be checked using the prestack gluing condition given in [40, p.11]), we are only concerned with the fact that it is a stack in the sense that it satisfies descent up to homotopical equivalence of categories.

Summary 2.1.2. The association \mathcal{O}_X to an open subset $U \subset X$ of the category $\mathcal{O}_X(U)$ of orthogonal parametrized spectra on U is a stack of symmetric bimonoidal $(\infty$ -)categories.

2.2 Modules over the Structure Stack

We have replaced an ordinary ring R by the semi-ring category Sp of spectra and we are investigating invertible sheaves in this new context. We have introduced our basic "sheaf of rings" \mathcal{O}_X , namely the structure stack of parametrized spectra. In this section, we describe and classify locally free rank-one modules over this structure stack—we refer to these modules, briefly, as "haunts"—and we study their categories of global sections. These global sections are the fundamental objects of twisted parametrized stable homotopy theory and we call them "twisted parametrized spectra" or "specters" for short.

2.2.1 Haunts

A module over a stack \mathcal{R} of symmetric bimonoidal categories is a stack \mathcal{M} of symmetric monoidal categories together with an action $\mathcal{R} \times \mathcal{M} \to \mathcal{M}$ appropriately compatible (by analogy with a module over a ring) with the monoidal structures. We do not spell out this compatibility; see Dunn [12] for an extensive discussion of modules over semi-ring categories, Lurie [32] for some of the technicalities involved in semi-ring ∞ -categories and their modules, and remark 2.2.2 below for an explanation of why we do not attend to the details of these compatibility relations. Such a module \mathcal{M} over a symmetric bimonoidal stack \mathcal{R} is locally free of rank one if for all points $x \in X$ there exists an open set $U \subset X$ containing x and an object $S \in \mathcal{M}(U)$ such that the map $\mathcal{R}|_U \to \mathcal{M}|_U$ determined for $V \subset U$ by

$$\mathcal{R}(V) \to \mathcal{M}(V)$$

 $A \mapsto A \cdot S|_V$

is an equivalence of symmetric monoidal stacks.

Definition 2.2.1. A haunt on a space X is a locally free rank-one module over the structure stack \mathcal{O}_X of parametrized spectra on X.

Remark 2.2.2. Because the additive monoidal structure in the category \mathcal{O}_X of parametrized spectra is given by the categorical coproduct, most of the compatibility conditions [12] for \mathcal{O}_X -modules are automatically satisfied, provided the monoidal structure on the module is also the coproduct. Indeed, it is generally sufficient to treat \mathcal{O}_X as a multiplicative monoid and study stacks with an action of this monoid. As a point of philosophy, though, it is important to keep in mind that we are really dealing with modules over *ring* stacks.

Remark 2.2.3. We limit our attention to locally free rank-one modules over parametrized spectra, but we imagine that there may be quite interesting and intricate homotopy-theoretic information in the structure of higher rank modules.

In doing geometry over an ordinary ring R, we think of invertible sheaves as

line bundles. Such a line bundle is most easily and explicitly described by taking trivial *R*-bundles over an open cover $\{U_i\}$ and specifying appropriate gluing data $r_{ij}: R \times U_{ij} \to R \times U_{ij}$ on the two-fold intersections $U_{ij} = U_i \cap U_j$. By analogy, we think of haunts as bundles whose fibre is the category Sp of spectra. To specify such a bundle, we can take a trivial Sp-bundle over an open cover and give gluing data $r_{ij}: Sp \times U_{ij} \to Sp \times U_{ij}$ on the intersections; more precisely, this gluing data amounts to automorphisms $r_{ij}: \mathcal{O}_X(U_{ij}) \to \mathcal{O}_X(U_{ij})$. Before formalizing this viewpoint, we give two examples.

Example 2.2.4. Cover the base space $X = S^1$ by two open semicircles U_0 and U_1 and denote by V and W the two components of the intersection U_{01} . Glue $\mathcal{O}_X|_{U_0}$ and $\mathcal{O}_X|_{U_1}$ together along V by the identity map on $\mathcal{O}_X(V)$ and along W by the map

$$\mathcal{O}_X(W) \to \mathcal{O}_X(W)$$

 $T \mapsto T \wedge_W (\mathbf{S}^n \times W)$

In other words, the monodromy around the circle is the map $Sp \to Sp$ given by suspension by S^n . Schematically the resulting haunt appears as in figure 2-1. As we will see, every haunt over S^1 is equivalent to this suspension haunt for some integer n.

Figure 2-1: A bundle over the circle with fibre the category of spectra



Example 2.2.5. Now take the base space X to be S³ with its usual hemispherical cover by two open sets D_0^3 and D_1^3 ; the intersection of these open sets is the equatorial band S² × ($-\epsilon, \epsilon$). Let S²×S² denote the nontrivial S² bundle over S². We can define a

haunt over X by gluing the trivial bundles $Sp \times D_0^3$ and $Sp \times D_1^3$ as follows:

$$\mathcal{O}_X(\mathrm{S}^2 \times (-\epsilon, \epsilon)) \to \mathcal{O}_X(\mathrm{S}^2 \times (-\epsilon, \epsilon))$$
$$T \mapsto T \wedge_{\mathrm{S}^2 \times (-\epsilon, \epsilon)} ((\mathrm{S}^2 \rtimes \mathrm{S}^2) \times (-\epsilon, \epsilon))$$

This gluing produces the only nontrivial haunt over S^3 .

Line bundles are built by gluing together trivial *R*-bundles over intersections. At a point, this gluing is determined by an *R*-module automorphism of the ring *R*, that is by an element of $\operatorname{Aut}_R R$. Of course, any such automorphism is multiplication by a unit of *R*; in other words, $\operatorname{Aut}_R R \cong R^{\times}$. The gluing data for a line bundle is therefore a 1-cocycle with values in R^{\times} or more precisely, with values in the sheaf \mathcal{O}_X^{\times} of R^{\times} -valued functions. Up to equivalence, these bundles are classified by the cohomology group $H^1(X; \mathcal{O}_X^{\times})$.

Appropriately interpreted, all these facts remain true when R is replaced by the category Sp of spectra—see table 2.1. Any automorphism of Sp as a module over itself is given by smashing with an invertible spectrum; moreover, there is an equivalence of categories $\operatorname{Aut}_{Sp} Sp \cong Sp^{\times}$. Here the objects of Sp^{\times} are the invertible spectra and the morphisms are weak equivalences of spectra. This category is denoted $\operatorname{Pic}(S^0)$ in the literature and we will use that notation to refer to both the category and its realization. The gluing data for a haunt is a 1-cocycle with values in $\operatorname{Pic}(S^0)$, by which we mean a cocycle with values in the associated sheaf $\mathcal{P}ic(S^0)$ of " $\operatorname{Pic}(S^0)$ -valued functions". This "sheaf" is a stack of monoidal categories; the category $\mathcal{P}ic(S^0)(U)$ has objects invertible parametrized spectra on U and morphisms weak equivalences. The classification is as expected:

Proposition 2.2.6. Haunts on X are classified up to equivalence by the cohomology group $H^1(X; \mathcal{P}ic(S^0))$ or equivalently by the group of homotopy classes of maps $[X, B \operatorname{Pic}(S^0)].$

Remark 2.2.7. The reader who (sensibly enough) grimaces at the prospect of cohomology with coefficients in a stack of monoidal categories can happily focus on the latter characterization in terms of homotopy classes of maps, which will be established in detail in section 3.2.1.

Any invertible spectrum is weakly equivalent to a sphere of some integer dimension, so $B \operatorname{Pic}(S^0)$ has the homotopy type of $B(\mathbb{Z} \times BGL_1(S^0))$ where $GL_1(S^0)$ denotes the module automorphisms of the sphere spectrum, also known as the group of stable self equivalences of the sphere. In particular, $\pi_1(B \operatorname{Pic}(S^0)) = \mathbb{Z}$ and $\pi_3(B \operatorname{Pic}(S^0)) = \mathbb{Z}/2$, explaining the classifications mentioned in examples 2.2.4 and 2.2.5. We will discuss the homotopy groups of $B \operatorname{Pic}(S^0)$ in more detail in the context of specter invariants in section 4.

2.2.2 Specters

As previously mentioned, specters generalize parametrized spectra in the same way that sections of line bundles generalize functions:

Definition 2.2.8. A twisted parametrized spectrum or specter on X is a global section of a haunt over X. That is, it is an object $S \in \mathcal{M}(X)$ of the category of global sections of a locally free rank-one module \mathcal{M} over the structure stack \mathcal{O}_X of parametrized spectra on X.

An ordinary *R*-valued function on *X* is determined by a 0-cocycle with values in *R*, that is by a 0-cochain $f \in C^0(X; R) := C^0(X; \mathcal{O}_X)$ such that the coboundary vanishes: $\delta f = 0$. Suppose $c \in Z^1(X; R^{\times}) := Z^1(X; \mathcal{O}_X^{\times})$ is a 1-cocycle defining a line bundle L(c) with fibre *R*. A section of L(c) is presented by a 0-cochain $f \in C^0(X; \mathcal{O}_X)$ cobounding the cocycle *c*, which is to say such that $\delta f = c$. This section need not trivialize the line bundle because it is allowed to take non-invertible values, unlike the defining cocycle for the bundle.

Analogously, a parametrized spectrum on X can be described by a 0-cocycle with values in the stack of parametrized spectra. This amounts to giving a 0-cochain $f \in C^0(X; Sp) := C^0(X; \mathcal{O}_X)$, namely a parametrized spectrum on each open set of a cover, together with a compatible system of equivalences on intersections; this system of equivalences is concisely encoded in the equation $\delta f \cong 0$. Let $c \in Z^1(X; Sp^{\times}) :=$ $Z^1(X; \mathcal{P}ic(S^0))$ denote a 1-cocycle defining a haunt L(c); concretely, this means that for a cover $\{U_i\}$ we have invertible parametrized spectra $c_{ij} \in \mathcal{P}ic(S^0)(U_{ij})$ on two-fold intersections together with fixed weak equivalences $\phi_{ijk} : c_{ij} \wedge_{U_{ijk}} c_{jk} \to c_{ik}$ satisfying the obvious coherence relation on four-fold intersections. A specter for this haunt, that is a section of L(c), is most easily presented by a 0-cochain $f \in C^0(X; \mathcal{O}_X)$ together with an identification $\delta f \cong c$ of the coboundary of f with the cocycle c. What this means is that on two-fold intersections compatible equivalences are given between $f_i \wedge_{U_{ij}} c_{ij}$ and f_j . This cochain presentation is well suited to giving explicit examples of specters.

Example 2.2.9. Let L_n denote the haunt over S^1 whose monodromy is suspension by S^n . This haunt is depicted in figure 2-1 and can be presented, roughly speaking, as follows: take two copies of the stack of parametrized spectra on an interval, that is of \mathcal{O}_{D^1} , and glue them together at the two pairs of endpoints by the maps $Sp \xrightarrow{\text{id}} Sp$ and $Sp \xrightarrow{\wedge S^n} Sp$ respectively. We now define a specter T for L_n . Over the first interval D_0^1 the specter is a trivial parametrized spectrum with fibre S^n ; that is $T|_{D_0^1} = S^n \times D_0^1$. Over the second interval D_1^1 the specter is a cone on $S^n \sqcup S^0$; that is, writing $D_1^1 = C(* \sqcup *)$, we have $T|_{D_1^1} = C(S^n \sqcup S^0)$. See figure 2-2.

Figure 2-2: A twisted parametrized spectrum over the circle



Example 2.2.10. Let L_{S^3} denote the nontrivial haunt over S^3 described in example 2.2.5. Roughly speaking, L_{S^3} is constructed using the equatorial gluing function $\psi : \mathcal{O}_{S^2} \to \mathcal{O}_{S^2}$ given by $\psi(P) = P \wedge_{S^2} (S^2 \rtimes S^2)$. Define a specter T for L_{S^3} by $T = (S^0 \times D^3) \cup_{\psi} C(S^2 \rtimes S^2)$; that is, on one hemisphere the specter is a trivial

parametrized spectrum with fibre S^0 and on the other hemisphere (thought of as the cone $C(S^2)$) the specter is the cone on the bundle $S^2 \rtimes S^2$.

Remark 2.2.11. The procedure (illustrated in these examples) of piecing together parametrized spaces to give a global geometric object has also appeared in a preprint by Furuta [21] under the rubric "prespectra with parametrized universe". Furuta's viewpoint is similar to ours in spirit, but his 'parametrized prespectra' are substantially more rigid than specters; in particular, few specters can be realized as 'parametrized prespectra'. It would therefore seem to be difficult to develop a reasonable homotopy theory of twisted spectra (which is essential for applications to Floer homotopy) using the rigid geometry of 'parametrized prespectra'.

In section 2.1 we emphasized the fact that the homotopy theory of parametrized spectra should be conceived as a homotopy theory of fibrations of spectra. The above examples of specters have singularities and are therefore not locally fibrations of spectra. There is a fibrant replacement functor which takes such a singular specter and returns a locally fibrant specter with the same "global homotopy type"; (see section 4 for a discussion of global homotopy invariants of specters). On the one hand, it is easier to explicitly describe and compute invariants of singular specters; on the other hand, it is easier to characterize and classify specters using fibrant presentations.

The following model for fibrant replacement of specters bears close resemblance to the one May and Sigurdsson use for parametrized spectra; a more thorough technical treatment of the replacement functor can be found in their manuscript [37]. Suppose S is a specter over the base X. Let PX denote the path fibration on X and s and t the source and target maps $PX \to X$. A fibrant model F(S) for the specter S is very roughly given by $s_!(t^*(S))$ —here $s_!$ denotes integration over the fibre in a sense analogous to that given for parametrized spectra in [37]. In other words, the points of the fibre $F(S)_x$ at a point $x \in X$ are pairs consisting of a path in X from x to yand a point of the fibre S_y . We proceed to some examples of fibrant specters.

Example 2.2.12. As before, let L_n denote the haunt over S^1 whose monodromy is suspension by S^n . Define a specter T for L_n as follows: the fibre T_x of T at every

point $x \in S^1$ is $\bigvee_{i \in \mathbb{Z}} S^{n \cdot i}$ and the monodromy operator is the natural equivalence $\Sigma^n(\bigvee S^{n \cdot i}) \simeq \bigvee S^{n \cdot i}$. This is a model for the specter in example 2.2.9 which is locally a fibration of spectra. More generally, any specter for the haunt L_n on S^1 can be described by giving a spectrum A together with an equivalence of A with its n-th suspension:

$$\{ \text{Specters}/(L_n, S^1) \} \iff \{ \text{Spectra } A \text{ with } \phi : \Sigma^n(A) \simeq A \}$$

Example 2.2.13. There is only one nontrivial haunt over S^2 ; call it L_{S^2} . Let U and V denote the two hemispheres of S^2 . The haunt L_{S^2} is constructed, roughly speaking, by gluing \mathcal{O}_U and \mathcal{O}_V along the equatorial S^1 using the function $\psi : \mathcal{O}_{S^1} \to \mathcal{O}_{S^1}$ given by $\psi(P) = P \wedge_{S^1} (S^0 \rtimes S^1)$; here $S^0 \rtimes S^1$ denotes the nontrivial S^0 bundle over S^1 . Suppose we want to construct a specter T for L_{S^2} that is locally a fibration of spectra. The restriction $T|_U$ of the specter to one hemisphere U will be equivalent to the parametrized spectrum $A \times U$, for some spectrum A. On the boundary of U, the specter is $A \times \partial U = A \times S^1$, and therefore on the boundary of V the specter must be $\psi(A \times S^1) = (A \times S^1) \wedge_{S^1} (S^0 \rtimes S^1)$ —let us denote this last parametrized spectrum by $\mathbf{A}^{\mathrm{tw}(-1)}$. As $\mathbf{A}^{\mathrm{tw}(-1)}$ is a fibrant parametrized spectrum over S^1 with fibre A, it is constructed by gluing together two copies of $A \times D^1$ along the boundaries $A \times S^0$. Such a gluing is determined by a map $S^0 \to \operatorname{Aut}(A)$, where $\operatorname{Aut}(A)$ denotes the homotopy automorphisms of the spectrum A. In the case of $\mathbf{A}^{\mathrm{tw}(-1)}$ this gluing is the map $-1_A: S^0 \to \operatorname{Aut}(A)$ taking one point to id_A and the other point to $-\operatorname{id}_A$. The specter T restricts on the hemisphere V to a fibrant parametrized spectrum $T|_V$ with boundary $\mathbf{A}^{\mathrm{tw}(-1)}$ —this parametrized spectrum defines a nullhomotopy of the gluing map -1_A of $\mathbf{A}^{\mathrm{tw}(-1)}$. In summary:

{Specters/ L_{S^2} } \longleftrightarrow {Spectra A with $\phi: D^1 \to \operatorname{Aut}(A)$ s.t. $\phi_0 = \operatorname{id}_A, \phi_1 = -\operatorname{id}_A$ }

Example 2.2.14. Recall the nontrivial haunt L_{S^3} over S^3 determined by the equatorial gluing function $\psi(P) = P \wedge_{S^2} (S^2 \rtimes S^2)$. Note that this haunt is isomorphic to the haunt determined by the gluing function $\psi'(P) = P \wedge_{S^2} (S^0 \rtimes S^2)$, where $S^0 \rtimes S^2$ is the unique nontrivial S^0 -spectrum bundle over S^2 . Suppose T is a specter for L_{S^3} that is locally a fibration of spectra. Then, as in the previous example, T restricted to one hemisphere U is a trivial bundle with fibre spectrum A. The boundary $T|_{\partial U}$ of this restriction is $A \times \partial U = A \times S^2$ and the boundary $T|_{\partial V}$ of the other restriction must therefore be $\psi'(A \times S^2) = (A \times S^2) \wedge_{S^2} (S^0 \rtimes S^2)$ —denote this last parametrized spectrum by $\mathbf{A}^{\mathrm{tw}(\eta)}$. The fibrant parametrized spectrum $\mathbf{A}^{\mathrm{tw}(\eta)}$ over S^2 is determined by the gluing function $\eta_A : S^1 \to \mathrm{Aut}(A)$; here η_A is the function $\eta \wedge A$ where $\eta : S^1 \to$ $\mathrm{Aut}(S^0)$ is the nontrivial element of the first stable stem. The restriction $T|_V$ of the specter to the second hemisphere constitutes a nullhomotopy of the gluing function η_A for the boundary parametrized spectrum $\mathbf{A}^{\mathrm{tw}(\eta)}$. Again we have a classification:

$$\{\text{Specters}/L_{S^3}\} \iff \{\text{Spectra } A \text{ with } \phi: D^2 \to \text{Aut}(A) \text{ s.t. } \phi_{S^1} = \eta_A \}$$

 $\sim \sim \sim \sim \sim$

We have seen a variety of examples of specters and have described all the specters associated to a few particular haunts. We now systematically investigate the equivalence classification of specters, giving a homotopy-theoretic description of specters that naturally parallels the characterization of haunts in proposition 2.2.6.

Ordinary line bundles with flat connection on X are classified by maps $X \to BR^{\times}$. There is a universal R^{\times} -bundle ER^{\times} over BR^{\times} and an associated R-bundle $P(R) := ER^{\times} \times_{R^{\times}} R$. A section s of the line bundle L(c) associated to a map $c : X \to BR^{\times}$ is determined by a lift of c to a map $s : X \to P(R)$:

$$\begin{array}{c} R \longrightarrow P(R) \\ s \nearrow^{\not{}} & \downarrow \\ X \xrightarrow{} BR^{\times} \end{array}$$

We describe the parametrized stable homotopy analog of this description. Haunts (the line bundles over the structure stack) on X are classified by maps $X \to B \operatorname{Pic}(S^0)$; as before $\operatorname{Pic}(S^0)$ denotes the realization of the category of invertible spectra. Let Sp_w denote the realization of the subcategory of weak equivalences of the category

of spectra, and note that $\operatorname{Pic}(S^0)$ acts on Sp_w . There is a universal $\operatorname{Pic}(S^0)$ -bundle $E\operatorname{Pic}(S^0)$ over $B\operatorname{Pic}(S^0)$ and an associated Sp_w -bundle $P(Sp_w) := E\operatorname{Pic}(S^0) \times_{\operatorname{Pic}(S^0)} Sp_w$. We can consider lifts of the classifying map $c: X \to B\operatorname{Pic}(S^0)$ to $P(Sp_w)$:



Indeed, such lifts classify specters:

Proposition 2.2.15. Let $c: X \to B \operatorname{Pic}(S^0)$ be the classifying map for a haunt L(c). Weak equivalence classes of specters for the haunt L(c) are in one-to-one correspondence with homotopy classes of lifts of c to maps $X \to P(Sp_w)$.

That there should be such a homotopy-theoretic classification of specters was suggested to us by Bill Dwyer. A proof of a slightly stronger result will appear in section 3.2.2.

We conclude our discussion of haunts and specters with a few remarks about products. Given two ordinary invertible sheaves (line bundles) L and L' over X we can form their tensor product $L \otimes_{\mathcal{O}_X} L'$. Analogously, given two haunts we should be able to form their tensor product. Making sense of tensoring two modules over a stack of semi-ring categories would require a bit of doing; (see Dunn [12] for a definition of the tensor product of modules over a bipermutative category). Using the classification of haunts in proposition 2.2.6 we can side step this categorical tensor construction: define the product $L \otimes L'$ of two haunts to be the haunt classified by the product (in the group structure on $B \operatorname{Pic}(S^0)$) of the classifying maps $c, c' : X \to B \operatorname{Pic}(S^0)$. There is a product of specters covering this tensor product of haunts. The classifying projection $P(Sp_w) \to B \operatorname{Pic}(S^0)$ for specters is a map of multiplicative monoids. Given two specters T and T' classified by lifts $s, s' : X \to P(Sp_w)$ of the haunt maps $c, c' : X \to B \operatorname{Pic}(S^0)$, we simply define the product $T \wedge T'$ to be the specter (for the haunt $L \otimes L'$) classified by the product lift $s \cdot s' : X \to P(Sp_w)$.

Chapter 3

Local Systems of Categories of Spectra

In section 2, we described haunts as invertible sheaves, or more specifically as locally free rank-one modules over the structure stack of parametrized spectra; specters, the twisted parametrized spectra, were global sections of these modules. In this section, we leave behind that sheaf-theoretic approach and reformulate haunts as local systems of categories of spectra. Such a local system is determined by assigning to each open set in a cover a category equivalent to the category of ordinary, not parametrized, spectra and to the two-fold intersections in the cover compatible equivalences of these categories of spectra.

In order to formalize this local systems viewpoint, we need a clear notion of equivalence between two categories of spectra, which is to say we need a "category of categories of spectra"—such a category is typically referred to as a homotopy theory of homotopy theories. We chose to work with the model category of simplicial categories, which contains in particular the subcategory of simplicial categories weakly equivalent to the simplicial localization of the category of spectra. When the base space B has the structure of a simplicial complex, and the open sets of the cover of B are the stars of the simplicies, then the data of a local system amounts to a functor from the diagram category of simplicies of B into (a subcategory of weak equivalences in) the category of simplicial categories. This diagram functor approach has the disadvantage

that it clouds the conceptual simplicity and the geometry of haunts and specters, but it has the advantage that it avoids the technicalities of stacks of ∞ -categories and thereby eases the proofs of propositions 2.2.6 and 2.2.15.

After proving those propositions, we give *another* reformulation of haunts and specters: haunts are precisely A_{∞} ring spectra arising as Thom spectra of multiplicative stable spherical fibrations on loop spaces, and the associated specters are simply modules over the ring spectrum. This last description has the advantage of being entirely elementary—it avoids both modules over stacks of parametrized spectra and diagrams in the model category of simplicial categories—but it thoroughly obscures various constructions with and applications of specters, and so in section 4 we return to our original sheaf-theoretic perspective.

3.1 The Homotopy Theory of Homotopy Theories

As it will play a central role in our discussion of local systems in section 3.2, we recall what is known about the homotopy theory of homotopy theories. In recent decades, model categories have been the predominant notion of abstract homotopy theory. However, there is not known to be a model structure on the category of model categories, and this is a huge impediment to constructing (as we are doing in this paper) bundles of homotopy theories. We must consider a weaker notion of abstract homotopy theory in order to have a decent homotopy theory of homotopy theories. There are various options, including the model category of Segal categories (due to Hirschowitz-Simpson [23]), the model category of complete Segal spaces (due to Rezk [42]), and the model category of simplicial categories (due to Dwyer-Hirschhorn-Kan [13] and Bergner [4]). Which we pick does not matter because all three model category of simplicial categories are Quillen equivalent (a result due to Bergner [4]); we work with the model category of simplicial categories, as this is the simplest to describe.

By a 'simplicial category' we will mean a category enriched over simplicial sets. Dwyer and Kan [15] realized that to a model category M there is canonically associated a simplicial category $L^H M$, the hammock localization of M, which encodes all of the homotopy-theoretic information contained in M. This is the sense in which simplicial categories are a faithful representation of abstract homotopy theories. In particular, the homotopy category $\operatorname{Ho}(M)$ of M is recovered as the category of components $\pi_0(L^H M)$ of the hammock localization; here the category of components $\pi_0(C)$ of a simplicial category C (also called the homotopy category of C) has the same objects as C, but has morphisms $\operatorname{Hom}_{\pi_0(C)}(x, y) = \pi_0(\operatorname{Hom}_C(x, y))$. A morphism $f: x \to y$ in a simplicial category C is called a homotopy equivalence if it becomes an isomorphism in $\pi_0(C)$.

The model structure on the category of simplicial categories is as follows. A map $\phi: C \to D$ of simplicial categories is a weak equivalence if it is a Dwyer-Kan equivalence, namely if ϕ is a weak equivalence on Hom sets and an equivalence on homotopy categories; that is, ϕ is a weak equivalence if $\phi: \operatorname{Hom}_C(x, y) \to \operatorname{Hom}_D(\phi(x), \phi(y))$ is a weak equivalence of simplicial sets for all objects $x, y \in C$, and if $\phi: \pi_0(C) \to \pi_0(D)$ is an equivalence of categories. A map $\phi: C \to D$ is a fibration if it is a fibration on Hom sets and if all homotopy equivalences in D lift to C; that is, ϕ is a fibration if $\phi: \operatorname{Hom}_C(x, y) \to \operatorname{Hom}_D(\phi(x), \phi(y))$ is a fibration of simplicial sets for all $x, y \in C$, and if for all objects $x \in C$ and all homotopy equivalences $h: \phi(x) \to z$ in D, there exists a homotopy equivalence $\tilde{h}: x \to y$ in C such that $\phi(\tilde{h}) = h$. Cofibrations of simplicial categories are determined, as usual, by the left lifting property. These define a model structure on the category of simplicial categories [4]. Given an object in a model category, there is a good notion of the space of automorphisms of that object, and in the following we will be focused on the automorphisms of the category of spectra (thought of as a simplicial category via its hammock localization).

Remark 3.1.1. Considering that our objection to model categories as a representation of homotopy theory was that there is no obvious model category of model categories, it seems odd to have insisted on having a model category of simplicial categories rather than merely a simplicial category of simplicial categories. Of course, we can recover a simplicial category of simplicial categories as the hammock localization of the model category of simplicial categories, but it would be better not to have to rely on the crutch of a model structure. The real solution to this and many other problems is to work directly in the ∞ -category of ∞ -categories. We do not do this because the details of such an ∞ -categorical theory are not yet fully in place; see Lurie [31, 32], though, for substantial progress in that direction.

3.2 Diagrams of Simplicial Categories

We described haunts as locally free rank-one modules over the structure stack of parametrized spectra. We now reinterpret this notion in terms of diagrams of simplicial categories weakly equivalent to the hammock localization of the category of spectra. We then express the category of specters for such a haunt as a homotopy limit in the model category of simplicial categories.

3.2.1 Haunts

Let H be a haunt on the space X. By definition, if U is a sufficiently small open set in X, then the category H(U) is homotopy equivalent to the category $\mathcal{O}_X(U)$ of parametrized spectra on U. If the subspace U is moreover contractible, then $\mathcal{O}_X(U)$ is homotopy (indeed Quillen) equivalent to the category Sp of spectra. (Roughly speaking, this follows because a fibrant parametrized spectrum is a quasifibration and any quasifibration over a contractible space is trivializable—see [37].) Let $\{U_i\}$ denote a contractible cover of X; to specify a haunt on X it will be sufficient, by the above remarks, to assign to the subsets U_i categories C_i that are each appropriately equivalent to the category of spectra and to the two-fold intersections $\{U_{ij}\}$ a collection of compatible equivalences $C_i \cong C_j$. We formalize these compatibilities using diagram functors into the category of simplicial categories.

First we fix some notation. The space X is homotopy equivalent to the realization of a simplicial set B; let s(B) denote the category of simplicies of B, that is the category whose objects are the simplicies of B and whose morphisms are the face and degeneracy maps. Let sCat denote the model category of simplicial categories, and let Sp now denote the object of sCat given by the hammock localization of the category of spectra. Let w(sCat, Sp) denote the weak equivalence component of sCat
containing Sp; that is, the objects of w(sCat, Sp) are simplicial categories that can be connected to Sp by a zig-zag of weak equivalences, and the morphisms are weak equivalences of simplicial categories.

Remark 3.2.1. We pause to consider a few set-theoretic issues. The category sCat is really the category of small simplicial categories. The hammock localization of the category of spectra is not only not small, it need not even have small Hom sets; we therefore chose a small simplicial category homotopically equivalent to that hammock localization—by the notation Sp we will implicitly refer to that small replacement. In a similar vein, the weak equivalence component of Sp in sCat is not a small category; we will need to use it in constructions that only apply to small categories, so we chose a small subcategory of this weak equivalence component that is homotopically equivalent to the full component—we implicitly refer to that small replacement by the notation w(sCat, Sp). We will not henceforth distinguish between such (simplicial) categories and their small replacements.

Recharacterization 3.2.2. A haunt over a simplicial set B is a functor from s(B), the category of simplicies of B, to w(sCat, Sp), the weak equivalence component of the category of simplicial categories containing the category of spectra. The category of haunts over B, denoted Haunt_B, is the full diagram category $w(sCat, Sp)^{s(B)}$.

We immediately have a notion of the space of haunts, namely the realization $|N. \text{Haunt}_B|$ of the nerve of this diagram category. Note that we will not in general distinguish between simplicial sets and their realizations. There is a natural candidate for a classifying space for haunts, namely N.w(sCat, Sp). The idea that a weak equivalence component of an object of a model category can function as a classifying complex is of course due to Dwyer and Kan [16]. Indeed, there is a suggestive homotopy equivalence $N.w(sCat, Sp) \simeq B$ haut(Sp). Here B haut(Sp) is the nerve of the simplicial category with one object and with morphisms the simplicial monoid haut(Sp) of homotopy automorphisms of the category of spectra; (this simplicial monoid is defined to be the sub-simplicial monoid of $\text{Hom}_{L^H(sCat)}(Sp, Sp)$). The

classification of haunts can therefore be expressed as follows.

Theorem 3.2.3. The space N. Haunt_B of haunts over a simplicial set B is weakly homotopy equivalent to the (derived) mapping space Hom(B, B haut(Sp)). In other words, B haut(Sp) is a classifying space for haunts.

Remark 3.2.4. We adopt the convention that all mapping spaces are implicitly derived unless otherwise noted. We will also take "holim" and "hocolim" to refer to the homotopically invariant homotopy limit and colimit functors; these are sometimes referred to as the corrected homotopy limit and colimit and can be defined respectively by composing functorial objectwise fibrant or cofibrant replacement with the Bousfield-Kan holim or hocolim functor.

Before proving the theorem, we state one lemma:

Lemma 3.2.5. Suppose M is a model category that is Quillen equivalent to a cofibrantly generated simplicial model category. Let B be a simplicial set and let s(B) denote the category of simplicies of B. For any object X of M there is a weak homotopy equivalence

$$N.(w(M,X)^{s(B)}) \simeq \underset{s(B)}{\operatorname{holim}} N.w(M,X)$$

Dwyer and Kan prove this for M equal to the category of simplicial sets [16, Thm 3.4], but their proof works for any cofibrantly generated simplicial model category. Moreover, both sides of the equivalence are weakly homotopy invariant under Quillen equivalence between not-necessarily-simplicial model categories; this follows using various results from [14, 15].

Proof of Theorem 3.2.3. We have the chain of equivalences:

$$N. \operatorname{Haunt}_{B} \equiv N.(w(sCat, Sp)^{s(B)}) \simeq \underset{s(B)}{\operatorname{holim}} N.w(sCat, Sp)$$
$$\simeq \operatorname{Hom}(N.s(B), N.w(sCat, Sp))$$
$$\simeq \operatorname{Hom}(B, B \operatorname{haut}(Sp))$$

The category of simplicial categories is Quillen equivalent to the category of complete Segal spaces [4] which is a cofibrantly generated simplicial model category [42]; the first equivalence therefore follows from the above lemma. The second equivalence is a consequence of [22, Prop 18.2.6], and the third is immediate.

3.2.2 Specters

We now discuss specters in this new context of diagrams of simplicial categories. In the sheaf-theoretic framework of section 2 a specter was a global section of a haunt. A haunt is now a functor H from the category of simplicies s(B) into the category sCat of simplicial categories. Naturally enough, a "section" of such a diagram H of simplicial categories should be some appropriately consistent choice of objects $\{x_b \in H(b)\}_{b \in s(B)}$ of the simplicial categories H(b) in the diagram—we can think of an object $x_b \in H(b)$ as a locally constant section of the haunt H restricted to the simplex b. It would be too much to ask that the collection of objects x_b be strictly compatible with the morphisms in the diagram H. Instead, we merely demand that there be chosen homotopies to "glue the objects together"—this gluing data is formally encoded in a homotopy limit.

Recharacterization 3.2.6. Let H be a haunt over B, that is a functor from the category of simplicies s(B) to the category of simplicial categories that lands in the weak equivalence component w(sCat, Sp) of the category of spectra. The (simplicial) category of specters for the haunt H, denoted Specter_H, is defined to be the homotopy limit holim H.

Of course, this homotopy limit is only defined because we have a model structure on the category of simplicial categories, and as usual we mean the homotopically invariant homotopy limit. We also have an associated *space of specters* for the haunt H, namely $N.w(\text{Specter}_H)$. Here w(C) denotes the sub-simplicial category of the simplicial category C whose objects are the same as those of C but whose morphisms $\text{Hom}_{w(C)}(a, b)$ are the components of $\text{Hom}_C(a, b)$ projecting to isomorphisms in $\text{Hom}_{\pi_0(C)}(a, b)$; note that N.w(C) is, a priori, a bisimplicial set and we implicitly take its diagonal.

There isn't a classifying space for specters, per se, but there is a classifying fibration; that is, there is a fibration $\psi : Uhaut(Sp) \to Bhaut(Sp)$ such that the space of specters for a fixed haunt $H : B \to Bhaut(Sp)$ is homotopy equivalent to the space of lifts of H along ψ . The fibre of ψ should be the "space of spectra" and we think of the total space Uhaut(Sp) as the "universal haunt". In fact, this classifying fibration comes from a fibration of simplicial categories, which is defined as the diagonal map in the following diagram:



That is, we factor the left hand map by a weak equivalence followed by a fibration Ψ ; this fibration is the desired specter classifying fibration of simplicial categories. Here $\operatorname{hocolim}(-)$ refers to the homotopy colimit of the inclusion $w(sCat, Sp) \to sCat$.

The analogous specter classifying fibration of spaces is the right hand vertical arrow in the diagram

The fibration factorization here defines the space $U \operatorname{haut}(Sp)$. Note that the fibre of ψ over the point $X \in w(sCat, Sp)$ is weakly equivalent to N.w(X) which is in turn weakly equivalent to the space of spectra N.w(Sp). Morally speaking, the bundle $U\operatorname{haut}(Sp)$ is the bundle $E \operatorname{haut}(Sp) \times_{\operatorname{haut}(Sp)} N.w(Sp)$ associated to the tautological bundle $E \operatorname{haut}(Sp) \to B \operatorname{haut}(Sp)$. However, on its face the action of the simplicial monoid $\operatorname{haut}(Sp)$ on N.w(Sp) is only defined up to weak homotopy, which is insufficient for defining the associated bundle; presumably the action can be made strict,

but we do not pursue that here.

We now state the simplicial-category-level classification result for specters:

Theorem 3.2.7. Let $H : s(B) \to w(sCat, Sp)$ be a functor defining a fixed haunt on the simplicial set B. This functor determines an associated map $h : \operatorname{hocolim} * \to \underset{w(sCat,Sp)}{\operatorname{hocolim}} *$ classifying the haunt. The category of specters $\operatorname{Specter}_{H}$ for this haunt is weakly equivalent, as a simplicial category, to the category of lifts of h along the specter classifying fibration:



In other words the category of specters is weakly equivalent to a (derived) mapping space in the overcategory of $\underset{w(sCat,Sp)}{\text{hocolim}}$ *, namely

$$\operatorname{Specter}_{H} \simeq \operatorname{Hom}_{\operatorname{bcolim}_{w(sCat,Sp)}} * \left(\operatorname{hocolim}_{s(B)} *, \operatorname{hocolim}_{w(sCat,Sp)}(-) \right)$$

Proof. The first step in the proof is a consequence of the following general lemma:

Lemma 3.2.8. Let M be either an ∞ -topos (such as simplicial sets or spaces) or a model category of homotopy theories (such as simplicial categories, complete Segal spaces, or quasi-categories). Let D be a small category and $F : D \to M$ a functor. Denote by ϕ : hocolim $F \to \text{hocolim} *$ the natural projection. Provided F takes all morphisms in D to weak equivalences in M, the homotopy limit of F is weakly equivalent to the object of derived sections of the map ϕ ; that is

$$\underset{D}{\operatorname{holim}} F \simeq \operatorname{Hom}_{\operatorname{hocolim}} * \left(\underset{D}{\operatorname{hocolim}} *, \underset{D}{\operatorname{hocolim}} F \right)$$

In the case of specters, we therefore have the equivalence

$$\operatorname{Specter}_{H} \equiv \operatorname{holim}_{s(B)} H \simeq \operatorname{Hom}_{\operatorname{hocolim} *} \left(\operatorname{hocolim}_{s(B)} *, \operatorname{hocolim}_{s(B)} H \right)$$

We want to translate this mapping space into a space of lifts of the classifying map $h: \operatornamewithlimits{hocolim}_{s(B)}* \to \operatornamewithlimits{hocolim}_{w(sCat,Sp)}*.$ We begin by rewriting one of the homotopy colimits in terms of a larger indexing category:

$$\operatorname{hocolim}_{s(B)} H \simeq h^* \left(\operatorname{hocolim}_{w(sCat, Sp)}(-) \right)$$

Here h^* denotes the derived pullback from the overcategory in sCat of hocolim * to w(sCat,Sp)the overcategory of hocolim *. Next, we have a Quillen adjunction s(B)

$$sCat/\left(\operatorname{hocolim}_{s(B)}^{*}\right) \xrightarrow{h^{*}}_{h_{!}} sCat/\left(\operatorname{hocolim}_{w(sCat,Sp)}^{*}\right)$$

where the pushforward h_{l} is given by precomposition with the map h. This adjunction leads to an equivalence of function complexes

$$\operatorname{Hom}_{\operatorname{hocolim} *} \left(\operatorname{hocolim}_{s(B)} *, h^* \left(\operatorname{hocolim}_{w(sCat, Sp)}(-) \right) \right) \simeq \operatorname{Hom}_{\operatorname{hocolim}_{w(sCat, Sp)} *} \left(\operatorname{hocolim}_{s(B)} *, \operatorname{hocolim}_{w(sCat, Sp)}(-) \right)$$
as desired.

as desired.

Not surprisingly, the analogous result at the level of spaces is the following.

Corollary 3.2.9. Let $h : B \to B \operatorname{haut}(Sp)$ denote the classifying map for a fixed haunt H over the simplicial set B. The associated space of specters N.w Specter_H is weakly homotopy equivalent to the space $\operatorname{Hom}_{B \operatorname{haut}(Sp)}(B, U \operatorname{haut}(Sp))$ of maps from B to the universal haunt U haut (Sp) that commute with projection to B haut (Sp). In other words, the space of specters for the haunt H is weakly equivalent to the space of lifts in the diagram

$$Uhaut(Sp) \longleftarrow N.w(Sp)$$

$$B \xrightarrow{\checkmark} B haut(Sp)$$

Proof. The chain of equivalences is

$$N.w \operatorname{Specter}_{H} = N.w \operatorname{holim}_{s(B)} H \simeq N.w \operatorname{Hom}_{\operatorname{hocolim}_{w(sCat,Sp)} *} \left(\operatorname{hocolim}_{s(B)} *, \operatorname{hocolim}_{w(sCat,Sp)}(-) \right)$$
$$= N.w \operatorname{Hom}_{\operatorname{hocolim}_{w(sCat,Sp)} *} \left(\operatorname{hocolim}_{s(B)} *, \overline{\operatorname{hocolim}_{w(sCat,Sp)}(-)} \right)$$
$$\simeq N. \operatorname{Hom}_{\operatorname{hocolim}_{w(sCat,Sp)} *} \left(\operatorname{hocolim}_{s(B)} *, w \left(\overline{\operatorname{hocolim}_{w(sCat,Sp)}(-)} \right) \right)$$
$$\simeq \operatorname{Hom}_{N.w(sCat,Sp)} \left(N.s(B), N.w \left(\overline{\operatorname{hocolim}_{w(sCat,Sp)}(-)} \right) \right)$$
$$\simeq \operatorname{Hom}_{B \operatorname{haut}(Sp)}(B, U \operatorname{haut}(Sp))$$

The first homotopy equivalence is a consequence of the theorem, and the second line follows from the definition of the Hom set as a derived mapping space. Next note that the functor $w: sCat \rightarrow hGpd$ from simplicial categories to homotopy groupoids is right adjoint to the inclusion—the homotopy equivalence in the third line follows because hocolim * and hocolim * are already homotopy groupoids. The category of s(B) * w(sCat,Sp) * are already homotopy groupoids. The category of homotopy groupoids is in fact Quillen equivalent to the category of simplicial sets; the equivalence in the fourth line follows, and the fifth line is immediate.

3.3 A_{∞} Thom Spectra on Loop Spaces

We begin this section by identifying the monoid haut(Sp) of automorphisms of the category of spectra with a classifying space $\mathbb{Z} \times BG$ for stable spherical fibrations. Using this identification, we can associate to a haunt over X an A_{∞} ring spectrum arising as a Thom spectrum over the loop space of X. This in turn allows us to recharacterize the category of specters for the haunt as a category of modules over that ring spectrum. This recharacterization on the one hand obscures the intrinsic symmetry of specters and breaks the natural connection with parametrized homotopy theory, which is essential to the definition of specter invariants in section 4; on the other hand, because module spectra are familiar objects, the change in perspective demystifies specters and will be important in applications to symplectic Floer homotopy [9].

3.3.1 Automorphisms of the Category of Spectra

Haunts over a space X are classified by maps from X to the space B haut(Sp), a deloop of the simplicial monoid of homotopy automorphisms of the category of spectra. We investigate the homotopy type of this classifying space. The category of spectra has a natural monoidal structure, the smash product; given an invertible spectrum J, the functor $J \wedge -: Sp \to Sp$ smashing with J determines a self homotopy equivalence of the category of spectra. Roughly speaking, this association determines a map from the category of invertible spectra (which we called $\operatorname{Pic}(S^0)$ in section 2) to the space of self equivalences haut(Sp). That this map is a weak equivalence is well known to experts, but we are not aware of a statement or a proof in the literature:

Theorem 3.3.1. Let $Pic(S^0)$, the Picard category, denote the subcategory of the category of spectra whose objects are invertible spectra and whose morphisms are weak equivalences. There is a weak equivalence

$$\operatorname{Pic}(S^0) \simeq \operatorname{haut}(Sp)$$

from the nerve of the Picard category to the simplicial set of self homotopy equivalences of the category Sp of spectra.

Proof. We merely sketch the proof. The category of spectra is a model category representing a particular homotopy theory, and we can work in any of a number of equivalent categories of homotopy theories. We have primarily utilized sCat, the category of simplicial categories, but for this theorem it is more convenient to work in qCat, the category of quasi-categories—there is a Quillen equivalence between sCat and qCat [32]. Recall that a quasi-category is a simplicial set that satisfies a weak Kan condition, namely that a horn $\partial \Delta^n \setminus \Delta_i^{n-1}$ fills in provided the missing face Δ_i^{n-1} is internal, that is 0 < i < n; this weak Kan condition reflects the idea that morphisms

(edges) in a category are composable, but need not be invertible up to homotopy. By qCat we refer, in fact, to the category of simplicial sets equipped with a model structure in which quasicategories are precisely the fibrant objects.

Let Sp denote a quasicategory modeling the category of spectra; we presume that Sp is equipped with a monoidal structure modeling the smash product. In this context $\operatorname{Pic}(S^0)$ is a subquasicategory of Sp which is described as follows. The vertices of $\operatorname{Pic}(S^0)$ are the invertible objects in Sp, that is the vertices $v \in Sp$ such that there exists a $w \in Sp$ with $v \wedge w$ weakly equivalent to $S^0 \in Sp$; the k-simplicies of $\operatorname{Pic}(S^0)$ are the k-simplicies of Sp all of whose vertices are invertible and all of whose edges are weak equivalences. By definition, the simplicial monoid $\operatorname{haut}(Sp)$ has k-simplicies the set of weak equivalences $\Delta^k \times Sp \xrightarrow{\simeq} Sp$. There is now a natural map

$$\mu : \operatorname{Pic}(S^0) \to \operatorname{haut}(Sp)$$

which takes a k-simplex P in $\operatorname{Pic}(S^0)_k$ to the composite $\Delta^k \times Sp \xrightarrow{P \times \operatorname{id}} \operatorname{Pic}(S^0) \times Sp \to Sp \times Sp \to Sp$. This composite is an equivalence and is therefore a k-simplex in haut(Sp).

We would like μ to be an equivalence. It suffices to show that any map F: $(\Delta^k, \partial \Delta^k) \rightarrow (\operatorname{haut}(Sp), \operatorname{Pic}(S^0))$ is homotopic, relative to its boundary, to a map $F': (\Delta^k, \partial \Delta^k) \rightarrow (\operatorname{Pic}(S^0), \operatorname{Pic}(S^0))$. Suppose k = 0, so F is simply an equivalence $Sp \rightarrow Sp$; we take F' to be the map $Sp \rightarrow Sp$ given by smashing with $F(S^0)$, that is, $F' = F(S^0) \in \operatorname{Pic}(S^0)$. One extremely convenient feature of quasicategories (as distinguished from, for example, model categories) is that one can naturally take homotopy colimits over any simplicial set, not only over a category; this feature is helpful in defining a comparison map between F and F'. The quasicategory Sp is in particular a simplicial set, and given an object $X \in Sp$, let (-/X) denote the "overcategory" of X, that is the subsimplicial set of Sp whose 0-simplicies are maps $Y \rightarrow X$. Moreover, denote by $(-/X)_{\rm sph}$ the corresponding "spherical subcategory", that is the full subcategory whose 0-simplicies are the maps $S^i \rightarrow X$. We have the comparison map

$$F(S^0) \wedge X \simeq \underset{(-/X)_{\mathrm{sph}}}{\mathrm{hocolim}} F(S^i) \to \underset{(-/X)}{\mathrm{hocolim}} F(Y) \simeq F(X)$$

This map is an equivalence when $X = S^0$, and the left hand side preserves homotopy colimits. It is a consequence of Lurie's extensive work on quasicategories [32] that because F is an equivalence, it preserves homotopy colimits. The comparison map is therefore a natural weak equivalence, as desired. The cases of higher k could be handled similarly.

Corollary 3.3.2. The simplicial set haut(Sp) of self equivalences of the category of spectra has the homotopy type $\mathbb{Z} \times BG$ where G is the space of stable self homotopy equivalences of the sphere, that is $G = haut(S^0) \simeq colim haut(S^n)$ where in the last expression S^n denotes the ordinary n-sphere.

Proof. Given the theorem, this is a consequence of the weak equivalence $\operatorname{Pic}(S^0) \simeq \mathbb{Z} \times BG$. To see that equivalence, first note that any invertible spectrum is weakly equivalent to some shift S^n of the sphere spectrum. Thus the category $\operatorname{Pic}(S^0)$ has \mathbb{Z} components; the n-th component is all spectra weakly equivalent to S^n together with all weak equivalences between them. By Dwyer and Kan's classification theorem [16], this component has the homotopy type B haut $(S^n) \simeq B$ haut (S^0) . \Box

3.3.2 Specters as Module Spectra

Armed with the identification of the classifying space B haut(Sp) with $B(\mathbb{Z} \times BG)$, we can describe the A_{∞} ring spectrum corresponding to a haunt. Let $h : X \to B$ haut $(Sp) \simeq B(\mathbb{Z} \times BG)$ be the classifying map for a haunt over the space X. The map $\Omega h : \Omega X \to \mathbb{Z} \times BG$ defines a stable spherical fibration, which we will denote $\eta(h)$, over ΩX . Because Ωh is a loop map, the spherical fibration $\eta(h)$ is multiplicative and the associated Thom spectrum $\text{Th}(\eta(h))$ is therefore an A_{∞} ring spectrum [33]. The fibration $\eta(h)$ can be thought of more geometrically as follows. The haunt is a local system or bundle over X whose fibre is the category of spectra, and the monodromy of the haunt around a loop $\ell \in \Omega X$ is an invertible spectrum, namely the fibre $\eta(h)_{\ell}$; the multiplicative structure of the spherical fibration corresponds, naturally enough, to the composition of the loop monodromies.

If the Thom spectrum $\operatorname{Th}(\eta(h))$ encodes the structure of the haunt h, we might expect to be able to describe the associated category of specters in terms of this Thom spectrum. Let \mathbb{T} be a specter for the haunt h and suppose \mathbb{T} is fibrant in the sense that it is locally isomorphic to a quasifibration of spectra. Given a loop $\ell : S^1 \to X$ in X, we can pull \mathbb{T} back to a specter $\ell^*\mathbb{T}$ on S^1 . From example 2.2.12 we know that this specter is determined by giving the spectrum T at the basepoint of S^1 together with an equivalence $\phi_{\ell} : \eta(h)_{\ell} \wedge T \simeq T$; in other words, we need to glue T back to itself, but shifted by the monodromy sphere $\eta(h)_{\ell}$ along the given loop. The family of compatible equivalences $\{\phi_\ell\}_{\ell\in\Omega X}$ amounts precisely to an action of $\operatorname{Th}(\eta(h))$ on T. To a specter for the haunt h we can therefore associate a module over the ring spectrum $\operatorname{Th}(\eta(h))$; indeed there is an equivalence of categories:

Proposition 3.3.3. Let H denote a fixed haunt over X with classifying map $h: X \to B(\mathbb{Z} \times BG)$. The loop map Ωh determines a multiplicative stable spherical fibration $\eta(h)$ with associated Thom spectrum $\operatorname{Th}(\eta(h))$ an A_{∞} ring spectrum. There is a weak equivalence of simplicial categories

$$\operatorname{Specter}_{H} \simeq L^{H}(\operatorname{Th}(\eta(h)) \operatorname{-mod})$$

between the category of specters for H and the hammock localization of the category of module spectra over the Thom spectrum $Th(\eta(h))$.

That specters can be thought of as modules over a ring spectrum was also realized by Mike Hopkins and Jeff Smith, and the formulation here owes various details to discussions with them.

Sketch of proof. We have already done most of the work in establishing, in theo-

rem 3.2.7, that

$$\operatorname{Specter}_{H} \simeq \operatorname{Hom}_{\operatorname{w(sCat,Sp)}} * \left(\operatorname{hocolim}_{s(B)} *, \operatorname{hocolim}_{w(sCat,Sp)}(-) \right)$$

Here *B* is a simplicial set with the homotopy type of *X*. By theorem 3.3.1 the base space $\underset{w(sCat,Sp)}{\operatorname{hocolim}} * \simeq N.w(sCat, Sp)$ has the homotopy type $B\operatorname{Pic}(S^0)$. In particular, we can model this space by the simplicial category, also denoted $B\operatorname{Pic}(S^0)$, with just one object * and with morphism space the (simplicial) monoid $\operatorname{Pic}(S^0)$. Similarly, the total space $\underset{w(sCat,Sp)}{\operatorname{hocolim}}(-)$ of the classifying fibration can be explicitly modeled by a simplicial category, denoted $U\operatorname{Pic}(S^0)$, as follows. The objects of $U\operatorname{Pic}(S^0)$ are ordinary cofibrant and fibrant spectra. The morphism space $\operatorname{Hom}_{U\operatorname{Pic}(S^0)}(T,S)$ of $U\operatorname{Pic}(S^0)$ is the homotopy colimit of the functor $\operatorname{Hom}_{Sp}(-\wedge T, S) : \operatorname{Pic}(S^0)_{cf} \to sSet$. Here $\operatorname{Pic}(S^0)_{cf}$ denotes the category of cofibrant and fibrant invertible spectra. The idea behind this construction is that roughly speaking a morphism from *T* to *S* in $U\operatorname{Pic}(S^0)$ should consist of a pair (γ, ϕ) of an invertible spectrum $\gamma \in \operatorname{Pic}(S^0)_{cf}$, the $\operatorname{Pic}(S^0)_{cf} \to Set \to S$. Because $\operatorname{hocolim} * \simeq \operatorname{Pic}(S^0)_{cf}$, the $\operatorname{Pic}(S^0)_{cf} \to Set \to S$. Because $\operatorname{hocolim} * \cong \operatorname{Pic}(S^0)_{cf}$, the $\operatorname{Pic}(S^0)_{cf} \to Set \to S$. Because $\operatorname{hocolim} * \cong \operatorname{Pic}(S^0)_{cf}$, the $\operatorname{Pic}(S^0)_{cf} \to Set \to S$. The projection $\operatorname{Hom}_{Sp}(-\wedge T, S) \to *$ induces a map $U\operatorname{Pic}(S^0) \to B\operatorname{Pic}(S^0)$. Thinking of $X \simeq \operatorname{hocolim} *$ as a simplicial category, we therefore have the reformulation: s(B)

$$\operatorname{Hom}_{\substack{\operatorname{hocolim}\\w(sCat,Sp)}} * \left(\operatorname{hocolim}_{s(B)} *, \operatorname{hocolim}_{w(sCat,Sp)}(-) \right) \simeq \operatorname{Hom}_{B\operatorname{Pic}(S^0)}(X, U\operatorname{Pic}(S^0))$$

The final equivalence is

$$\operatorname{Hom}_{B\operatorname{Pic}(S^0)}(X, U\operatorname{Pic}(S^0)) \simeq L^H(\operatorname{Th}(\eta(h))\operatorname{-mod})$$

On this count we merely indicate the map from the right to the left hand side. We can model X by the simplicial category with one object and with morphism space ΩX . The map $h : X \to B \operatorname{Pic}(S^0)$ corresponds to the map of morphism spaces $\Omega X \to \operatorname{Pic}(S^0)$ classifying the spherical fibration $\eta(h)$. Given a $\operatorname{Th}(\eta(h))$ module T, one immediately has, for any k-simplex $s : \Delta^k \to \Omega X$, a morphism $\operatorname{Th}(\eta(h)|_s) \wedge T \to T$. By a shift in perspective, the stable spherical bundle $\eta(h)|_s$ on Δ^k naturally corresponds to a k-simplex in the morphism category $\operatorname{Pic}(S^0)$ of $B\operatorname{Pic}(S^0)$, and the map $\operatorname{Th}(\eta(h)|_s) \wedge T \to T$ then provides a lift of this morphism k-simplex to $U\operatorname{Pic}(S^0)$, as desired.

Example 3.3.4. Consider again the haunt L_n over S^1 whose monodromy is suspension by S^n . This haunt is classified by the map $S^1 \to B(\mathbb{Z} \times BG)$ representing $n \in \mathbb{Z} = \pi_1(B(\mathbb{Z} \times BG))$. The loop of this map, $\Omega S^1 \simeq \mathbb{Z} \xrightarrow{n} \mathbb{Z} \to \mathbb{Z} \times BG$, classifies the fibration over \mathbb{Z} whose fibre at i is $S^{n \cdot i}$. The associated Thom spectrum is $\bigvee_{i \in \mathbb{Z}} S^{n \cdot i}$, which we considered as a specter already in example 2.2.12. The set of specters for L_n is, as we saw in that example, the same as the set of module spectra over this distinguished specter $\bigvee_{i \in \mathbb{Z}} S^{n \cdot i}$.

Chapter 4

Invariants of Specters

A parametrized spectrum P over a space X has two naturally associated spectra, namely the total spectrum P/X representing the homotopy type of P and the spectrum of sections $\Gamma(P)$ representing the cohomotopy type of P. The generalized homology groups of these associated spectra provide invariants of the parametrized spectrum. A specter, that is a twisted parametrized spectrum, has no globally defined homotopy or cohomotopy type analogous to the total spectrum P/X or the spectrum of sections $\Gamma(P)$. Nevertheless, it is frequently possible to define invariants associated to a specter by first applying a generalized homology functor and *then* taking an associated global spectrum, rather than vice versa as is typical in parametrized homotopy theory.

We return to the stack-theoretic perspective of section 2. There, we treated the category of spectra as a ring and defined haunts to be locally free rank-one modules over a parametrized version of this ring; specters were global sections of these modules. We begin this section by describing an analogous construction where the basic ring is the category of R-modules for a commutative ring spectrum R. This leads to a notion of R-haunt and R-specter. We then see how to associate to a haunt H an R-haunt H_R and to a specter T for H an R-specter T_R for H_R ; this base change is the aforementioned "generalized homology functor". When the R-haunt H_R is trivializable, the R-specter T_R has the form of a parametrized R-module and therefore has a global homotopy type T_R/X . The homotopy groups of T_R/X are the R-homology

invariants of the original specter T. We describe a few examples of these specter invariants and discuss a spectral sequence for computing them.

4.1 *R*-Haunts and *R*-Specters

The category of spectra, or equivalently the category of modules over the sphere spectrum S^0 , has a smash product which, roughly speaking, gives it the structure of a commutative ring. The stack of parametrized spectra on a space X therefore functions as a sheaf of rings, and we characterized haunts as locally free rank-one modules over this stack. There are natural subcategories of the category of spectra that have their own commutative products, and we study modules over the stacks associated to these categorical rings. Specifically, if R is an A_{∞} ring spectrum, then we have the category R-mod of modules over R. If R is moreover E_{∞} , then there is a natural product \wedge_R which, roughly speaking, gives R-mod the structure of a commutative ring; see Dunn [12] for a detailed discussion of the monoidal structures on such module categories. We can associate to an open set $U \subset X$ the category $\mathcal{O}_X^R(U)$ of parametrized *R*-modules on *U*. These parametrized *R*-modules form a stack, which again has a monoidal structure coming from \wedge_R . Morally, an R-haunt is a locally free rank-one module over this structure stack \mathcal{O}_X^R of parametrized Rmodules. The stack \mathcal{O}_X^R is naturally a stack of ∞ -categories, and by a module over this stack we refer to a stack of ∞ -categories with an appropriate action of \mathcal{O}_X^R . A more complete definition of R-haunt is therefore as follows:

Definition 4.1.1. An *R*-haunt is a stack \mathcal{M} of ∞ -categories on a space X with an action of the monoidal stack of ∞ -categories \mathcal{O}_X^R of parametrized *R*-modules satisfying the following condition: the stack \mathcal{M} is locally free of rank one in the sense that for all points $x \in X$ there exists an open set $U \subset X$ containing x and an object $Q \in \mathcal{M}(U)$ such that the map $\mathcal{O}_X^R|_U \to \mathcal{M}|_U$ given by $A \mapsto A \cdot Q$ is an equivalence of stacks of ∞ -categories. A twisted parametrized *R*-module or *R*-specter is a global section of an *R*-haunt.

In order to make this precise, one must give a thorough treatment of stacks of

 ∞ -categories and of monoidal stacks of ∞ -categories; we do not do this, but note that Lurie's work [32] provides key elements of such a treatment. Also note that, as in remark 2.2.2, the additive structure on these stacks is given by the categorical coproduct and so is justifiably ignored—this saves us the horror of contemplating symmetric bimonoidal stacks of ∞ -categories.

An *R*-haunt is locally equivalent to the stack of parametrized *R*-modules and as such can be specified concretely in terms of gluing functions. Suppose $\{U_i\}$ is an open cover of *X*; on an intersection U_{ij} a gluing function is a self equivalence of $\mathcal{O}_X^R|_{U_{ij}}$ as a module over itself. Such an equivalence is given by smashing with an invertible parametrized *R*-module. Let $\operatorname{Pic}(R)$ denote the category of invertible *R*-modules together with their homotopy equivalences, and let $\mathcal{P}ic(R)$ denote the corresponding sheaf of invertible parametrized *R*-modules. The gluing data for an *R*-haunt is therefore a 1-cocycle *c* with values in $\mathcal{P}ic(R)$, which is to say a compatible system c_{ij} of *R*-module gluing functions on the one-fold intersections of the cover. Sensibly enough, *R*-haunts on *X* are classified by homotopy classes of maps from *X* to $B\operatorname{Pic}(R)$.

An *R*-specter looks locally like a parametrized *R*-module, but has a global twist determined by the *R*-haunt. Such a twisted parametrized *R*-module can be presented as follows: to describe an *R*-specter for the *R*-haunt associated to a gluing function c for the cover $\{U_i\}$ it suffices to give a parametrized *R*-module f_i on each open set U_i together with compatible equivalences between $f_i \wedge_{(R,U_{ij})} c_{ij}$ and f_j . Compare section 2.2.2. There is a classification of *R*-specters analogous to that of ordinary specters in proposition 2.2.15, but we do not go into detail.

In section 3.2.1 we formulated haunts not as stacks of ∞ -categories, but as diagrams in the category of ∞ -categories—or more precisely in the category sCat of simplicial categories. The diagram had the homotopy type of the base space X, and the functor to sCat took objects to simplicial categories weakly equivalent to the category of spectra, and morphisms to weak equivalences. A similar approach is

 $\sim \sim \sim \sim \sim$

possible for *R*-haunts but it requires more work. Specifically, let R-mod denote the ∞ -category of *R*-modules; this is an element of the category ∞ -*Cat* of ∞ -categories. Now define (R-mod)-mod to be the subcategory of ∞ -*Cat* of modules over R-mod; in turn w((R-mod)-mod, R-mod) denotes the subcategory of (R-mod)-mod of objects weakly equivalent to R-mod, together with the (R-mod)-module weak equivalences between them. Finally, an *R*-haunt would be a functor from an appropriate diagram homotopy equivalent to X into w((R-mod)-mod, R-mod). The associated ∞ -category of *R*-specters would be the homotopy limit of this functor.

There is also a Thom spectrum approach to R-haunts and R-specters. Suppose an R-haunt H_R is classified by the map $h_R : X \to B \operatorname{Pic}(R)$. The loop of this map $\Omega h_R : \Omega X \to \operatorname{Pic}(R)$ classifies a fibration $\eta(h_R)$ of invertible R-modules over ΩX ; (indeed, the homotopy type of $\operatorname{Pic}(R)$ is $\operatorname{Pic}^0(R) \times BGL_1(R)$ where $\operatorname{Pic}^0(R)$ denotes the equivalence classes of invertible R-modules, and $GL_1(R)$ denotes the R-module self-equivalences of R). This fibration $\eta(h_R)$ is in particular a parametrized spectrum and has an associated total spectrum $\eta(h_R)/X$ which we denote suggestively $\operatorname{Th}(\eta(h_R))$. This total spectrum is an associative R-algebra and it encodes the structure of the R-haunt. An associated R-specter is simply a $\operatorname{Th}(\eta(h_R))$ -module, by which we mean an R-module M together with an appropriately compatible action $\operatorname{Th}(\eta(h_R)) \wedge_R M \to M$.

4.2 *R*-Homology of Specters

We now describe how to associate to haunts and specters respectively R-haunts and Rspecters and we discuss the basic construction of specter invariants. There is a natural
map S⁰-mod \rightarrow R-mod from the category of spectra to the category of R-modules,
given by $T \mapsto T \wedge_{S^0} R$. This is a map of rings and it underlies the fundamental
base-change operation from specters to R-specters. Given a haunt, that is a locally
free rank-one module \mathcal{M} over the stack of parametrized spectra \mathcal{O}_X , we can form the
tensor stack $\mathcal{M} \otimes_{\mathcal{O}_X} \mathcal{O}_X^R$ where \mathcal{O}_X^R is the stack of parametrized R-modules. This
tensor will be an R-haunt. (See Dunn [12] for a definition of tensoring over ring

categories.) Any global section $P \in \mathcal{M}(X)$ of \mathcal{M} transforms to the section $P \otimes R$; we therefore have an *R*-specter associated to any ordinary specter.

In more down-to-earth terms, a haunt is presented by gluing together trivial bundles using invertible parametrized S^0 -modules c_{ij} ; the gluing functions for the associated R-haunt are simply $c_{ij} \wedge R$. Similarly, a specter is locally given by parametrized spectra f_i and the associated R-specter is presented by the parametrized R-modules $f_i \wedge R$. Implicit here is the fact that the map S⁰-mod \rightarrow R-mod restricts to a map $\operatorname{Pic}(S^0) \to \operatorname{Pic}(R)$ of Picard groups and this latter map deloops to a map $B\operatorname{Pic}(S^0) \to$ $B \operatorname{Pic}(R)$ of classifying spaces. Thus we also have the purely homotopy-theoretic characterization of the haunt transformation, namely that the R-haunt associated to the haunt $X \to B \operatorname{Pic}(S^0)$ is classified by the composite $X \to B \operatorname{Pic}(S^0) \to B \operatorname{Pic}(R)$. Remark 4.2.1. The map S^0 -mod \rightarrow R-mod given by smashing with R makes sense for any A_{∞} ring spectrum R. This map moreover induces a map $GL_1(S^0) \to GL_1(R)$ and even a map $BGL_1(S^0) \to BGL_1(R)$. However, in order to build the R-haunt associated to an ordinary haunt, we need moreover a map $B(\mathbb{Z} \times BGL_1(S^0)) \rightarrow B(\mathbb{Z} \times BGL_1(S^0))$ $B(\operatorname{Pic}^{0}(R) \times BGL_{1}(R))$. Even barring the issue of what $\operatorname{Pic}^{0}(R)$ should mean, the space $BGL_1(R)$ cannot deloop unless R is commutative. This provides another indication that in order to have R-homology invariants of specters, R must be a commu-

cation that in order to have A-homology invariants of specters, A must be a commutative ring spectrum. This commutativity requirement on the generalized homology invariants might appear surprising and like a fluke of the formulation, but in fact it reflects an essential aspect of the mathematical structure; later on we will see that the slogan is 'semi-infinite homotopy types only have commutative generalized homology invariants'.

Given a specter P for the haunt H, we have "taken its R-homology" and produced an R-specter P_R for the R-haunt H_R . The essential idea behind specter invariants is that the structure of this R-haunt H_R might be substantially simpler than that of H. In particular, if H_R is trivializable, that is if the composite $X \to B \operatorname{Pic}(S^0) \to$ $B \operatorname{Pic}(R)$ is null homotopic, then any trivialization $\tau : H_R \xrightarrow{\sim} (X \times R \text{-mod})$ transforms the R-specter P_R into a parametrized R-module $\tau(P_R)$. The associated total spectrum $\tau(P_R)/X$ represents a global R-homotopy type for the specter P, even though P does not itself have a global homotopy type. The homotopy groups of $\tau(P_R)/X$ are what we might call the *R*-homology groups of *P*, denoted $R_*^{\tau}(P)$; these are the most straightforward and most easily computable invariants of specters. Note that these groups definitely do depend on the trivialization τ , but this ambiguity can be identified and controlled.

Summary 4.2.2. Let P be a specter for the haunt H and suppose that τ is a trivialization of the associated R-haunt H_R . Then the R-homology groups of the specter P are defined to be the homotopy groups of the total spectrum of the trivialization of the associated R-specter P_R :

$$R_i^{\tau}(P) := \pi_i(\tau(P_R)/X)$$

The potential ambiguity in the trivialization of a trivializable R-haunt H_R is governed by the space of automorphisms of the trivial R-haunt $X \times R$ -mod. More specifically, the space of trivializations of H_R is a torsor for $\operatorname{Aut}(X \times R\text{-mod}) \simeq$ $\operatorname{Hom}(X, \operatorname{Pic}(R))$. Homotopic trivializations τ and τ' determine the same invariants $R_*^{\tau}(P) \simeq R_*^{\tau'}(P)$, and so we need only consider the set of components of the space of trivializations of H_R —this set of components is a torsor for $[X, \operatorname{Pic}(R)] =$ $[X, \operatorname{Pic}^0(R) \times BGL_1R]$. It often happens that $[X, BGL_1R]$ has only one element; if Xis connected, the set of components of the space of trivializations is then a torsor for $\operatorname{Pic}^0(R)$. In this situation, which is to say when we have a specter P for a haunt Hwith H_R trivializable and $[X, \operatorname{Pic}(R)] = \operatorname{Pic}^0(R)$, we can describe how the R-homology of P is affected by a change in trivialization as follows. A given trivialization τ can be modified by an invertible R-module $M \in \operatorname{Pic}^0(R)$ to the trivialization $M \cdot \tau$, and we have

$$R_*^{M \cdot \tau}(P) = \pi_*((M \cdot \tau)(P_R)/X) = \pi_*((\tau(P_R) \wedge_R M)/X) = \pi_*(\tau(P_R)/X \wedge_R M).$$

If the ring spectrum R is such that the homotopy $\pi_*(M)$ of an invertible module M is projective over R_* , then we conclude that $R_*^{M,\tau}(P) = R_*^{\tau}(P) \otimes_{R_*} \pi_*(M)$ —in other words, changing the trivialization shifts the homology of P by the homotopy of an invertible R-module. This shift ambiguity is always present; we think of a homology group determined up to such a shift as uniquely determined and we ignore the τ -dependency, writing simply $R_*(P)$.

 $\sim \sim \sim \sim \sim$

To have homology invariants of a specter P over a haunt H, we need to find a ring spectrum R such that the composite $X \to B \operatorname{Pic}(S^0) \to B \operatorname{Pic}(R)$ is nullhomotopic. As such, we need a thorough understanding of the homotopy types of various $B \operatorname{Pic}(R)$ and of the transformations $B \operatorname{Pic}(S^0) \to B \operatorname{Pic}(R)$. We already noted that $B \operatorname{Pic}(R) \simeq$ $B(\operatorname{Pic}^0(R) \times BGL_1(R))$. By definition, $GL_1(R)$ consists of the unit components of the zero space of the spectrum R, so its zero-th homotopy group is $\pi_0(R)^{\times}$ and its higher homotopy agrees with that of R. Barring the issue of computing $\operatorname{Pic}^0(R)$, which in general is a difficult problem, this allows us to write down the homotopy groups of $B \operatorname{Pic}(R)$ in terms of those of R. These groups are listed for a few common ring spectra in table 4.1.

Classifying Space	Homotopy Groups											
	i = 1	2	3	4	5	6	7	8				
$B\operatorname{Pic}(S^0)$	Z	$\mathbb{Z}/2$	$\mathbb{Z}/2$	$\mathbb{Z}/2$	$\mathbb{Z}/24$	0	0	$\mathbb{Z}/2$				
$B \operatorname{Pic}(\operatorname{H}\mathbb{Z})$	Z	$\mathbb{Z}/2$	0	0	0	0	0	0				
$B\operatorname{Pic}(K)$	$\mathbb{Z}/2$	$\mathbb{Z}/2$	0	\mathbb{Z}	0	\mathbb{Z}	0	\mathbb{Z}				
$B\operatorname{Pic}(MU)$	(\mathbb{Z})	$\mathbb{Z}/2$	0	\mathbb{Z}	0	\mathbb{Z}^2	0	\mathbb{Z}^3				

Table 4.1: The homotopy groups $\pi_i(B\operatorname{Pic}(R))$ of the classifying spaces for R-haunts

The parenthetical group $\pi_1(B \operatorname{Pic}(MU))$ is conjectural. We now describe three simple examples of specters and their potential homology invariants. The first specter has no global homology type, the second has a uniquely determined global homology type, and the third has a global homology type that depends on the choice of trivialization.

Example 4.2.3. Consider the specter described in example 2.2.9: the base space is S^1 , the haunt L_n has monodromy S^n , and the specter T is $S^n \times D^1 \to D^1$ on one semicircle and the cone $C(S^n \sqcup S^0) \to C(* \sqcup *) = D^1$ on the other semicircle. The

classifying map of L_n represents $n \in \mathbb{Z} = \pi_1(B \operatorname{Pic}(S^0))$; in particular there is no global parametrized spectrum corresponding to T and therefore no ordinary homotopy type. Moreover, the map $B(\operatorname{Pic}(S^0)) \to B(\operatorname{Pic}(\mathbb{HZ}))$ is an isomorphism on π_1 , so the homology haunt $(L_n)_{\mathbb{HZ}}$ is still nontrivial; correspondingly, the homology specter $T_{\mathbb{HZ}}$ is not a parametrized \mathbb{HZ} -module and so T does not have homology invariants.

Example 4.2.4. Next consider the specter from example 2.2.10: the base space is S^3 , the haunt L is the haunt determined by the equatorial transition function $- \wedge_{S^2}$ $(S^2 \rtimes S^2)$, and the specter P is $S^0 \times D^3$ on one hemisphere and $C(S^2 \rtimes S^2)$ on the other. On the one hand, the classifying map $S^3 \to B \operatorname{Pic}(S^0)$ for L is nontrivial, representing $1 \in \mathbb{Z}/2 = \pi_3(B \operatorname{Pic}(S^0))$; on the other hand, there are no nontrivial HZ-haunts on S^3 and so L_{HZ} is trivializable. Thus, even though the specter Phas no global homotopy type, it does have homology invariants. Moreover, $\operatorname{Pic}(\mathrm{HZ})$ has homotopy only in degrees 0 and 1, so the set of trivializations of L_{HZ} , namely $[S^3, \operatorname{Pic}(\mathrm{HZ})] \cong \operatorname{Pic}^0(\mathrm{HZ}) = \{\Sigma^n \mathrm{HZ}\}$, is as small as possible. The homology invariants of P are therefore uniquely determined up to degree shift, that is up to tensoring with $\pi_*(\Sigma^n \mathrm{HZ})$.

Let us calculate the homology $H_*(P)$ of the specter P. Roughly speaking, the spectrum HZ cannot see the difference between the transition functions $-\wedge_{S^2}(S^2 \rtimes S^2)$ and $-\wedge_{S^2}(S^2 \times S^2)$. As a result, the parametrized HZ-module P_{HZ} is equivalent to the one obtained by gluing together $\text{HZ} \times D^3$ and $C(\Sigma^2 \text{HZ} \times S^2)$ using the map $-\wedge_{S^2}(S^2 \times S^2)$. By desuspending the second hemisphere, this is in turn equivalent to the parametrized HZ-module $(\text{HZ} \times D^3) \cup_{S^2} (C(\text{HZ} \times S^2))$. This last HZ-module is simply the reduced homology of S^3 ; thus $H_*(P)$ is Z in a single degree and zero in all other degrees. We will formalize this sort of computation in a moment.

Example 4.2.5. Let P be the parametrized spectrum $S^0 \times (S^2 \times S^1)$ over $S^2 \times S^1$. This spectrum defines a specter for the trivial haunt H and as such P has homology invariants. However, there are two distinct trivialization τ_0 and τ_1 of H_{HZ} . They yield respectively the parametrized HZ-modules $\tau_0(P_{\text{HZ}}) = \text{HZ} \times (S^2 \times S^1)$ and $\tau_1(P_{\text{HZ}}) =$ $\text{HZ} \rtimes (S^2 \times S^1)$; this last module exhibits a mobius transformation of the fibre HZ along the S^1 factor of the base. The homology groups of these two trivializations are $H^{\tau_0}_*(P) \cong \{\mathbb{Z} : \mathbb{Z} : \mathbb{Z} : \mathbb{Z} \}$ and $H^{\tau_1}_*(P) \cong \{\mathbb{Z}/2 : 0 : \mathbb{Z}/2 : 0\}$. The difference between these two groups is characteristic of the ambiguity involved in HZ-specters on nonsimply connected base spaces. This dependence on the trivialization indicates that HZ-specter invariants are not naturally graded abelian groups but rather objects in a more subtle algebraic category.

Typically specters are constructed by specifying parametrized spectra over the open sets of a cover; these parametrized spectra do not agree on the intersections but instead are glued together by the transition functions of the haunt. This explicit local presentation suggests a method for computing specter invariants in terms of the local homology invariants of the defining parametrized spectra:

Proposition 4.2.6. Suppose H is a haunt, on a connected space X, whose associated R-haunt H_R admits a trivialization $\tau : H_R \xrightarrow{\sim} (X \times R \text{-mod})$. Let P be a specter for H and let P_R denote the associated R-specter. Then there is a "Mayer-Vietoris" spectral sequence

$$E_{pq}^2 = H_p(X; \underline{\pi_q(\tau(P_R))}) \Rightarrow R_{p+q}^{\tau}(P).$$

Here $\pi_q(\tau(P_R))$ denotes the cosheaf $U \mapsto \pi_q((\tau(P_R)|_U)/U)$.

Suppose $\{U_i\}$ is a fixed contractible cover of X, and let $- \wedge_{U_{ij}} \rho_{ij}$ be transition functions defining H. Suppose the specter P is presented by parametrized spectra P_i on U_i together with identifications $\gamma_{ij} : P_i|_{U_{ij}} \wedge_{U_{ij}} \rho_{ij} \xrightarrow{\sim} P_j|_{U_{ij}}$. Then the above spectral sequence has the form

$$E_{pq}^{1} = \bigoplus_{i_1 < \dots < i_p} R_q(P_{i_1}|_{U_{i_1 \dots i_p}}) \Rightarrow R_{p+q}^{\tau}(P).$$

If the trivialization is given by automorphisms τ_i : $(U_i \times R\text{-mod}) \xrightarrow{\sim} (U_i \times R\text{-mod})$ appropriately compatible with the R-haunt transition functions $(\rho_{ij})_R = \rho_{ij} \wedge_{U_{ij}} R$, then the d^1 differential is given by the maps

$$\begin{aligned} R_q(P_{i_1}|_{U_{i_1\dots i_p}}) &\xrightarrow{(\iota_k)_*} R_q(P_{i_1}|_{U_{i_1\dots \widehat{i_k}\dots i_p}}) \\ R_q(P_{i_1}|_{U_{i_1\dots i_p}}) &\xrightarrow{(\iota_1)_* \circ (\tau_{i_2})_*^{-1} \circ (\tau_{i_1})_*} R_q(P_{i_2}|_{U_{i_2\dots i_p}}) \end{aligned}$$

Here $\iota_k: U_{i_1...\hat{i_k}...i_p} \to U_{i_1...i_p}$ and $\iota_1: U_{i_2...i_p} \to U_{i_1...i_p}$ denote the inclusions.

The first half of this proposition is just a statement, in ordinary parametrized homotopy theory, about the parametrized spectrum $\tau(P_R)$; it is not in itself particularly useful because one must expressly identify $\tau(P_R)$ in terms of the original specter P in order to compute the cosheaf homology. The second half is more explicit and addresses the situation that actually arises with twisted parametrized spectra. In particular, the above E^1 term does not depend on the trivialization τ and can be immediately computed in any given case.

Chapter 5

Polarized Hilbert Manifolds and Semi-Infinite Spectra

Thusfar our discussion has been purely topological: the category of spectra has a complicated space of automorphisms and it is natural to study bundles of categories of spectra and their associated sections, namely twisted parametrized spectra or "specters". These bundles are, however, intimately connected to the geometry of infinite-dimensional manifolds, and homotopy-theoretic invariants of such manifolds often take the form of twisted parametrized spectra. In the first part of this section, we describe the relevant geometry, namely symplectic polarizations of real Hilbert bundles, and we show how a manifold equipped with this structure gives rise to a bundle of categories of spectra, that is to a haunt. We also discuss a related structure, a "unitary" polarization and investigate the invariants of specters for haunts associated to unitary polarizations; the resulting description of these invariants provides an extensive generalization of the Cohen-Jones-Segal complex-oriented Floer invariants [8]. In the second part of this section, we introduce a conjectural construction of the category of specters, for a given polarized bundle, in terms of parametrized semi-infinitely indexed spectra. Specifically, instead of indexing spectra on finite-dimensional subspaces of a countably infinite-dimensional vector space, we introduce spectra indexed on the semi-infinite subspaces of a Hilbert space that are compatible with a fixed polarization. A parametrized version of these semi-infinite

spectra provides an explicit geometric viewpoint on the category of specters.

5.1 The Homotopy Theory of Polarized Bundles

We describe four types of polarizations, namely symplectic and complex on a real Hilbert space and symplectic and complex on a complex Hilbert space, and discuss the homotopy types of the corresponding classifying spaces—it turns out that there is no distinction between the two notions of polarization on a complex Hilbert space. We then describe polarized bundles and note that a symplectic polarization on a real Hilbert bundle gives rise to a haunt. We conclude by investigating the special class of "unitary" polarized bundles, namely symplectic polarizations of a real bundle that lift to polarizations of a complex bundle. In particular we show that under mild conditions, a specter for a unitary polarization admits HZ-, K-, and MU-homology invariants.

5.1.1 Polarizations of Hilbert Space

A finite dimensional vector bundle on a space X is classified by a map from X to the classifying space BO(n). The topology of the classifying space is governed by the (non-trivial) topology of the orthogonal group O(n) of automorphisms of \mathbb{R}^n . By contrast, the orthogonal group $O(\mathcal{H})$ of Hilbert space is contractible [27] and therefore the classifying space $BO(\mathcal{H})$ carries no topological information; indeed, all Hilbert bundles on a given space are isomorphic. In particular, if X is a Hilbert bundle, that is an infinite-dimensional manifold whose tangent bundle is a Hilbert bundle, then the tangent bundle of X carries no information at all about the topology of X. The situation is not as bad as it might seem, however, because many naturally occurring infinite-dimensional manifolds come equipped with a polarization. This polarization is a reduction of the structure group of X from the orthogonal group $O(\mathcal{H})$ to the socalled restricted orthogonal group $O_{\rm res}(\mathcal{H})$. This latter group does have an interesting topology, and so we can recover information about such a polarized Hilbert manifold X from its polarized tangent bundle.

There are various notions that go under the name "polarization" and we spend a moment describing and distinguishing them; references include [41, 44, 8] but the reader is warned that the terminology and definitions in those papers disagree with one another and at points with our treatment. We fix an infinite-dimensional separable real Hilbert space \mathcal{H} . Morally, a polarization of \mathcal{H} is an equivalence class of decompositions $V \oplus W$ of \mathcal{H} or $\mathcal{H}_{\mathbb{C}} = \mathcal{H} \otimes \mathbb{C}$ arising from an eigenvalue decomposition of an appropriate operator $J : \mathcal{H} \to \mathcal{H}$. The subspaces V and W are sums of collections of eigenspaces of J; that the polarization is an equivalence class of decompositions rather than a single decomposition reflects an ambiguity over whether to assign certain eigenspaces to V or to W. For example, if J is a self-adjoint Fredholm operator, then the associated decompositions $V \oplus W$ are roughly those in which V contains almost all the eigenspaces for negative eigenvalues of J and W contains almost all the eigenspaces for positive eigenvalues of J. If on the other hand J is a skew-adjoint Fredholm operator, then the decompositions are those in which V contains almost all the eigenspaces for positive imaginary eigenvalues and W contains almost all the eigenspaces for negative imaginary eigenvalues.

In practice many polarizations arise from self- and skew-adjoint Fredholm operators as above, but we can simplify the definitions of polarizations if we restrict attention to self- and skew-adjoint orthogonal isomorphisms. That is, suppose $J : \mathcal{H} \to \mathcal{H}$ is a self-adjoint orthogonal isomorphism; in this case, $J^2 = 1$ and so \mathcal{H} is split into the +1 and -1 eigenspaces V and V^{\perp} . Of course, any orthogonal decomposition arises as the eigenvalue decomposition of such an operator, and so decompositions $\mathcal{H} = V \oplus V^{\perp}$ are in one-to-one correspondence with orthogonal operators J with $J^2 = 1$. An equivalence class of such decompositions defines a "symplectic" polarization, as follows.

Definition 5.1.1. A symplectic polarization on a real Hilbert space \mathcal{H} is a collection of orthogonal decompositions $\{\mathcal{H} = V \oplus V^{\perp}\}$ satisfying the following conditions:

- both V and V^{\perp} are infinite dimensional,
- for any two decompositions $V \oplus V^{\perp}$ and $W \oplus W^{\perp}$ in the collection, the projec-

tions $V \to W$ and $V^{\perp} \to W^{\perp}$ are Fredholm and the projections $V \to W^{\perp}$ and $V^{\perp} \to W$ are Hilbert-Schmidt,

• any decomposition $W \oplus W^{\perp}$ satisfying the second property with respect to a decomposition $V \oplus V^{\perp}$ in the collection is in the collection.

From now on, whenever we mention a decomposition of a Hilbert space, we implicitly assume that both factors of the decomposition are infinite dimensional. Corresponding to the above definition we have a restricted orthogonal group:

Definition 5.1.2. Let $V \oplus V^{\perp}$ be a fixed decomposition of the real Hilbert space \mathcal{H} . The symplectic restricted orthogonal group $O^s_{res}(\mathcal{H})$ is the subgroup of the orthogonal group $O(\mathcal{H})$ of operators ϕ such that $\phi(V) \oplus \phi(V^{\perp})$ is in the same symplectic polarization class as $V \oplus V^{\perp}$.

Note that this is *not* the group that Pressley and Segal [41] refer to as the restricted orthogonal group. Indeed we will see later that it has a radically different homotopy type than their $O_{\text{res}}(\mathcal{H})$.

Note 5.1.3. The space of symplectic polarizations of a real Hilbert space is the quotient $O(\mathcal{H})/O_{\rm res}^s(\mathcal{H})$.

Now by contrast, suppose we had begun with a skew-adjoint orthogonal isomorphism $J : \mathcal{H} \to \mathcal{H}$; in this case, $J^2 = -1$ and so $\mathcal{H}_{\mathbb{C}}$ is decomposed into the +i and -i eigenspaces W and \overline{W} . Indeed, for any decomposition $\mathcal{H}_{\mathbb{C}} = W \oplus \overline{W}$ of $\mathcal{H}_{\mathbb{C}}$ into a subspace W and its conjugate \overline{W} , the following three conditions are equivalent:

- the decomposition is orthogonal with respect to the Hermitian metric $\langle -, \rangle$ on $\mathcal{H}_{\mathbb{C}}$ extending the inner product on \mathcal{H} ,
- the subspaces W and \overline{W} are isotropic with respect to the bilinear form $(a, b) = \langle a, \overline{b} \rangle$ on $\mathcal{H}_{\mathbb{C}}$,
- there is an orthogonal operator $J : \mathcal{H} \to \mathcal{H}$ with $J^2 = -1$ having W and \overline{W} as its +i and -i eigenspaces respectively.

We refer to such decompositions as "orthogonal". An equivalence class of these orthogonal decompositions defines a "complex" polarization:

Definition 5.1.4. A complex polarization on a real Hilbert space \mathcal{H} is a collection of orthogonal decompositions $\{\mathcal{H}_{\mathbb{C}} = W \oplus \overline{W}\}$ satisfying the same conditions as in definition 5.1.1. Given a fixed orthogonal decomposition $W \oplus \overline{W}$, the complex restricted orthogonal group $O_{\text{res}}^c(\mathcal{H})$ on the real Hilbert space \mathcal{H} is the subgroup of $O(\mathcal{H})$ of operators ϕ such that $\phi(W) \oplus \phi(\overline{W})$ is in the same complex polarization class as $W \oplus \overline{W}$.

This complex restricted orthogonal group is what Pressley and Segal [41] refer to as $O_{\text{res}}(\mathcal{H})$.

Note 5.1.5. The space of complex polarizations of a real Hilbert space is the quotient $O(\mathcal{H})/O_{\text{res}}^{c}(\mathcal{H}).$

There is yet another notion that goes under the name polarization. Let \mathcal{H} now be a complex Hilbert space. Suppose $J : \mathcal{H} \to \mathcal{H}$ is a self-adjoint unitary operator; then $J^2 = 1$ and \mathcal{H} is decomposed into +1 and -1 eigenspaces. Two such decompositions $V \oplus V^{\perp}$ and $W \oplus W^{\perp}$ are considered equivalent if, as in definition 5.1.1, the projections $V \to W$ and $V^{\perp} \to W^{\perp}$ are Fredholm and the other two projections are Hilbert-Schmidt. An equivalence class of these decompositions defines a polarization of the complex Hilbert space \mathcal{H} ; the group of unitary operators preserving such a polarization is called $U_{\rm res}(\mathcal{H})$ and the space of such polarizations is $U(\mathcal{H})/U_{\rm res}(\mathcal{H})$. Similarly if J is skew-adjoint unitary, then $J^2 = -1$ and \mathcal{H} is decomposed into +iand -i eigenspaces. Equivalence classes of these decompositions also give a notion of polarization, but because \mathcal{H} is complex there is a one-to-one correspondence between self- and skew-adjoint unitary operators and the two notions of polarization coincide. This correspondence, which in a sense encodes the two-fold complex Bott periodicity, can obscure the distinction between the two notions in the real case.

We briefly discuss the homotopy types of these various spaces of polarizations. As above, the symplectic restricted orthogonal group $O^s_{res}(\mathcal{H})$ of a real Hilbert space is the space of orthogonal operators $\phi : \mathcal{H} \to \mathcal{H}$ such that the projections $\phi(V) \to V$ and $\phi(V^{\perp}) \to V^{\perp}$ are Fredholm and the projections $\phi(V) \to V^{\perp}$ and $\phi(V^{\perp}) \to V$ are Hilbert-Schmidt, for a fixed decomposition $V \oplus V^{\perp}$. Suppose that we have chosen $\phi|_V$ such that $\phi(V) \to V$ is Fredholm and $\phi(V) \to V^{\perp}$ is Hilbert-Schmidt. Then the subspace $\phi(V^{\perp})$ is necessarily $\phi(V)^{\perp}$ and we can chose $\phi|_{V^{\perp}}$ to be any orthogonal isomorphism $V^{\perp} \xrightarrow{\cong} \phi(V)^{\perp}$ —the space of such choices is of course contractible. Specifying $\phi|_V$ amounts to choosing a Fredholm map $V \to V$ and a Hilbert-Schmidt map $V \to V^{\perp}$. The space of Hilbert-Schmidt operators is contractible and so $O_{\rm res}^s(\mathcal{H})$ has the homotopy type of the space of Fredholm operators, namely $\mathbb{Z} \times BO$. The associated space of polarizations $O(\mathcal{H})/O_{\rm res}^s(\mathcal{H})$ therefore has the homotopy type $B(\mathbb{Z} \times BO) \simeq U/O$. By contrast, Pressley and Segal [41] show that $O_{\rm res}^c(\mathcal{H})$ has the homotopy type $O/U \simeq \Omega O$ and so the space of polarizations $O(\mathcal{H})/O_{\rm res}^r(\mathcal{H})$ has the homotopy type $B(O/U) \simeq B(\Omega O) \simeq O$. When \mathcal{H} is a complex Hilbert space, the restricted group $U_{\rm res}(\mathcal{H})$ has the homotopy type $\mathbb{Z} \times BU$ and the space of polarizations is $U(\mathcal{H})/U_{\rm res}(\mathcal{H}) \simeq B(\mathbb{Z} \times BU) \simeq B(\Omega U) \simeq U$. The homotopy types of these various spaces of polarizations are summarized in table 5.1.

Polarization Type	Structure Group	Classifying Space
symplectic on $\mathcal{H}_{\mathbb{R}}$	$O^s_{\mathrm{res}} \simeq \mathbb{Z} \times BO$	$B(\mathbb{Z} \times BO) \simeq U/O$
complex on $\mathcal{H}_{\mathbb{R}}$	$O_{\rm res}^c \simeq \Omega O$	$B(\Omega O) \simeq O$
symplectic on $\mathcal{H}_{\mathbb{C}}$	$\overline{U_{\rm res}} \simeq \overline{\mathbb{Z} \times BU}$	$B(\mathbb{Z} \times BU) \simeq U$
complex on $\mathcal{H}_{\mathbb{C}}$	$U_{\rm res}\simeq \Omega U$	$B(\Omega U) \simeq U$

Table 5.1: The homotopy types of the classifying spaces for polarizations

We will be primarily concerned with symplectic polarizations of real Hilbert spaces and unless otherwise indicated, "polarization" will refer to this notion.

5.1.2 Symplectic Polarizations and Haunts

A priori a Hilbert bundle E on a space X is classified by a map $X \to BO(\mathcal{H})$ —of course this map contains no topological information. To give a polarization of this bundle is to specify, continuously in X, a polarization on each fibre of E:

Definition 5.1.6. A polarization of the Hilbert bundle E on the space X is a reduction of the structure group of E from $O(\mathcal{H})$ to $O^s_{res}(\mathcal{H})$. In other words it is a lift of the classifying map $X \to BO(\mathcal{H})$ to a map $X \to BO_{res}^s(\mathcal{H})$.

As there is no harm in doing so, we usually think of a polarization on the Hilbert bundle E simply as a map $X \to BO_{res}^s(\mathcal{H}) \simeq B(\mathbb{Z} \times BO)$. A polarization of a Hilbert manifold is simply a polarization of its (trivial) tangent bundle. The fundamental link between the geometry of polarizations and twisted parametrized stable homotopy theory is the association

$$\left\{\begin{array}{c} \text{Polarized Hilbert} \\ \text{bundles on } X \end{array}\right\} \rightsquigarrow \left\{\begin{array}{c} \text{Haunts} \\ \text{on } X \end{array}\right\}$$

This association is determined by composing the classifying map $X \to B(\mathbb{Z} \times BO)$ of the polarized bundle with the deloop of the J-homomorphism $B(\mathbb{Z} \times BO) \xrightarrow{BJ} B(\mathbb{Z} \times BG) = B(\mathbb{Z} \times BGL_1(S^0)).$

The basic philosophy behind this correspondence is that geometric structures on infinite dimensional manifolds are intimately connected with polarizations and that homotopy-theoretic information about these structures can be encoded in twisted parametrized spectra for the haunt associated to the polarization. The specific nature of this connection will be the subject of future work with Michael Hopkins [9] and Ciprian Manolescu [10]. Here we record a few illustrative examples of polarized manifolds and their associated haunts.

There are two widely utilized sources of polarized manifolds: the first is loop spaces of symplectic and almost complex manifolds, and the second is moduli spaces of connections in gauge theory—see for example [8]. We discuss the first source of examples. Given a symplectic manifold M, a choice of metric determines an almost complex structure $M \xrightarrow{c} BU(n)$ on the tangent bundle of M. The loop of this classifying map, or indeed of the classifying map for any almost complex manifold, can be used to determine a polarization on LM:

$$LM \xrightarrow{Lc} LBU(n) \to \Omega BU(n) \to \Omega BU \simeq U \to U/O \simeq B(\mathbb{Z} \times BO)$$

This polarization and its associated haunt can be highly nontrivial.

Example 5.1.7. Let S^6 have its usual almost complex structure. The haunt associated to the resulting polarization of the loop space LS^6 is nontrivial. Indeed, the classifying map

$$LS^6 \to U \to U/O \to B(\mathbb{Z} \times BG)$$

for this haunt restricts on $S^5 \hookrightarrow LS^6$ to a generator of $\pi_5(B(\mathbb{Z} \times BG)) = \pi_3(S^0) = \mathbb{Z}/24$. To see this, note that because $\pi_4(S^3) = \mathbb{Z}/2$, the Hurewicz map $\pi_6(BSU(3)) \to H_6(BSU(3))$ is multiplication by 2. Because the Euler characteristic of S^6 is 2, this implies that the almost complex structure $S^6 \to BSU(3)$ is a generator of $\pi_6(BSU(3)) \cong \pi_6(BSU)$. The loop $LS^6 \to LBU$ of this almost complex structure therefore induces an isomorphism on π_5 and the claim follows:

A specter for the resulting canonical haunt on LS^6 will have no global homotopy type. We will see in a moment though that any such specter has HZ-, K-, and MU-homology invariants.

5.1.3 Unitary Polarizations and Specter Invariants

Many examples of polarized manifolds have the property that the polarization map $X \to U/O$ factors through a map $X \to U$; (this is true for instance of example 5.1.7 above). As we saw earlier, the stable unitary group U classifies polarizations on complex Hilbert bundles E. The projection map $U \to U/O$ corresponds to viewing a polarization of E as a symplectic polarization of the underlying real Hilbert bundle $E_{\mathbb{R}}$; (similarly, the inclusion map $U \to O$ corresponds to viewing the polarization of E as a complex polarization of the underlying real Hilbert bundle bundle is unitary polarization of the underlying real Hilbert bundle $E_{\mathbb{R}}$). For lack of better terminology, we say that a symplectic polarization $X \to U/O$ of a real Hilbert bundle.

Haunts associated to unitary polarizations are much better behaved than arbitrary

haunts in a sense we now describe, and as a result their specters have a much simplified invariant theory. Suppose $X \to U \to U/O$ is the classifying map for a unitary polarization. The associated haunt H has a corresponding HZ-haunt H_{HZ} classified by the composite

$$X \to U \to U/O \to B(\mathbb{Z} \times BG) \to B(\mathbb{Z} \times B\mathbb{Z}/2) \simeq B\mathbb{Z} \times B^2\mathbb{Z}/2.$$

The homology group $H^2(U; \mathbb{Z}/2)$ is zero, so this composite factors through $B\mathbb{Z}$. Let $q \in \mathbb{Z}$ denote the smallest nonzero integer in the image of $H_1(X;\mathbb{Z}) \to H_1(B\mathbb{Z};\mathbb{Z})$, and suppose T is a specter for the haunt H on X. The haunt $H_{H\mathbb{Z}}$ is nontrivial; thus $T_{H\mathbb{Z}}$ does not have the form of a parametrized H \mathbb{Z} -module and so has no associated global H \mathbb{Z} -module, therefore no corresponding chain complex and no homology groups, per se. There is nevertheless a \mathbb{Z}/q -graded chain complex associated to $T_{H\mathbb{Z}}$ and therefore T has \mathbb{Z}/q -graded homology groups for invariants. This fact nicely explains the idea (by now prevalent in the literature) that semi-infinite and Floer homology theories are naturally graded not by the integers but by finite cyclic groups.

If the unitary polarization $X \to U \to U/O$ is trivial on H^1 , we have a more complete description of the corresponding haunt and its associated invariants:

Proposition 5.1.8. Let X be a space having the homotopy type of a finite CW complex and let E be a Hilbert bundle on X equipped with a unitary polarization $X \to U \to U/O$. Suppose the induced map $H^1(U;\mathbb{Z}) \to H^1(X;\mathbb{Z})$ is zero. Then any specter for the haunt H associated to the polarized bundle E admits global HZ-, K-, and MU-homology invariants.

Proof. Because of the H^1 condition, the classifying map for the polarization factors through SU, and because X is homotopy finite, this map in turn factors through some SU(n). The haunt H is therefore classified by a map $X \to SU(n) \to B(\mathbb{Z} \times BG)$. Let R denote one of the spectra HZ, K, or MU. The R-haunt H_R is classified by the composition $X \to SU(n) \to B(\mathbb{Z} \times BG) = B\operatorname{Pic}(S^0) \to B\operatorname{Pic}(R)$. It is not of course the case that the map $B\operatorname{Pic}(S^0) \to B\operatorname{Pic}(R)$ is null, but the composition $SU(n) \to B\operatorname{Pic}(S^0) \to B\operatorname{Pic}(R)$ will be null for the spectra R in question. This we can see by considering the map from the Atiyah-Hirzebruch spectral sequence for $B \operatorname{Pic}(S^0)^*(SU(n))$ to the Atiyah-Hirzebruch spectral sequence for $B \operatorname{Pic}(R)^*(SU(n))$. If for example R = K and n = 3 the map of E_2 terms is as follows:

0	0	0	0	0	0	0	0	0	0		0	0	0	0	0	0	0	0	0
1	\mathbb{Z}	0	0	\mathbb{Z}	0	\mathbb{Z}	0	0	\mathbb{Z}		$\mathbb{Z}/2$	0	0	$\mathbb{Z}/2$	0	$\mathbb{Z}/2$	0	0	$\mathbb{Z}/2$
2	$\mathbb{Z}/2$	0	0	$\mathbb{Z}/2$	0	$\mathbb{Z}/2$	0	0	$\mathbb{Z}/2$		$\mathbb{Z}/2$	0	0	$\mathbb{Z}/2$	0	$\mathbb{Z}/2$	0	0	$\mathbb{Z}/2$
3	$\mathbb{Z}/2$	0	0	$\mathbb{Z}/2$	0	$\mathbb{Z}/2$	0	0	$\mathbb{Z}/2$		0	0	0	0	0	0	0	0	0
4	$\mathbb{Z}/2$	0	0	$\mathbb{Z}/2$	0	$\mathbb{Z}/2$	0	0	$\mathbb{Z}/2$	\longrightarrow	Z	0	0	\mathbb{Z}	0	\mathbb{Z}	0	0	\mathbb{Z}
5	$\mathbb{Z}/24$	0	0	$\mathbb{Z}/24$	0	$\mathbb{Z}/24$	0	0	$\mathbb{Z}/24$		0	0	0	0	0	0	0	0	0
6	0	0	0	0	0	0	0	0	0		Z	0	0	\mathbb{Z}	0	$\mathbb Z$	0	0	\mathbb{Z}
7	0	0	0	0	0	0	0	0	0		0	0	0	0	0	0	0	0	0
8	$\mathbb{Z}/2$	0	0	$\mathbb{Z}/2$	0	$\mathbb{Z}/2$	0	0	$\mathbb{Z}/2$		Z	0	0	\mathbb{Z}	0	\mathbb{Z}	0	0	\mathbb{Z}

Total degree zero terms are boxed. The map is necessarily zero at E_2 in total degree zero and therefore $B\operatorname{Pic}(S^0)^0(SU(3)) \to B\operatorname{Pic}(K)^0(SU(3))$ is zero. The cases of HZ and MU and of other n are similar. We therefore conclude that the R-haunt $X \to B\operatorname{Pic}(R)$ is trivializable and so any corresponding specter has R-homology invariants.

The proposition says that for a large class of polarized manifolds, the associated semi-infinite homotopy types, namely the specters, will have homology, K-theory, and complex bordism invariants. This provides an explanation of and a substantial generalization of the remark in Cohen-Jones-Segal [8] that trivially polarized manifolds should have semi-infinite homology, K-theory, and complex bordism invariants.

5.2 Semi-Infinitely Indexed Spectra

In section 5.1.2 we saw that the classifying space for polarizations maps to the classifying space for haunts; any polarized bundle therefore has an associated haunt and a corresponding category of specters. In this section we will sketch, using a notion of parametrized semi-infinitely indexed spectra, a conjectural realization of this category of specters in terms of the geometry of the polarized bundle.

We begin by defining (non-parametrized) semi-infinitely indexed spectra. Classically a prespectrum E is presented by specifying a space $E(\mathbb{R}^n)$ for each integer n together with appropriate structure maps. This notion naturally evolved (in work of May and company [36, 30]) into that of coordinate free prespectra; a coordinate free prespectrum is presented by giving a space E(V) for every finite dimensional subspace V of a fixed countably-infinite dimensional inner product space \mathbb{R}^{∞} , together with appropriate structure maps $\Sigma^{W-V}E(V) \to E(W)$ for each inclusion $V \subset W$. A prespectrum is a spectrum if the adjoint structure maps $E(V) \to \Omega^{W-V}E(W)$ are homeomorphisms. Such a spectrum has the property that if V and W have the same dimension, then E(V) and E(W) are homeomorphic, and moreover E(V)varies continuously with V. This can be seen by noting that for any compact family $\{V_t\}$ of finite-dimensional subspaces of \mathbb{R}^{∞} , there is a finite-dimensional subspace Wsuch that W contains all the subspaces of the family; the spaces $E(V_t)$ are therefore determined as $\Omega^{W-V_t}E(W)$, which is evidently a continuous family. Later on, Elmendorf [17] included as part of the definition of a spectrum the requirement that the spaces E(V) vary continuously in V.

The fundamental idea behind semi-infinitely indexed spectra is that the natural subspaces of a polarized Hilbert space \mathcal{H} are not the finite dimensional subspaces but the "negative energy" or "semi-infinite" subspaces, that is the subspaces V such that $V \oplus V^{\perp}$ is a decomposition of \mathcal{H} in the given polarization class. A semi-infinitely indexed spectrum is then roughly an assignment of a space E(V) to each such semi-infinite V, together with appropriate structure maps. It is not the case that given a compact family $\{V_t\}$ of semi-infinite subspaces, there exists a semi-infinite subspace W containing all the V_t ; indeed, there exist decompositions $V \oplus V^{\perp}$ and $W \oplus W^{\perp}$ in the same polarization class such that V and W span the whole Hilbert space \mathcal{H} . As such, it is important that we impose a continuity condition on the spaces E(V)—we do so roughly along the lines of [17] and we thank Mike Hopkins for bringing that reference to our attention.

Definition 5.2.1. Let X be a space together with a distinguished subset $X^{(2)}$ of X^2 and a finite dimensional vector bundle γ on $X^{(2)}$. Define $X^{(3)}$ to be $\{(a, b, c) \in X^3 | (a, b), (a, c), (b, c) \in X^{(2)}\}$. Let p_{12} , p_{13} , and p_{23} denote the projections $X^{(3)} \rightarrow X^{(2)}$ to the indicated factors, and similarly denote by p_1 , p_2 , and p_3 the projections

 $X^{(3)} \to X$. Suppose there is an identification $p_{13}^*\gamma = p_{12}^*\gamma \oplus p_{23}^*\gamma$. Then an Xprespectrum is a bundle T on X together with a map $\sigma : S^{\gamma} \wedge_{X^{(2)}} p_1^*T \to p_2^*T$ such that the diagram

commutes. This data forms an X-spectrum if the adjoint of σ is a homeomorphism.

Suppose \mathcal{H} is equipped with a fixed polarization. Let $Gr_{res}(\mathcal{H})$ denote the grassmannian of decompositions $V \oplus V^{\perp}$ in the polarization class. The space $Gr_{res}^{(2)}(\mathcal{H})$ is the set of pairs of decompositions $(V \oplus V^{\perp}, W \oplus W^{\perp})$ such that $V \subset W$, and the vector bundle γ is the orthogonal complement $V^{\perp W}$.

Definition 5.2.2. A semi-infinitely indexed (pre)spectrum, or "semi-infinite (pre)spectrum" for short, is a $Gr_{res}(\mathcal{H})$ -(pre)spectrum.

We bother with the abstract definition 5.2.1 because it facilitates comparisons between X-spectra as X varies. In particular, fix a decomposition $\mathcal{H} = \mathcal{H}^- \oplus \mathcal{H}^+$ in the polarization class and let $Gr(\mathcal{H}^+)$ denote the grassmannian of finite dimensional subspaces of \mathcal{H}^+ . We will refer to $Gr(\mathcal{H}^+)$ -spectra as Hilbert spectra. Furthermore, denote by \mathbb{R}^∞ a countably-infinite-dimensional dense subspace of \mathcal{H}^+ and let $Gr(\mathbb{R}^\infty)$ be the grassmannian of finite dimensional subspaces of \mathbb{R}^∞ . We now have natural restriction maps

$$Gr_{res}(\mathcal{H})$$
-spectra $\to Gr(\mathcal{H}^+)$ -spectra
 $(V \subset \mathcal{H} \rightsquigarrow E(V)) \mapsto (W \subset \mathcal{H}^+ \rightsquigarrow E(\mathcal{H}^- \oplus W))$
 $Gr(\mathcal{H}^+)$ -spectra $\to Gr(\mathbb{R}^\infty)$ -spectra
 $(W \subset \mathcal{H}^+ \rightsquigarrow F(W)) \mapsto (U \subset \mathbb{R}^\infty \rightsquigarrow F(U))$

A detailed treatment of the theory of semi-infinitely indexed spectra and their relationship to ordinary spectra will appear elsewhere—in particular one must construct
a semi-infinite spectrification functor and semi-infinite sphere spectra (leading to a semi-infinite notion of stable weak equivalence) and one must build left adjoints to the two restriction maps above. For now we leave as a conjecture the following:

Conjecture 5.2.3. There is a notion of weak equivalence of semi-infinite spectra and a notion of weak equivalence of Hilbert spectra such that the above restriction map from $Gr_{res}(\mathcal{H})$ -spectra to $Gr(\mathcal{H}^+)$ -spectra induces an equivalence of (simplicial) homotopy categories. Similarly the restriction map from $Gr(\mathcal{H}^+)$ -spectra to $Gr(\mathbb{R}^\infty)$ -spectra induces an equivalence of homotopy categories.

 $\sim \sim \sim \sim \sim$

Parametrized semi-infinite spectra are no more difficult to define than semi-infinite spectra. The idea of parametrized universes (of the countably-infinite variety) for spectra first appeared in Elmendorf [17]. Though we were not aware of this reference during our development of parametrized semi-infinite spectra, it is a very clean presentation of the classical case and we follow it in spirit. Elmendorf's real insight was not so much the use of parametrized universes, per se, as the realization that one could build a category of spectra on all (parametrized) universes at once and that this larger category was substantially better than the category of spectra indexed on a single universe. Unfortunately, basic facts about countably-infinite universes that made this possible fail to be true of semi-infinite universes, and so we necessarily shy aware from this aspect of Elmendorf's treatment.

Let E be a polarized Hilbert bundle on a space X. The restricted grassmannian $Gr_{res}(E)$ of this bundle is the space of semi-infinite subspaces V of fibres E_p , $p \in X$, of E; that is, the subspaces V are such that $V \oplus V^{\perp}$ is a decomposition in the polarization class of E_p . The space of pairs $Gr_{res}^{(2)}(E)$ and the finite-dimensional bundle γ are defined as before. We immediately have

Definition 5.2.4. A parametrized semi-infinite spectrum for the polarized bundle E is a $Gr_{res}(E)$ -spectrum.

Granting conjecture 5.2.3, the claim is quite simply that the category of parametrized semi-infinite spectra for the polarized Hilbert bundle E on the space X is homotopi-

cally equivalent to the category of specters (twisted parametrized spectra) for the haunt on X associated to the polarized bundle. Thereby, semi-infinite spectra provide a geometric realization of the homotopy-theoretic correspondence, via the classifying map $B(\mathbb{Z} \times BO) \rightarrow B(\mathbb{Z} \times BG)$, between polarizations and haunts.

Part II

On the Twisted K-Homology of Simple Lie Groups

Chapter 6

Introduction

By way of motivation we present six interpretations of twisted K-theory. These interpretations inform the methods and perspectives adopted in the paper but are otherwise unnecessary for what follows. We then summarize our results on the twisted K-homology of simple Lie groups and overview our main techniques, namely the twisted Rothenberg-Steenrod spectral sequence, Tate resolutions, Bott generating varieties, and twisted $Spin^c$ bordism.

6.1 Six Interpretations of Twisted *K*-Theory

6.1.1 1-Dimensional Elements in Elliptic Cohomology

A twisting on a space X of a cohomology theory represented by a spectrum R is a bundle of spectra on X with fibre R and the associated twisted cohomology of X is given by the homotopy classes of sections of this bundle. Such twistings are classified by maps from X to the classifying space B Aut R of homotopy automorphisms of the spectrum R. If R is an A_{∞} ring spectrum, the classifying space BGL_1R of homotopy units in R maps to B Aut R and thereby classifies a subset of the twistings—we refer to these twistings as elementary.

The classifying space $BGL_1H\mathbb{C}$ for elementary twistings of ordinary cohomology with complex coefficients is $B\mathbb{C}^*$; (here $H\mathbb{C}$ denotes the Eilenberg-MacLane spectrum for \mathbb{C}). There is a map $B\mathbb{C}^* \to \mathbb{Z} \times BU$ of this classifying space into the representing space for K-theory; any twisting $X \to BGL_1 \mathbb{HC}$ for the ordinary cohomology of X therefore determines a K-theory class on X. Of course, there is a natural geometric interpretation of the K-theory classes arising in this way, namely as the classes represented by flat line bundles on X. The twisted cohomology of X is simply the cohomology of X with coefficients in the line bundle, reinterpreted as the homotopy classes of sections of an associated \mathbb{HC} bundle.

The classifying space BGL_1K for elementary twistings of complex K-theory splits, as an infinite loop space, as $T \times S$. The factor T is a $K(\mathbb{Z},3)$ bundle over $K(\mathbb{Z}/2,1)$ which splits as a space but has nontrivial infinite loop structure classified by $\beta Sq^2 \in$ $H^{3}(H(\mathbb{Z}/2);\mathbb{Z})$. There is a natural infinite loop map $T \to TMF$ from T to the representing space for topological modular forms, and so by projecting through Ta map $BGL_1K \to TMF$. In particular an elementary twisting of K-theory for X determines a TMF-class on X. (Notice that TMF is the analog of real K-theory, that is of KO, and so the map $BGL_1K \rightarrow TMF$ corresponds to the composite $BGL_1 H\mathbb{C} \to \mathbb{Z} \times BU \to \mathbb{Z} \times BO$; it is not known whether there exists an appropriate factorization $BGL_1K \to E \to TMF$ for every elliptic spectrum E.) The geometric interpretation of these TMF classes is simplified if we restrict our attention to those classes coming from twistings involving only the $K(\mathbb{Z},3)$ factor of T. Such a twisting is determined by a map $X \to K(\mathbb{Z},3)$ or equivalently by a BS^1 bundle on X. We think of this bundle as a stack locally isomorphic to the sheaf of line bundles on Xand as such as a 1-dimensional 2-vector bundle on X. In this sense we imagine the TMF classes coming from K-theory twistings as 1-dimensional elliptic elements and twisted K-theory as K-theory with coefficients in this "elliptic line bundle".

6.1.2 Projective Hilbert Space Bundles

There is a very simple and well-known reformulation of twistings of K-theory as projective Hilbert space bundles and of the corresponding twisted K-theory groups as families of Fredholm operators on these bundles. Indeed, the space of unitary operators on Hilbert space is contractible, so the group of projective unitary operators has the homotopy type of BS^1 . As such a twisting $\alpha : X \to K(\mathbb{Z}, 3)$ of K-theory determines a projective bundle $\mathcal{H}(\alpha)$ of Hilbert spaces on X. The space of Fredholm operators on a Hilbert space has the homotopy type of $\mathbb{Z} \times BU$ and depends only on the projectivization of the Hilbert space. Sections of the $\mathbb{Z} \times BU$ bundle associated to the twisting α can therefore be thought of as Fredholm operators on the projective bundle $\mathcal{H}(\alpha)$. It remains to develop a general index theory for elliptic operators on these projective bundles, but substantial progress has been made by Mathai, Melrose, and Singer [35], who prove an index theorem in the case that the twisting α is a torsion class in $H^3(X;\mathbb{Z})$.

6.1.3 $K(\mathbb{Z},2)$ -Equivariant K-Theory

We would like to discuss an algebro-geometric model for twisted K-theory, and the proper formulation is suggested by reinterpreting twisted K-theory as a $K(\mathbb{Z}, 2)$ equivariant theory; this formulation will also hint at connections with the representation theory of loop groups. As before, a twisting is a map $\alpha : X \to BK(\mathbb{Z}, 2)$ defining a principal $K(\mathbb{Z}, 2)$ -bundle $P(\alpha)$ on X. The set of sections of the associated bundle $P(\alpha) \times_{K(\mathbb{Z},2)} (\mathbb{Z} \times BU)$ is the same as the set of $K(\mathbb{Z}, 2)$ -equivariant maps from $P(\alpha)$ to $\mathbb{Z} \times BU$; that is, the twisted K-theory of X is the " $K(\mathbb{Z}, 2)$ -equivariant" K-theory of $P(\alpha)$. In particular, elements of the twisted K-theory of X are represented by virtual vector bundles on the total space $P(\alpha)$ of the $K(\mathbb{Z}, 2)$ -principal bundle associated to the twisting; these vector bundles V are required to be $K(\mathbb{Z}, 2)$ -equivariant in the sense that for a line $L \in K(\mathbb{Z}, 2)$, the virtual vector space $V_{L\cdot x}$ at the point $L \cdot x \in P(\alpha)$ is equal to $L \otimes V_x$, for all points $x \in P(\alpha)$.

6.1.4 Perfect Complexes of α -Twisted Sheaves

Our 'space' X will now be a scheme, and a twisting of K-theory is a \mathbb{G}_m -gerbe on X. These gerbes are classified by $H^2(X;\mathbb{G}_m)$ and can be thought of as stacks locally isomorphic to the category of invertible sheaves. Elements of the twisted K-theory of X for a twisting gerbe α should be virtual sheaves of locally free \mathcal{O}_{α} -modules on α

that are $B\mathbb{G}_m$ -equivariant in an appropriate sense. More precisely an element of the twisted K-theory of X is a perfect complex of α -twisted sheaves on the gerbe α , that is a complex of α -twisted sheaves locally quasiisomorphic to a finite length complex of free finite rank \mathcal{O}_{α} -modules. In the topological situation the analogue of the perfect complex on α is a two term complex of bundles on $P(\alpha)$, each of countably infinite rank, with a differential that is locally an isomorphism off of a finite rank subbundle. We would like to emphasize that this notion of α -twisted K-theory elements on the scheme X does not depend on the class $\alpha \in H^2(X; \mathbb{G}_m)$ being torsion.

6.1.5 Central Extensions of Loop Groups

We now specialize to the case (which indeed will be our primary focus in this paper) that our space is a connected simply connected compact Lie group G. A twisting map $\alpha : G \to K(\mathbb{Z}, 3)$ gives a map from the free loop space LG to the classifying space BS^1 by the composition $LG \to LK(\mathbb{Z}, 3) \to \Omega K(\mathbb{Z}, 3) \simeq BS^1$, and thereby gives a principal S^1 -bundle on LG. The total space \widetilde{LG} of this principal bundle can be given a group structure as an S^1 -central extension of LG. The classifying space $B\widetilde{\Omega G}$ of the based loop central extension $\widetilde{\Omega G} \subset \widetilde{LG}$ is precisely the total space $P(\alpha)$ of the principal $K(\mathbb{Z}, 2)$ bundle over G. Moreover, to an irreducible highest-weight representation of \widetilde{LG} one can associate an equivariant map from $P(\alpha)$ to $\mathbb{Z} \times BU$ and thereby an element of the twisted K-theory of G [38]. The precise relation between the representation theory of loop groups and twisted K-theory is described by Freed, Hopkins, and Teleman [19]—they prove that the group of positive energy unitary representations of \widetilde{LG} is the twisted G-equivariant K-theory of G.

6.1.6 B-Fields and D-Branes

A great deal of the limelight focused on twisted K-theory has come from the widespread realization that certain boundary conditions in string theory naturally represent elements in the twisted K-theory of spacetime. In this context the twistings are represented by nontrivial Neveu-Schwarz B-fields; the elements of twisted K-theory are D-branes, submanifolds of spacetime with a twisted $Spin^c$ structure on their normal bundles. More generally, such a submanifold M may be equipped with a vector bundle V and the class represented by the pair (M, V) is the pushforward of V to the twisted K-theory of the ambient spacetime X. When the space X is a Lie group, as in this paper, the twisted K-theory can be thought of as a topological model for the space of D-branes in a Wess-Zumino-Witten model for conformal field theory. Frequently the spacetime X is itself a $Spin^c$ manifold; the D-branes are then twisted $Spin^c$ submanifolds and represent elements in the twisted K-homology of X. In this case, a D-brane M naturally represents a class in a more refined group, the twisted $Spin^c$ bordism of X, and there is a twisted index map that recovers the twisted Khomology class of M. This perspective guides the discussion of the twisted $Spin^c$ bordism of Lie groups in the last section of this paper.

6.2 Results

We prove that the twisted K-homology ring of a simple Lie group is an exterior algebra tensor a cyclic group, we give a detailed description of the orders of these cyclic groups in terms of the dimensions of irreducible representations of related groups, and we show that these orders originate, via a twisted index map, from relations in the twisted $Spin^c$ bordism group.

Theorem 6.2.1. Let G be a compact, connected, simply connected, simple Lie group of rank n. The twisted K-homology ring of G with nonzero twisting class $k \in$ $H^{3}(G;\mathbb{Z}) \cong \mathbb{Z}$ is an exterior algebra of rank n-1 tensor a cyclic group:

$$K^{\tau(k)}(G) \cong \Lambda[x_1, \dots, x_{n-1}] \otimes \mathbb{Z}/c(G, k).$$

Here c(G, k) is an integer depending on the group and the twisting.

This fact was first noticed in the case of SU(n) by Hopkins. The proof is in section 8 for groups other than Spin(n), and in section 9.4 for Spin(n).

Theorem 6.2.2. For the classical groups, the cyclic orders c(G, k), k > 0, of the twisted K-homology groups of G are:

$$\begin{aligned} c(SU(n+1),k) &= \gcd\left\{\binom{k+i}{i} - 1 : 1 \le i \le n\right\} \\ c(Sp(n),k) &= \gcd\left\{\sum_{-k \le j \le -1} \binom{2j+2(i-1)}{2(i-1)} : 1 \le i \le n\right\} \\ c(Spin(4n-1),k) &= \gcd\left\{\binom{k}{i} : 1 \le i \le 2n-2\right\} \\ &\cup \{2\binom{k}{2n-1}\} \cup \{2\binom{k}{2i+1} + \binom{k}{2i} : n \le i \le 2n-2\} \} \\ c(Spin(4n+1),k) &= \gcd\left\{\binom{k}{i} : 1 \le i \le 2n-1\} \cup \{2\binom{k}{2i+1} + \binom{k}{2i} : n \le i \le 2n-1\} \} \\ c(Spin(4n+2),k) &= \gcd\left\{\binom{k}{i} : 1 \le i \le 2n\} \\ &\cup \{2\binom{k}{2n+1}\} \cup \{2\binom{k}{2i+1} + \binom{k}{2i} : n+1 \le i \le 2n-1\} \} \\ c(Spin(4n),k) &= \gcd\left\{\binom{k}{i} : 1 \le i \le 2n-1\} \cup \{2\binom{k}{2i+1} + \binom{k}{2i} : n \le i \le 2n-1\} \right\}. \end{aligned}$$

(Note that c(G, -k) = c(G, k). The formulas for c(Spin(4n-1), k) and c(Spin(4n), k) exclude the degenerate case n = 1.) The proofs for SU(n), Sp(n), and Spin(n) occur respectively in sections 9.2, 9.3, and 9.4. A general method for computation, applicable to the exceptional groups, is discussed in section 9.5, and the cyclic order for G_2 is given in section 9.2.

Proposition 6.2.3. Let G be as in Theorem 6.2.1. Suppose M_i is a collection of $Spin^c$ manifolds over ΩG whose fundamental classes generate $K.\Omega G$ as an algebra. Then there are twisted $Spin^c$ structures on the bordisms $W_i = M_i \times I$ such that the cyclic order of the twisted K-homology of G is $gcd(ind(\partial W_1), \ldots, ind(\partial W_n))$, where ind : $MSpin^c_* \to K.*$ is the index map from $Spin^c$ bordism to K-homology.

The proof of this proposition is the focus of section 10.2.

6.3 Techniques and Overview

The primary tool for calculating twisted K-homology rings is the twisted Rothenberg-Steenrod spectral sequence; this is the original method used by Hopkins in the case G = SU(n). The spectral sequence is:

$$E^2 = \operatorname{Tor}^{K_{\cdot}\Omega G}(\mathbb{Z}, \mathbb{Z}_{\tau(k)}) \Rightarrow K_{\cdot}^{\tau(k)}(G)$$

where $\mathbb{Z}_{\tau(k)}$ is the integers with a twisted $K.\Omega G$ -module structure depending on k. In section 7.1 we present various generalities about twisted homology theories; then in section 7.2 we use a method of Segal [43] to construct this Rothenberg-Steenrod spectral sequence in twisted K-homology.

As the K-homology rings of loop spaces of simple Lie groups are known, our primary task is computing the Tor groups over these rings. Remarkably, for $G \neq Spin(n)$ this can be done without identifying the twisted $K.\Omega G$ -module structure on \mathbb{Z} . These Tor groups are calculated in section 8 by an iterated series of filtration spectral sequences applied to a judiciously chosen Tate resolution. The spectral sequences are seen to collapse and to be extension-free, completing the proof of Theorem 6.2.1 for $G \neq Spin(n)$.

The Tor computation for Spin(n) requires a detailed knowledge of the twisted module structure on \mathbb{Z} ; this module structure is also precisely what is needed to identify the cyclic orders of the twisted K-homology groups. The best way to identify this module structure is via generating varieties for the loop space of the group, and this is the subject of section 9. Sections 9.2, 9.3, and 9.5 describe generating varieties for various groups, compute the cyclic orders in the corresponding cases, and discuss a general method for determining the cyclic order. Section 9.4 describes the twisted module structure for Spin(n) and presents the belated Tor calculation for this group.

The computation in section 9 of the cyclic order in terms of the dimensions of irreducible representations does not give much geometric insight into these torsion groups. We give, in section 10, an interpretation of these orders in terms of relations in the twisted $Spin^c$ bordism group of G. The main tool, presented in section 10.1, is

a cocycle model for twisted $Spin^c$ bordism. This model allows explicit descriptions of nullbordisms of particular $Spin^c$ manifolds over G corresponding to relations in the twisted K-homology of G—see section 10.2. We conclude in section 10.3 by discussing potential representatives in $MSpin^{c,\tau}(G)$ for the exterior generators of $K^{\tau}(G)$.

Chapter 7

Twisted K-Theory and the Rothenberg-Steenrod Spectral Sequence

7.1 Twisted Homology Theories

We review the definitions and basic properties of twisted homology and cohomology theories. There are by now various models for these theories, but the following perspective owes as much to Goodwillie as to folklore.

For a spectrum F, the cohomology of a space X with coefficients in F can be defined as

$$F^n(X) := \operatorname{colim} \Gamma_h(X, X \times \Omega^i F_{i+n});$$

here $\Gamma_h(X, E)$ refers to homotopy classes of sections of the (here trivial) bundle Eon X. The maps in the colimit are induced by applying the usual structure maps $\Omega^i F_{i+n} \to \Omega^{i+1} \Sigma F_{i+n} \to \Omega^{i+1} F_{i+1+n}$ fibrewise to the bundle $X \times \Omega^i F_{i+n} \to X$. Now let E be a bundle of based spectra over X, with fibre spectrum F; this means in particular that for each i we have a fibration $E_i \to X$, a section $X \to E_i$, and a fibrewise structure map $\Sigma_X E_i \to E_{i+1}$. (Note that Σ_X denotes fibrewise suspension and Ω_X will denote the fibrewise loops.) The cohomology of X with coefficients in E is defined to be

$$E^n(X) := \operatorname{colim} \Gamma_h(X, \Omega^i_X E_{i+n})$$

where the colimit maps are, as expected, induced by $\Omega_X^i E_{i+n} \to \Omega_X^{i+1} \Sigma_X E_{i+n} \to \Omega_X^{i+1} E_{i+1+n}$.

The parallel in homology is similar. The homology of X with coefficients in F is

$$F_n(X) := \operatorname{colim}[S^{i+n}, (X \times F_i)/X],$$

with maps induced by $\Sigma((X \times F_i)/X) = (X \times \Sigma F_i)/X \to (X \times F_{i+1})/X$. As above, when E is a bundle of based spectra, we have a 'base point' section $X \to E_i$ for all *i*. The homology of X with coefficients in E is

$$E_n(X) := \operatorname{colim}[S^{i+n}, E_i/X];$$

the colimit maps are induced by $\Sigma(E_i/X) = (\Sigma_X E_i)/X \to E_{i+1}/X$.

For completeness we also mention the reduced analogs of homology and cohomology with coefficients in a bundle of spectra. The reduced cohomology with coefficients in a trivial F bundle can be given as

$$\tilde{F}^n(X) := \operatorname{colim} \Gamma^b_h(X, X \times \Omega^i F_{i+n}),$$

that is as the colimit of homotopy classes of sections taking the base point of X to the basepoint of $\Omega^i F_{i+n}$. The reduced cohomology with coefficients in E is then

$$\tilde{E}^n(X) := \operatorname{colim} \Gamma^b_h(X, \Omega^i_X E_{i+n});$$

the maps are induced as before. Similarly, the reduced homology with coefficients in a trivial bundle is

$$\tilde{F}_n(X) := \operatorname{colim}[S^{i+n}, (X \times F_i)/(X \vee F_i)].$$

The twisted reduced homology is finally

$$\tilde{E}_n(X) := \operatorname{colim}[S^{i+n}, E_i/(X \lor F_i)];$$

the maps are induced by $\Sigma(E_i/(X \vee F_i)) = (\Sigma_X E_i)/(X \vee \Sigma F_i) \to E_{i+1}/(X \vee F_{i+1}).$ Of course, these reduced groups are special cases of the relative groups:

$$E^{n}(X,A) := \operatorname{colim} \Gamma_{h}(X,A;\Omega^{i}_{X}E_{i+n},s(A)),$$

where s is the distinguished base point section; similarly,

$$E_n(X, A) := \operatorname{colim}[S^{i+n}, (E_i/X)/((E_i|_A)/A)].$$

The most important fact about twisted homology theories is that they are honest homology theories in an appropriate category. Indeed, consider the category of pairs (X, A) of spaces, where A is a closed subspace of X and X is equipped with a bundle E of based spectra with fibre spectrum F. From the above description of the homology $E_n(X, A)$, it is immediate that twisted homology on this category of pairs is a homology theory in the classical sense.

 $\sim \sim \sim \sim \sim$

In this paper we will only be concerned with bundles of spectra associated to principal $K(\mathbb{Z}, 2)$ bundles over our space X. As usual, we fix a model for $K(\mathbb{Z}, 3)$ and select a particular universal $K(\mathbb{Z}, 2)$ bundle on it. A map $\alpha : X \to K(\mathbb{Z}, 3)$ gives a principal $K(\mathbb{Z}, 2)$ bundle $P(\alpha)$ on X, classified up to isomorphism by the homotopy class of the map. For any basepoint-preserving action of $K(\mathbb{Z}, 2)$ on a spectrum F, we can form the associated F bundle to $P(\alpha)$. The resulting bundle $P(\alpha) \times_{K(\mathbb{Z}, 2)} F$ is a bundle of based spectra on X, as above. Note that on the level of spaces, the action of $K(\mathbb{Z}, 2)$ on F is given by maps $K(\mathbb{Z}, 2)_+ \wedge F_i = (K(\mathbb{Z}, 2) \times F_i)/(K(\mathbb{Z}, 2) \times *) \to F_i$, and we often denote the spectrum action simply by a map $K(\mathbb{Z}, 2)_+ \wedge F \to F$. Our primary examples are twisted $Spin^c$ -bordism and twisted K-theory. The $K(\mathbb{Z}, 2)$ bundle

$$K(\mathbb{Z},2) = BU(1) \rightarrow BSpin^c \rightarrow BSO$$

is principal, with classifying map $BSO \xrightarrow{\beta w_2} BBU(1) = K(\mathbb{Z},3)$ classifying the integral Bockstein of the second Stiefel-Whitney class. In particular we have an action $K(\mathbb{Z},2) \times BSpin^c \to BSpin^c$; on Thom spaces this action is $K(\mathbb{Z},2)_+ \wedge MSpin^c \to MSpin^c$, that is, a based action of $K(\mathbb{Z},2)$ on the $Spin^c$ Thom spectrum. The α -twisted $Spin^c$ -bordism groups are then, of course, the stable homotopy groups $\pi_i((P(\alpha) \times_{K(\mathbb{Z},2)} MSpin^c)/X).$

The K-theory spectrum K is a module over $Spin^c$ -bordism by the usual index map $MSpin^c \xrightarrow{\text{ind}} K$. Taking the above based action $K(\mathbb{Z}, 2)_+ \wedge MSpin^c \xrightarrow{\phi} MSpin^c$ and smashing over $MSpin^c$ with K, we have a compatible based action on K-theory:

$$\begin{array}{c} K(\mathbb{Z},2)_{+} \wedge MSpin^{c} \xrightarrow{\phi} MSpin^{c} \\ \downarrow^{\text{id} \wedge \text{ind}} \downarrow & \downarrow^{\text{ind}} \\ K(\mathbb{Z},2)_{+} \wedge K \xrightarrow{\phi \wedge_{MSpin^{c}}(\text{id})} K \end{array}$$

The corresponding map on associated principal bundles $P(\alpha) \times_{K(\mathbb{Z},2)} MSpin^c \rightarrow P(\alpha) \times_{K(\mathbb{Z},2)} K$ induces a map from twisted $Spin^c$ -bordism to twisted K-theory which we call the twisted index map. This map will be important in section 10.

Twisted K-theory can be defined more directly by choosing an explicit model for $\mathbb{Z} \times BU$ (typically the space of Fredholm operators on a fixed Hilbert space \mathcal{H}) that admits an explicit action by some model for BU(1) (typically the space of projective unitary operators on \mathcal{H}); see, for example, Atiyah [3]. Whatever the formal definition, the geometric action being modeled is the following: a complex line L (representing a point in BU(1)) acts on a virtual-dimension-zero (or stable) vector space V (representing a point in BU) by tensor product, that is, $V \mapsto L \otimes V$.

It is worth noting, though, that this heuristic action of tensoring a vector bundle with a line can be misleading if we pay insufficient attention to the virtual dimension zero condition. It is tempting to think of elements of α -twisted K-cohomology as sections of an α -twisted gerbe of rank n, for some sufficiently large n; (such a section is locally a rank-n vector bundle, twisted globally by α). However, in this paper we are dealing with non-torsion twistings, and therefore no nontrivial element of twisted Kcohomology is representable by a section of any finite rank gerbe. We are inescapably in either a virtual-dimension-zero or an infinite-dimensional situation—which would seem to be a matter of personal penchant.

7.2 The Twisted Rothenberg-Steenrod Spectral Sequence

The "twisted" Rothenberg-Steenrod spectral sequence computing the twisted K-homology of a space is in fact the ordinary Rothenberg-Steenrod (a.k.a. homology Eilenberg-Moore) spectral sequence in an appropriate category, and as such requires little comment. We briefly recall the spectral sequence in generality, then describe its application to the geometric bar complex on the loop space of a simple Lie group.

We work in the category \mathcal{K} of pairs (X; E), where X is a space and E is a bundle of based spectra on X with fibre the K-theory spectrum; the morphisms are those bundle maps that are homotopy equivalences on each fibre. Similarly, we have a category of triples (X, A; E) where A is a closed subspace of X and E is again a bundle on X. As mentioned in the last section, the functors

 $(X, A; E) \mapsto E_n(X, A) = \operatorname{colim}[S^{i+n}, (E_i/X)/((E_i|_A)/A)]$

form a homology theory in the classical sense. In particular, for any simplicial object S in \mathcal{K} , there is a spectral sequence a la Segal [43] with E^2 term $H_p(E_q(S.))$ converging to the homology of the realization $E_{p+q}(|S.|)$.

Let G be a simple, simply connected Lie group and $k \in H^2(\Omega G; \mathbb{Z}) = \mathbb{Z}$ an integer describing a line bundle L^{-k} on the loop space ΩG . On the one hand there is the trivial projection map in \mathcal{K} from $(\Omega G; \Omega G \times K)$ to (*; K). On the other hand, there is a twisted map $\tau(k) : (\Omega G; \Omega G \times K) \to (*; K)$ given by $\Omega G \times K \xrightarrow{k \times \mathrm{id}} K(\mathbb{Z}, 2) \times K \to K$, where the last map is the $K(\mathbb{Z}, 2)$ action on the spectrum K described in section 7.1. The geometric bar construction $B_{\tau}\Omega G = B.(*, \Omega G, *_{\tau})$ is a simplicial object in \mathcal{K} . To describe the corresponding spectral sequence we need only compute the effect of $\tau(k)$ in homology and identify the realization $|B_{\tau}\Omega G|$.

Given a class ϕ in the K-homology of ΩG the image of ϕ under $\tau(k)$ is evidently equal to the evaluation $\langle \tau(k)^*(1), \phi \rangle$, where $\langle -, - \rangle$ denotes the Kronecker pairing. The pullback $\tau(k)^*(1)$ is L^k , and the resulting map $K \Omega G \xrightarrow{\langle L^k, - \rangle} K$.* defines a module structure on K.* which we denote $(K.*)_{\tau}$. The E^2 term of our spectral sequence is therefore $\operatorname{Tor}^{K \Omega G}(K.*, (K.*)_{\tau})$.

As a space the realization of $B_{\tau}\Omega G$ is evidently $B\Omega G \simeq G$; we identify the Kbundle. The K-bundle on the realization is defined by a 1-cocycle $\tau(k)$ with values in $K(\mathbb{Z}, 2)$ and as such is classified by the image of $\tau(k)$ in $H^3(B\Omega G; \mathbb{Z})$. We have $H^3(B\Omega G) \cong H^3(\Sigma \Omega G)$ and it is enough to identify the restriction of $\tau(k)$ to the 1skeleton $\Sigma \Omega G$ of $B\Omega G$. It is, however, immediate that this cocycle on the 1-skeleton of the geometric bar construction $B_{\tau}\Omega G$ has homology invariant $k \in H^3(\Sigma \Omega G)$. In summary:

Proposition 7.2.1. There is a spectral sequence of algebras with E^2 term

$$E_{pq}^2 = \operatorname{Tor}_{p,q}^{K.\Omega G}(K.*, (K.*)_{\tau})$$

converging as an algebra to the twisted K-homology $K_{p+q}^{\tau}(G)$.

Chapter 8

Tate Resolutions and $\operatorname{Tor}^{K.\Omega G}(\mathbb{Z}, \mathbb{Z}_{\tau})$ for $G \neq Spin(n)$

For each group G, we describe the K-homology of the loop space of G, give an appropriate Tate resolution of $K.* = \mathbb{Z}$ over $K.\Omega G$, and compute the torsion group using a series of filtration spectral sequences.

We recall Tate's main result on algebra resolutions over a commutative Noetherian ring R. An ideal $I \subset R$ is said to be generated by the regular sequence $a_1, \ldots, a_r \in R$ if $I = (a_1, \ldots, a_r)$ and a_i is not a zero-divisor in $R/(a_1, \ldots, a_{i-1})$ for all i.

Theorem 8.0.2 (Tate [46]). Let $A \subset B$ be ideals of R generated respectively by the regular sequences (s_1, \ldots, s_m) and (t_1, \ldots, t_n) . For any choice of constants $c_{ji} \in R$ such that $s_j = \sum_{i=1}^n c_{ji}t_i$, the differential graded algebra

$$D = (R/A\langle T_1, \dots, T_n \rangle \{S_1, \dots, S_m\}; d(T_i) = [t_i], d(S_j) = \sum_{i=1}^n [c_{ji}]T_i)$$

is a resolution of R/B as an R/A-module. Here the T_i are strictly skew commutative generators of degree 1, and the S_j are divided power algebra generators of degree 2.

In particular, $\operatorname{Tor}_{R/A}(R/B, Q)$ will be given as the homology $H(D \otimes_{R/A} Q)$. In our applications, R will be a polynomial ring $\mathbb{Z}[x_1, \ldots, x_n]$, the ideal A will depend on the

group, the ideal B will be (x_1, \ldots, x_n) , and Q will be an R/A-module \mathbb{Z}_{τ} on which x_i acts by an integer c_i depending on the group and the twisting class.

8.1 Tor for SU(n+1) and Sp(n)

Elementary calculation shows that the integral cohomology rings of SU(n + 1) and Sp(n) are exterior algebras on n generators. Application of the spectral sequence $\operatorname{Ext}^{H^{\cdot}(G;k)}(k,k) \Rightarrow H_{\cdot}(\Omega G;k)$, k a field, then implies that the integral Pontryagin rings $H_{\cdot}(\Omega SU(n + 1))$ and $H_{\cdot}(\Omega Sp(n))$ are both polynomial on n generators, all in even degree. In each case the Atiyah-Hirzebruch spectral sequence for K-theory then collapses, and the K-theory Pontryagin ring is again polynomial.

The Tate resolution in this case is especially simple, as the ideal A is trivial. Let G denote either SU(n+1) or Sp(n) and $k \in \mathbb{Z} \cong H^3(G;\mathbb{Z})$ the twisting class. Choose reduced generators x_i of $K.\Omega G$, so that $K.\Omega G \cong \mathbb{Z}[x_1,\ldots,x_n]$. (Note that, unless otherwise noted, we treat K-theory as $\mathbb{Z}/2$ -graded.) The $K.\Omega G$ module structure on \mathbb{Z}_{τ} is given, as in section 7.2, by the map $K.\Omega G \to K.*$ sending a class x to $\langle L^k, x \rangle$, where L is a generating line bundle. We defer the explicit evaluation of these maps to section 9. For now, we denote by c_i the image of x_i in \mathbb{Z}_{τ} ; of course this constant depends on both the group and the twisting, but we tend to omit both dependencies from the notation. By Tate's theorem,

$$\operatorname{Tor}^{K.\Omega G}(\mathbb{Z}, \mathbb{Z}_{\tau}) = H(\mathbb{Z}[x_1, \dots, x_n] \langle T_1, \dots, T_n \rangle \otimes_{\mathbb{Z}[x_1, \dots, x_n]} \mathbb{Z}_{\tau}; d)$$
$$= H(\mathbb{Z} \langle T_1, \dots, T_n \rangle; dT_i = c_i).$$

To evaluate this homology group we employ the following general procedure. Suppose we know the homology of the subalgebra generated by T_1, \ldots, T_i . We filter the subalgebra generated by T_1, \ldots, T_{i+1} by powers of T_{i+1} and look at the associated spectral sequence. The only differential is d^1 , which is given by multiplication by c_{i+1} , and by induction we can thus compute the homology of the original algebra.

We assume for now that c_1 is not zero; this is indeed the case (see sections 9.2)

and 9.3). The homology of $(\mathbb{Z}\langle T_1 \rangle, d)$ is \mathbb{Z}/c_1 . The, quite degenerate, spectral sequence of the filtration of $(\mathbb{Z}\langle T_1, T_2 \rangle, d)$ by T_2 is therefore

$$\mathbb{Z}/c_1 \xleftarrow{c_2} \mathbb{Z}/c_1.$$

The homology is $\mathbb{Z}/g_{12}\langle y_2 \rangle$, where $g_{12} = \gcd\{c_1, c_2\}$ and y_2 is an exterior class. More generally we will denote by $g_{1..i}$ the greatest common divisor $\gcd\{c_1, c_2, \ldots, c_i\}$. The induction step is, as expected, the homology of

$$\mathbb{Z}/g_{1..i}\langle y_2,\ldots,y_i\rangle \xleftarrow{c_{i+1}} \mathbb{Z}/g_{1..i}\langle y_2,\ldots,y_i\rangle,$$

and the Tor groups are given by

$$\operatorname{Tor}^{K.\Omega SU(n+1)}(\mathbb{Z},\mathbb{Z}_{\tau}) = \mathbb{Z}/(g_{1..n}(SU(n+1),k))\langle y_2,\ldots,y_{n-1}\rangle$$
$$\operatorname{Tor}^{K.\Omega Sp(n)}(\mathbb{Z},\mathbb{Z}_{\tau}) = \mathbb{Z}/(g_{1..n}(Sp(n),k))\langle y_2,\ldots,y_{n-1}\rangle.$$

We belabor this calculation only because, when we come to more complicated examples, especially Spin(n), it will help to have a clear model.

8.2 Tor for the Exceptional Groups

The exceptional Lie groups are nature's best attempts to make a finite dimensional Lie group out of $K(\mathbb{Z},3)$. In particular they are homotopy equivalent to $K(\mathbb{Z},3)$ through a range of dimensions, and so their loop spaces are homotopy equivalent to $K(\mathbb{Z},2)$ through a similar range. The K-homology of $K(\mathbb{Z},2)$ is the subalgebra of $\mathbb{Q}[a]$ generated by $\{a, \binom{a}{2}, \binom{a}{3}, \ldots\}$; see [1]. Extensive computations by Duckworth [11] show that for G exceptional, the K-homology $K.\Omega G$ differs from a polynomial ring only in the aforementioned low-dimensional flirtation with $K(\mathbb{Z},2)$. For example, Duckworth proves that $K.\Omega E_8$ is a polynomial ring on seven generators tensor the subalgebra of $\mathbb{Q}[a]$ generated by the elements $\{a, \binom{a}{2}, \binom{a}{3}, \binom{a}{4}, \binom{a}{5}\}$. In order to use Tate resolutions, we must give explicit algebra presentations of these K-homology rings:

Proposition 8.2.1. The K-homology rings of the loop spaces of the exceptional Lie groups are given by

$$\begin{split} K.\Omega G_2 &= \frac{\mathbb{Z}[a, b, x_3]}{(a(a-1)-2b)} \\ K.\Omega F_4 &= \frac{\mathbb{Z}[a, b, c, x_4, x_5, x_6]}{(a(a-1)-2b, b(a-2)-3c)} \\ K.\Omega E_6 &= \frac{\mathbb{Z}[a, b, c, x_4, x_5, x_6, x_7, x_8]}{(a(a-1)-2b, b(a-2)-3c)} \\ K.\Omega E_7 &= \frac{\mathbb{Z}[a, b, c, d, x_5, x_6, x_7, x_8, x_9, x_{10}]}{(a(a-1)-2b, b(a-2)-3c, b(b+1)-a(b+c)-2d)} \\ K.\Omega E_8 &= \frac{\mathbb{Z}[a, b, c, d, e, x_6, x_7, x_8, x_9, x_{10}, x_{11}, x_{12}]}{(a(a-1)-2b, b(a-2)-3c, b(b+1)-a(b+c)-2d, d(a-4)-5e)} \end{split}$$

Note that the unsightly third relation in the rings for E_7 and E_8 is essential and cannot be replaced by the more sensible relation c(a-3) - 4d. We remark that, because the 'lettered' generators in these K-homology rings come from corresponding generators in $K.(K(\mathbb{Z}, 2))$, the twisted pushforwards of these elements are easily computed. In particular, the twisted pushforward of a, denoted again by c_1 , is just k, the twisted pushforward of b is $c_2 = \binom{k}{2}$, of c is $c_3 = \binom{k}{3}$, and so on, with each generator mapping to its respective binomial coefficient.

As always, our starting point is the Tate resolution:

Tor<sup>K.
$$\Omega G_2(\mathbb{Z}, \mathbb{Z}_{\tau}) = H(\mathbb{Z}\langle T_1, T_2, T_3 \rangle \{S_1\}; dT_i = c_i, dS_1 = (c_1 - 1)T_1 - 2T_2).$$</sup>

Consider the subalgebra generated by T_1 , T_2 , and S_1 . If k is even, we can rewrite this DGA as

$$(\mathbb{Z}\langle T_1', T_2' \rangle \{S_1\}; dT_1' = 0, dT_2' = \frac{k}{2}, dS_1 = T_1'),$$

where $T'_1 = (k-1)T_1 - 2T_2$ and $T'_2 = \frac{k}{2}T_1 - T_2$. The Kunneth theorem immediately shows that the homology of this DGA is $\mathbb{Z}/(\frac{k}{2})$. If k is odd, we instead change the basis to $T'_1 = \frac{k-1}{2}T_1 - T_2$ and $T'_2 = kT_1 - 2T_2$. The algebra then takes the form

$$(\mathbb{Z}\langle T_1', T_2' \rangle \{S_1\}; dT_1' = 0, dT_2' = k, dS_1 = 2T_1'),$$

and by the Kunneth theorem its homology is \mathbb{Z}/k . In other words, the homology of the subalgebra in question is, in any case, \mathbb{Z}/g_{12} , where as before $g_{12} = \gcd\{c_1, c_2\}$. Filtering as in section 8.1 we see that the full Tor group is $\mathbb{Z}/g_{123}\langle y_3\rangle$.

The Tate resolution for F_4 gives

Tor<sup>K.
$$\Omega F_4$$</sup>($\mathbb{Z}, \mathbb{Z}_{\tau}$) = $H(\mathbb{Z}\langle T_1, T_2, T_3, T_4, T_5, T_6 \rangle \{S_1, S_2\};$
 $dT_i = c_i, dS_1 = (c_1 - 1)T_1 - 2T_2, dS_2 = (c_1 - 2)T_2 - 3T_3).$

We focus on the subalgebra generated by $\{T_1, T_2, T_3, S_1, S_2\}$. The method used for G_2 , of changing basis to split the algebra into simpler pieces, works here as well; the basis change now depends on k modulo 6. We spell out only the case $k = 1 \pmod{6}$. As basis change for the T_i 's we take

$$\begin{pmatrix} \frac{k-1}{2} & -1 & 0\\ -\frac{k-1}{6} & \frac{k-1}{3} & -1\\ \frac{3k-1}{2} & -1-k & 3 \end{pmatrix}.$$

The algebra then has the form

$$(\mathbb{Z}\langle T_1', T_2', T_3'\rangle \{S_1, S_2\}; dT_1 = dT_2 = 0, dT_3 = k, dS_1 = 2T_1', dS_2 = 3T_2' + T_1').$$

The spectral sequence associated to the filtration of the $\{T'_1, T'_2, S_1, S_2\}$ subalgebra by

powers of S_2 is

There are, of course, no differentials beyond d^1 and the homology of the $\{T'_1, T'_2, S_1, S_2\}$ subalgebra is therefore $\mathbb{Z}/6$ in odd degree, 0 in positive even degree, and \mathbb{Z} in degree zero; consequently the homology of the $\{T'_1, T'_2, T'_3, S_1, S_2\}$ subalgebra is \mathbb{Z}/k concentrated in degree zero. In general, ie for k not necessarily congruent to 1 modulo 6, this \mathbb{Z}/k is replaced by \mathbb{Z}/g_{123} and the full Tor group for F_4 is $\mathbb{Z}/g_{1..6}\langle y_4, y_5, y_6\rangle$. The computation for E_6 is identical, but for two additional exterior generators in the final Tor group.

This basis change approach quickly becomes impractical: for E_8 the congruence of k modulo 60 determines the structure of the basis change and of the subsequent homology computation. If we are willing to give up our ability to write down explicit generators for the Tor groups, we can do the computation without such a case by case analysis. We briefly reconsider the groups G_2 and F_4 . For G_2 the main step was computing the homology of the DGA

$$D = (\mathbb{Z}\langle T_1, T_2 \rangle \{S_1\}; dT_i = c_i, dS_1 = (c_1 - 1)T_1 - 2T_2);$$

recall that $c_1 = k$ and $c_2 = \binom{k}{2}$. The homology of the $\{T_1, T_2\}$ subalgebra is $\mathbb{Z}/g_{12}\langle y_2 \rangle$,

where the generator y_2 can be taken to be $-(c_1/g_{12})T_2$ modulo terms involving T_1 ; (we will refer to terms with lower indices, sensibly enough, as 'lower terms' and so say, for example, that " y_2 is $-(c_1/g_{12})T_2$ modulo lower terms"). Thus, when we filter D by powers of S_1 , the homology of D becomes the homology of

$$(\mathbb{Z}/g_{12}\langle y_2\rangle \{S_1\}; dS_1 = (2g_{12}/c_1)y_2).$$

Note that $2g_{12}/c_1$ is an integer, so this expression makes sense. We observe that $2g_{12}/c_1$ is actually a unit in \mathbb{Z}/g_{12} ; indeed $g_{12} = c_1/\gcd(2, c_1)$ so

$$gcd(2g_{12}/c_1, g_{12}) = gcd(2/gcd(2, c_1), c_1/gcd(2, c_1)) = 1.$$

The homology of D is therefore simply \mathbb{Z}/g_{12} , as previously noted, and thus the full Tor group is again an exterior algebra tensor a cyclic group.

The case of F_4 (and therefore of E_6) is again similar. The main step is the computation of the homology of the DGA

$$D = (\mathbb{Z}\langle T_1, T_2, T_3 \rangle \{S_1, S_2\}; dT_i = c_i, dS_1 = (c_1 - 1)T_1 - 2T_2, dS_2 = (c_1 - 2)T_2 - 3T_3).$$

(Here again $c_i = {k \choose i}$.) Using the G_2 result we see that the homology of the $\{T_1, T_2, T_3, S_1\}$ subalgebra is $\mathbb{Z}/g_{123}\langle y_3 \rangle$ where y_3 is $-(g_{12}/g_{123})T_3$ modulo lower terms. As above the homology of D is thereby reduced to the homology of

$$(\mathbb{Z}/g_{123}\langle y_3\rangle \{S_2\}; dS_2 = (3g_{123}/g_{12})y_3).$$

Again, this differential is an isomorphism, ie $3g_{123}/g_{12}$ is a unit in g_{123} . The trick is the same: observe that $g_{123} = g_{12}/\text{gcd}(3, g_{12}) = c_1/(\text{gcd}(3, c_1) \text{gcd}(2, c_1))$; from this we have

$$gcd(3g_{123}/g_{12}, g_{123}) = gcd(3/gcd(3, c_1), c_1/(gcd(3, c_1) gcd(2, c_1)))$$
$$= gcd(3/gcd(3, c_1), c_1/gcd(3, c_1)) = 1.$$

The homology of D is thus, again, \mathbb{Z}/g_{123} . The full Tor group follows.

Despite the increased complexity of the K-homology rings of ΩE_7 and ΩE_8 , the Tor calculations in these cases are no more elaborate than for the other exceptional groups. The presentation in Proposition 8.2.1 suggests an appropriate Tate resolution and the Tor group over $K.\Omega E_7$ is given by the homology of the DGA

$$(\mathbb{Z}\langle T_1, \dots, T_{10} \rangle \{S_1, S_2, S_3\}; dT_i = c_i, dS_1 = (c_1 - 1)T_1 - 2T_2,$$

 $dS_2 = (c_1 - 2)T_2 - 3T_3, dS_3 = (c_2 + 1)T_2 - (c_2 + c_3)T_1 - 2T_4).$

Using the F_4 computation, we see that the homology of the $\{T_1, T_2, T_3, T_4, S_1, S_2\}$ subalgebra is $\mathbb{Z}/g_{1234}\langle y_4 \rangle$, where y_4 is $-(g_{123}/g_{1234})T_4$ modulo lower terms. The homology of the $\{T_1, T_2, T_3, T_4, S_1, S_2, S_3\}$ subalgebra is therefore the homology of

$$(\mathbb{Z}/g_{1234}\langle y_4\rangle \{S_3\}; dS_3 = (2g_{1234}/g_{123})y_4).$$

We observe that $2g_{1234}/g_{123}$ is a unit in \mathbb{Z}/g_{1234} and so the homology of this subalgebra is \mathbb{Z}/g_{1234} concentrated in degree zero. The full Tor group is finally $\mathbb{Z}/g_{1..10}\langle y_5, y_6, \ldots, y_{10}\rangle$. In this calculation it is critical that the third relation in the presentation of $K.\Omega E_7$ is b(b+1) - a(b+c) - 2d and not the expected c(a-3) - 4d. The latter relation would produce a differential $dS_3 = (4g_{1234}/g_{123})y_4$ and thereby (because $4g_{1234}/g_{123}$ is not always a unit in \mathbb{Z}/g_{1234}) a plethora of nontrivial higher torsion.

The Tor computation for E_8 is entirely analogous. The Tate resolution is dictated by the presentation in Proposition 8.2.1 and the necessary combinatorial fact is that $5g_{12345}/g_{1234}$ is a unit in \mathbb{Z}/g_{12345} .

8.3 Proof of Theorem 6.2.1

We can now establish the bulk of our main theorem. We assume the computation of the torsion group for Spin(n), which is carried out in section 9.4:

$$\operatorname{Tor}^{K.\Omega Spin(n)}(K.*, (K.*)_{\tau}) = \Lambda[x_1, \dots, x_{n-1}] \otimes \mathbb{Z}/c(Spin(n), k).$$

Though we have treated K-homology as $\mathbb{Z}/2$ -graded in our Tor computations, properly it is \mathbb{Z} -graded, and the E^2 term of the Rothenberg-Steenrod spectral sequence has the appearance:

$$\begin{array}{cccc} \operatorname{Tor}_{0}^{K_{0}\Omega G}(\mathbb{Z},\mathbb{Z}_{\tau}) & \operatorname{Tor}_{1}^{K_{0}\Omega G}(\mathbb{Z},\mathbb{Z}_{\tau}) & \operatorname{Tor}_{2}^{K_{0}\Omega G}(\mathbb{Z},\mathbb{Z}_{\tau}) & \cdots \\ & 0 & 0 & 0 \\ \operatorname{Tor}_{0}^{K_{0}\Omega G}(\mathbb{Z},\mathbb{Z}_{\tau}) & \operatorname{Tor}_{1}^{K_{0}\Omega G}(\mathbb{Z},\mathbb{Z}_{\tau}) & \operatorname{Tor}_{2}^{K_{0}\Omega G}(\mathbb{Z},\mathbb{Z}_{\tau}) & \cdots \\ & 0 & 0 & 0 \\ \operatorname{Tor}_{0}^{K_{0}\Omega G}(\mathbb{Z},\mathbb{Z}_{\tau}) & \operatorname{Tor}_{1}^{K_{0}\Omega G}(\mathbb{Z},\mathbb{Z}_{\tau}) & \operatorname{Tor}_{2}^{K_{0}\Omega G}(\mathbb{Z},\mathbb{Z}_{\tau}) & \cdots \end{array}$$

In our cases these torsion groups are generated in Tor-degree 1; the (homological) differentials vanish on the generators and thus the spectral sequence collapses at the E^2 term.

We show that there are no additive extensions. We have established that the E^{∞} term of the spectral sequence is a cyclic group, say \mathbb{Z}/c , tensor an exterior algebra. The filtration is homological, so the subgroup $(\mathbb{Z}/c)\{1\} \subset E^{\infty}$ generated by the identity element of the torsion group $\operatorname{Tor}^{K,\Omega G} = E^2 = E^{\infty}$ is actually a subgroup of the K-homology $K^{\tau}(G)$. The construction of the spectral sequence shows that this identity element in the torsion group corresponds to the identity element in the K-homology. The identity element $1 \in K^{\tau}(G)$ is therefore killed by multiplication by c, and so the entire K-homology ring is c-torsion, as desired.

For degree reasons, the only possible multiplicative extension is $y_i^2 = d \in K^{\tau}(G)$; that is the square of the K-homology class represented by an exterior generator $y_i \in \text{Tor}_1$ could be a constant integer, an element of Tor_0 . However, by construction the exterior classes y_i are represented by *reduced* classes in $K^{\tau}(G)$ and so their squares are also certainly reduced, eliminating the possibility of multiplicative extensions. \Box

Chapter 9

Generating Varieties, the Cyclic Order of $K^{\tau}G$, and $\operatorname{Tor}^{K.\Omega Spin(n)}(\mathbb{Z}, \mathbb{Z}_{\tau})$

The twisted K-homology of a simple Lie group is an exterior algebra tensor a cyclic group. The order of this cyclic group depends on the twisting class and is, as yet, determined by a mysterious set of constants. We will see that this cyclic order of the twisted K-homology $K^{\tau}G$ is the greatest common divisor of the dimensions of a particular set of representations of G. The main ingredient in computing the cyclic order is a detailed understanding of the twisted module structure of \mathbb{Z}_{τ} , that is of the twisting map $K.\Omega G \xrightarrow{\tau(k)} K.*$. Bott's theory of generating varieties allows us to produce explicit representatives of classes in $K.\Omega G$, as fundamental classes of complex algebraic varieties, and thereby to describe the twisting map.

9.1 Generating Varieties and Holomorphic Induction

9.1.1 Bott Generating Varieties

A generating variety for ΩG is, for us, a space V and a map $i: V \to \Omega G$ such that the images $i_*(H.V)$ and $i_*(K.V)$ of the homology and K-theory of V generate $H.\Omega'G$ and $K.\Omega'G$, respectively, as algebras, where $\Omega'G$ is the identity component of ΩG . In [6], Bott produced a beautiful, systematic family of generating varieties of the form G/H, as we now describe; these particular homogeneous spaces are better known as coadjoint orbits and as such are smooth complex algebraic varieties with an even-dimensional cell decomposition.

We briefly review Bott's construction. Let G be a compact and connected but not necessarily simply connected Lie group. Denote by $\Gamma_G = \ker(\exp : \mathfrak{t} \to T)$ the coweight lattice of G; we do not distinguish between a coweight and the corresponding circle in G. A coweight $\ell \in \Gamma_G$ is called generating if for every root $r \in \mathfrak{t}^*$ of G, there is an element w of the Weyl group such that $r(w \cdot \ell) = 1$. Note that the coweight lattice Γ_G of the group is contained in the coweight lattice Γ_W of the Lie algebra, which is also the coweight lattice of the adjoint form of G. A coweight is generating if, roughly, it is as short as possible in Γ_W ; thus, even if a group does not have a generating coweight, its adjoint form will. The simple rank 2 groups with generating coweights, namely PSU(3), PSp(2), and G_2 , are illustrated in Figure 9-1.

Figure 9-1: Generating Coweights for Rank 2 Lie Groups



Suppose $\ell \in \Gamma_G$ is a generating coweight for G, and let $C(\ell) \subset G$ denote the pointwise centralizer of the corresponding circle; note that $C(\ell)$ can also be described as the image under the exponential map of the subalgebra of \mathfrak{g} generated by the root spaces associated to roots r perpendicular to ℓ , that is to roots where $r(\ell) = 0$. The map

$$G \to \Omega G$$
$$g \mapsto g\ell g^{-1}\ell^{-1}$$

descends to a map on cosets $G/C(\ell) \to \Omega G$. The main theorem, which is due to Bott in homology and to Clarke [7] in K-theory, is that $G/C(\ell)$ is a generating variety for ΩG .

Suppose G is simply connected and ℓ is a generating circle for PG, the adjoint form of G. Then $PG/C_{PG}(\ell) = G/C_G(\tilde{\ell})$ where $\tilde{\ell}$ denotes a loop in G covering ℓ . The composite

$$G/C_G(\ell) = PG/C_{PG}(\ell) \to \Omega'PG = \Omega G$$

is therefore a generating variety for ΩG . For example, the generating varieties corresponding to the marked coweights in Figure 9-1 are SU(3)/U(2), Sp(2)/U(2), and $G_2/U(2)$ respectively. In general there may be more than one Bott generating variety for a group; we list a Bott generating variety for each of the classical groups in the following table:

Group	Generating Variety
SU(n+1)	SU(n+1)/U(n)
Spin(2n+1)	$\int Spin(2n+1)/(Spin(2n-1) \times_{\mathbb{Z}/2} Spin(2))$
Sp(n)	Sp(n)/U(n)
Spin(2n)	$Spin(2n)/(Spin(2n-2) \times_{\mathbb{Z}/2} Spin(2))$

Here the $\mathbb{Z}/2$ action on Spin(n) is the one whose quotient is SO(n).

We need to compute the twisted map $K.\Omega G \xrightarrow{\tau(k)} K.*$. To this end we want to represent the algebra generators of $K.\Omega G$ in a way that allows us to compute their twisted images. In our computations we utilize generating varieties to represent

 $\sim \sim \sim \sim \sim$

algebra generators in three independent ways; we refer to these briefly as representing them via subvarieties, via an evaluation dual basis, and via a Poincare dual basis.

In some cases we have a sufficiently explicit handle on the generating variety V for ΩG that we can describe a collection of maps $W_i \to V \to \Omega G$ such that W_i is a Koriented manifold and the images in $K.\Omega G$ of the K-homology fundamental classes of
the W_i are the desired algebra generators; frequently, though not always, the W_i are
subvarieties of the generating variety V. Variants of this 'subvariety' representation
are used for SU(n), G_2 , and Sp(n) in sections 9.2 and 9.3.

The K-cohomology of the Bott generating variety V is easily determined from the representation theory of G. Specifically, if the Bott generating variety V is the quotient G/H with G simply connected, then $K^{\cdot}V = R[H]/i^*(I[G])$, where $i : H \to G$ is the inclusion and I[G] is the augmentation ideal of the representation ring R[G]. If there is a minor miracle and we can write down a clean basis for this ring, then we can take an evaluation dual basis for the K-homology K.V; the image of this basis in $K.\Omega G$ will generate as an algebra and the twisting map will be easily computable. This is the approach taken for Spin(n) in section 9.4.

More commonly, any apparent basis for the K-cohomology of the generating variety is quite haphazard. In this case we consider the Poincare dual basis (to some chosen basis) for the K-homology K.V. Again the images of these classes in $K.\Omega G$ will generate, but computing their twisted images requires a bit more work. Specifically, we use holomorphic induction to write these images in terms of the dimensions of irreducible representations of G, as described in detail in the next section. This Poincare dual approach is the one that provides a general procedure and is the subject of section 9.5.

9.1.2 The Twisting Map via Holomorphic Induction

In section 7.2 we described the $K.\Omega G$ -module structure on \mathbb{Z}_{τ} by the twisting map

$$K.\Omega G \xrightarrow{\tau(k)} K.*$$
$$x \mapsto \langle L^k, x \rangle,$$

where $L \in K^{*}\Omega G$ is a generating line bundle. The purpose of this section is to outline the computation of the twisted image $\langle L^{k}, x \rangle$ when x is represented as the image of the Poincare dual of a bundle on some K-oriented manifold.

Let $i: W \to \Omega G$ be a map from a K-oriented manifold W to ΩG and let $\eta \in K^{\cdot}W$ be a bundle on W such that $i_*(D\eta) = x \in K.\Omega G$; here, D denotes the Poincare duality map. We first translate the evaluation $\langle L^k, x \rangle$ into a pushforward on W:

$$\langle L^k, i_*(\mathrm{D}\eta) \rangle = \langle i^*(L^k), \mathrm{D}\eta \rangle = \langle i^*(L^k), \eta \cap [W] \rangle = \langle i^*(L^k) \cup \eta, [W] \rangle.$$

The third equality is simply

$$\langle i^{*}(L^{k}), \eta \cap [W] \rangle = \pi_{*}(i^{*}(L^{k}) \cap (\eta \cap [W])) = \pi_{*}((i^{*}(L^{k}) \cup \eta) \cap [W]) = \langle i^{*}(L^{k}) \cup \eta, [W] \rangle,$$

where $\pi: W \to *$ denotes projection. We are thereby reduced to computing K-theory pushforwards $\langle \mu, [W] \rangle = \pi_!(\mu)$.

We can translate this K-theory pushforward to a cohomology pushforward using the Chern character. Multiplicative transformations of cohomology theories do not commute with pushforward, and the discrepancy in the case of the Chern character is the Todd genus. The Chern character is the identity on a point, and so we have

$$\langle \mu, [W]_K \rangle = \operatorname{ch}(\pi_!^K(\mu)) = \pi_!^H(\operatorname{ch}(\mu) \cup \operatorname{Td}(W)) = \langle \operatorname{ch}(\mu) \cup \operatorname{Td}(W), [W]_H \rangle.$$

Next we reduce this cohomology pushforward to a calculation in sheaf cohomology. We always work with manifolds W that are smooth projective complex algebraic varieties, and as such the Hirzebruch-Riemann-Roch theorem applies:

$$\langle \operatorname{ch}(\mu) \cup \operatorname{Td}(W), [W]_H \rangle = \Sigma(-1)^i H^i(W; \mu) =: \chi(\mu).$$

Here of course $H^i(W; \mu)$ denotes the cohomology of the sheaf of holomorphic sections of the bundle μ .

In some cases, this reduction is sufficient, as we can use Serre duality and related techniques to compute the sheaf cohomology groups. If, as in our computations it always is, W is a homogeneous space G/H which is a Kähler manifold, and μ is pulled back along $G \to BH$ from an irreducible representation of H, then we can take advantage of Bott's theory of holomorphic induction [5]. (If our original μ is not irreducible, we simply decompose it into irreducible components and apply holomorphic induction to each component.) To avoid reviewing the whole of Bott's theory, we describe holomorphic induction procedurally as it will arise in our computations.

Let G be a compact connected simply connected Lie group and H a centralizer of a circle in G; in particular H and G share a maximal torus and their weight lattices coincide. In this situation, as remarked earlier, the K-theory of the quotient G/H is simply the quotient of representation rings: $K(G/H) = R[H]/i^*(I[G])$. We may further assume that we have chosen an order on the roots of G such that H is generated by a subset of the simple roots of G; in particular this determines standard Weyl chambers for G and H. (That there is such a choice of order is the content of Wang's theorem and depends on H being the centralizer of a torus in G; see Bott [5].) Let μ denote simultaneously a weight in the Weyl chamber of H, the corresponding irreducible representation of H, and the associated bundle on G/H. Let ρ denote half the sum of the positive roots of G, and let S denote the union of the hyperplanes perpendicular to the roots of G. Further, for a weight ω of G, let $T(\omega)$ denote the unique weight in the Weyl chamber of G that is the image of ω under an element of the Weyl group. The index $\operatorname{ind}(\omega)$ of a weight ω not in S is the number of hyperplanes of S intersecting a straight line connecting ω and $T(\omega)$. Bott's theorem is the following:

Theorem 9.1.1 (Bott [5]). In the above situation,

- if $\mu + \rho \in S$ then $H'(G/H, \mu) = 0$;
- if $\mu + \rho \notin S$ then $H(G/H, \mu)$ is nonzero only in dimension $ind(\mu + \rho)$;
- in this case, $H^{ind(\mu+\rho)}(G/H,\mu)$ is isomorphic to the irreducible representation of G with highest weight $T(\mu+\rho) - \rho$.

When $\mu + \rho \in S$ we say that μ is singular. Thus, when μ is singular, $\chi(\mu) = 0$, and otherwise

$$\chi(\mu) = (-1)^{\operatorname{ind}(\mu+\rho)} \dim([T(\mu+\rho)-\rho]_G),$$

where $[-]_G$ denotes the irreducible representation of G with the specified highest weight.

In any given case, the dimension of this irreducible representation is easily computed using the Weyl character formula, and so the preceding method provides a systematic approach to computing the twisting map on a class represented by the image of the Poincare dual to a K-cohomology class of an appropriate homogeneous space. We proceed to specific examples.

9.2 Subvarieties of $\Omega SU(n+1)$ and ΩG_2

A generating variety for $\Omega SU(n+1)$ is $SU(n+1)/U(n) = \mathbb{C}P^n \xrightarrow{i} \Omega SU(n+1)$, and the induced map in homology is

$$\widetilde{H}.\mathbb{C}\mathrm{P}^n = \mathbb{Z}\{z_1, \dots, z_n\} \to \mathbb{Z}[w_1, \dots, w_n] = H.\Omega SU(n+1)$$

 $z_i \mapsto w_i.$

Here the classes z_i are represented by the fundamental homology classes of the subvarieties $\mathbb{CP}^i \subset \mathbb{CP}^n$. The Atiyah-Hirzebruch spectral sequence collapses for both $\Omega SU(n+1)$ and \mathbb{CP}^n and there are no extensions. In particular $K.\Omega SU(n+1)$ is polynomial on n generators, as previously noted, and $\tilde{K}.\mathbb{CP}^n$ is free abelian of rank n. **Lemma 9.2.1.** The set $\{[\mathbb{C}P^i]\}_{i=1}^n$ of fundamental K-homology classes of the subvarieties $\mathbb{C}P^i \subset \mathbb{C}P^n$ forms a basis for $\widetilde{K}.\mathbb{C}P^n$.

Proof. By induction it is enough to show that under the projection $K.\mathbb{CP}^i \to K.(\mathbb{CP}^i, \mathbb{CP}^{i-1}) = \mathbb{Z}$ the fundamental class of \mathbb{CP}^i maps to a generator. This follows immediately from the naturality of Poincare duality,

$$K.\mathbb{C}\mathrm{P}^{i} \longrightarrow K.(\mathbb{C}\mathrm{P}^{i}, \mathbb{C}\mathrm{P}^{i-1})$$

$$\| \qquad \|$$

$$K^{\cdot}\mathbb{C}\mathrm{P}^{i} \longrightarrow K^{\cdot}(\mathbb{C}\mathrm{P}^{i} - \mathbb{C}\mathrm{P}^{i-1}),$$

because the unit in $K^{\cdot}\mathbb{C}P^{i}$ certainly maps to a generator of $K^{\cdot}(\mathbb{C}P^{i} - \mathbb{C}P^{i-1})$. The images $i_{*}([\mathbb{C}P^{i}])$ generate $K.\Omega SU(n+1)$ as an algebra and we may therefore take $\{x_{i} = i_{*}([\mathbb{C}P^{i}]) - 1\}$ to be the reduced polynomial generators.

We now have to evaluate the pushforward $\langle L^k, [\mathbb{CP}^i] \rangle$, where we use L to denote the generating line bundle on $\mathbb{CP}^i = SU(i+1)/U(i)$; this L is of course the pullback of the generating line bundle on $\Omega SU(n+1)$. The bundle L corresponds to an irreducible representation of U(i), thus to a weight of U(i) and so a weight, also denoted L, of SU(i+1); this weight L is in the Weyl chamber of SU(i+1). The irreducible representation of SU(i+1) corresponding to L is the dual of the standard representation; (that it is the dual of the standard representation and not the standard representation is the effect of a sign choice—see the remark at the end of this section). It happens that the k-fold symmetric power of this representation is irreducible, and so the dimension of the irreducible representation corresponding to L^k is $\binom{k+i}{i}$. The image of $x_i = i_*([\mathbb{CP}^i]) - 1 \in K.\Omega SU(n+1)$ in \mathbb{Z}_{τ} is therefore $c_i = \binom{k+i}{i} - 1$, and the cyclic order of $K^{\tau}(SU(n+1))$ is

$$c(SU(n+1),k) = \gcd\left\{\binom{k+1}{1} - 1, \binom{k+2}{2} - 1, \dots, \binom{k+n}{n} - 1\right\}.$$

 $\sim \sim \sim \sim \sim$

The procedure for calculating the cyclic order of $K^{\tau}G_2$ is similar: we find fundamental class representatives for algebra generators of the homology of ΩG_2 and then
show that the corresponding K-homology fundamental classes also generate. The map $\Omega G_2 \to \mathbb{C}P^{\infty}$ classifying the generating line bundle is a homology equivalence through degree 4. Using this, the Serre spectral sequence for $\Omega SU(3) \to \Omega G_2 \to \Omega S^6$ shows that

$$H_{\cdot}\Omega G_{2} \cong \mathbb{Z}[a_{2}, a_{4}, a_{10}]/a_{2}^{2} = 2a_{4}.$$

The composition

$$\mathbb{CP}^2 \to \Omega SU(3) \to \Omega G_2 \to \mathbb{CP}^\infty$$

is simply the inclusion and as such, a_2 and a_4 in $H.\Omega G_2$ are represented respectively by the fundamental classes $[\mathbb{CP}^1]$ and $[\mathbb{CP}^2]$. The Bott generating variety for ΩG_2 is $G_2/U(2)$, where the U(2) in question is included in G_2 along a pair of complexconjugate short roots. The manifold $G_2/U(2)$ has dimension 10 and the image of its homology generates $H.\Omega G_2$; we may therefore choose a_{10} to be the image of the fundamental homology class $[G_2/U(2)]$.

The Atiyah-Hirzebruch spectral sequence for ΩG_2 collapses, and the low-degree equivalence between ΩG_2 and \mathbb{CP}^{∞} resolves the extension. The K-homology $K.\Omega G_2$ is thereby isomorphic to $\mathbb{Z}[a, b, x_3]/(a^2 + 3a = 2b)$.

Lemma 9.2.2. The reduced algebra generators of $K.\Omega G_2 \cong \mathbb{Z}[a, b, x_3]/(a^2 + 3a = 2b)$ may be taken to be the reduced fundamental K-homology classes $[\mathbb{CP}^1] - 1$, $[\mathbb{CP}^2] - 1$, and $[G_2/U(2)] - 1$ respectively.

Proof. Let $(G_2/U(2))_8$ denote the 8-skeleton of the generating variety, that is everything except the top cell. As in Lemma 9.2.1, the fundamental K-homology class of $G_2/U(2)$ maps to a generator of $K.(G_2/U(2), (G_2/U(2))_8)$. Comparing the Atiyah-Hirzebruch spectral sequences for $G_2/U(2)$ and ΩG_2 we see that $[G_2/U(2)]$ lives in filtration 10 in $K.\Omega G_2$ and projects to the generator a_{10} in $H_{10}\Omega G_2$. The fundamental K-homology classes $[\mathbb{CP}^1]$ and $[\mathbb{CP}^2]$ certainly project to the generators a_2 and a_4 respectively in the appropriate filtration quotients, and this completes the proof. \Box

We remark that these algebra generators differ by a change of basis from those implicitly chosen in section 8.2 and this explains the difference in the relation; the Tor computation and the cyclic order is not affected by the change.

We need only compute the pushforward $\langle L^k, [G_2/U(2)] \rangle$. The bundle L corresponds to the shortest weight μ perpendicular to the roots of U(2); as a weight of G_2 , μ is the long root of G_2 in the Weyl chamber. The pushforward is therefore the dimension of the irreducible representation of G_2 with highest weight $k\mu$. By the Weyl character formula (see for example [20]) this dimension is

$$\dim([k\mu]_{G_2}) = \frac{(k+1)(k+2)(2k+3)(3k+4)(3k+5)}{120}.$$

The cyclic order of $K^{\tau}G_2$ is finally

$$c(G_2,k) = \gcd\left\{\binom{k+1}{1} - 1, \binom{k+2}{2} - 1, \frac{(k+1)(k+2)(2k+3)(3k+4)(3k+5)}{120} - 1\right\}.$$

 $\sim \sim \sim \sim \sim$

A remark on signs is in order. If we have chosen a generating line bundle L on ΩG_2 a priori, the weight corresponding to L may be $-\mu$ instead of μ as claimed above. The dimension resulting from holomorphic induction on the weight $k(-\mu)$ is wildly different from the dimension associated to $k(\mu)$, and this might be cause for worry. However, the greatest common divisor is in all cases unaffected by the change. The easiest way around this ambiguity is to chose L such that the corresponding weight is μ and not $-\mu$; we must then pick the generating variety \mathbb{CP}^2 for $\Omega SU(3)$ in such a way that the given L corresponds there to the dual of the standard representation (and not to the standard representation) as described in the discussion of SU(n+1) above—this is easily accomplished. Similar remarks apply to all our computations and we make convenient sign choices without comment.

9.3 Generating Varieties for $\Omega Sp(n)$

The homology and K-homology of $\Omega Sp(n)$ are polynomial in n generators. The natural Bott generating variety for $\Omega Sp(n)$ is Sp(n)/U(n), which has homology and

K-homology of rank $n^2 + n$. Identifying the *n* elements which generate therefore requires more doing—we return to this question later. Luckily, $\Omega Sp(n)$ has smaller generating varieties—see [24, 39]; in particular we work with $(\mathbb{CP}^{2n-2})^{L^2}$, the Thom complex of the square of the tautological bundle.

Let $P_i(V)$ or P(V) denote the projectivization of the bundle V on \mathbb{CP}^i ; note that we can rewrite our generating variety $V(n) = (\mathbb{CP}^{2n-2})^{L^2}$ as $P_{2n-2}(L^2+1)/P_{2n-2}(L^2)$. We think of the quotient map $P(L^2+1) \rightarrow P(L^2+1)/P(L^2)$ as a resolution of our (quite singular) generating variety, and we represent homology and K-homology classes in V(n) (and thus in $\Omega Sp(n)$) as the images of fundamental classes of subvarieties of $P(L^2+1)$. The reduced homology of V(n) is free of rank one in each even degree between 2 and 4n - 2, and the degree 2i group is generated by the image of the fundamental class $[P_{i-1}(L^2+1)]$. In particular, the algebra generators $\{a_{4i-2}\}$ of $H.\Omega Sp(n) = \mathbb{Z}[a_2, a_6, a_{10}, \ldots, a_{4n-2}]$ are represented by the fundamental classes $[P_{2(i-1)}(L^2+1)]$, for $1 \leq i \leq n$.

The K-homology situation is the same.

Lemma 9.3.1. The reduced polynomial generators of the K-homology $K.\Omega Sp(n) \cong \mathbb{Z}[x_1, \ldots, x_n]$ can be taken to be the reduced K-homology fundamental classes $f_*[P_{2(i-1)}(L^2+1)] - 1$, $1 \leq i \leq n$; here f is the composite

$$P_{2(i-1)}(L^2+1) \to P_{2(n-2)}(L^2+1) \to P_{2(n-2)}(L^2+1)/P_{2(n-2)}(L^2) \to \Omega Sp(n).$$

The K-homology fundamental classes map, in the appropriate filtration quotients, to the homology fundamental classes; the proof is the same as for SU(n+1) and G_2 .

To evaluate the twisting map, specifically to calculate $\langle L^k, f_*[P_{2(i-1)}(L^2+1)]\rangle$, we need to identify the bundle $f^*(L^k)$. We do this by writing down a bundle on $P(L^2+1) = P_{2(i-1)}(L^2+1)$ that is trivial on $P(L^2) = P_{2(i-1)}(L^2)$, and show that the corresponding bundle on the quotient V(i) is the pullback $f'^*(L)$ where f' is the inclusion $V(i) \to V(n) \to \Omega Sp(n)$. Let γ be the tautological bundle on the total space $P(L^2+1)$ and let $\pi : P(L^2+1) \to \mathbb{CP}^{2(i-1)}$ be the bundle projection. The subspace $P(L^2)$ is of course just the base $\mathbb{CP}^{2(i-1)}$ and so γ restricts to $\pi^*(L^2)|_{P(L^2)}$ on $P(L^2)$. In particular then, the bundle $\gamma \otimes \pi^*(L^{-2})$ is trivial on the subspace $P(L^2)$ and so pulls back from a bundle ϕ on V(i). To see that ϕ is equal to $f'^*(L)$ (up to our usual sign ambiguity), and therefore that $\gamma \otimes \pi^*(L^{-2}) = f^*(L)$, it is enough to check that the first Chern class of $\gamma \otimes \pi^*(L^{-2})$ is a module generator of $H^2(P(L^2+1)) = \mathbb{Z}\{c_1(\gamma), \pi^*(c_1(L))\}$; this much is clear.

We now compute the pushforward

$$\langle L^k, f_*[P_{2(i-1)}(L^2+1)] \rangle = \langle (\gamma \otimes \pi^*(L^{-2}))^k, [P_{2(i-1)}(L^2+1)] \rangle.$$

First pushforward along the fibres:

$$\langle \gamma^k \otimes \pi^*(L^{-2k}), [P_{2(i-1)}(L^2+1)] \rangle = \langle \operatorname{Sym}^k(L^2+1) \otimes L^{-2k}, [\mathbb{CP}^{2(i-1)}] \rangle.$$

This is a parameterized version of the pushforward

$$\langle \gamma_{\text{taut}}^k, [\mathbb{C}\mathbb{P}^n] \rangle = \langle \gamma_{\text{taut}}^k, [P(\mathbb{C}^{n+1})] \rangle = \dim(\operatorname{Sym}^k(\mathbb{C}^{n+1})) = \binom{k+n}{k}$$

used in the preceding section. Next

$$\operatorname{Sym}^{k}(L^{2}+1) \otimes L^{-2k} = L^{-2k} + L^{-2k+2} + \ldots + 1$$

and so

$$\langle \operatorname{Sym}^{k}(L^{2}+1)\otimes L^{-2k}, [\mathbb{CP}^{2(i-1)}] \rangle = \binom{-2k+2(i-1)}{2(i-1)} + \binom{-2k+2+2(i-1)}{2(i-1)} + \dots + 1.$$

Here we use $\binom{a+b}{b}$ to denote $\frac{(a+b)(a+b-1)\dots(a)}{(b)(b-1)\dots(1)}$ even when a+b is negative; we implicitly observe that this expression does give the correct pushforward even when the bundle, as in the case of L^{-2k+2l} , corresponds to a weight that is not in the Weyl chamber of

SU(2(i-1)+1). This finishes our calculation of the cyclic order of $K^{\tau}Sp(n)$:

$$c(Sp(n),k) = \gcd\left\{\sum_{-k \le j \le -1} \binom{2j+2(i-1)}{2(i-1)} : 1 \le i \le n\right\}.$$

It would be more natural to express the cyclic order of $K^{\tau}Sp(n)$ in terms of the dimensions of irreducible representations of symplectic groups. This is possible if we work with subvarieties of the Bott generating variety V = Sp(n)/U(n). There is a natural collection of n subvarieties of V, namely $\{Sp(i)/U(i)\}$. It is not the case that the fundamental homology classes of these subvarieties represent algebra generators for $H.\Omega Sp(n)$; indeed, the algebra generators are in dimensions $\{4i - 2\}$, while these subvarieties have dimensions $\{i^2+2\}$. It is therefore remarkable that the K-homology of $\Omega Sp(n)$.

Conjecture. The K-homology ring $K \Omega Sp(n)$ is polynomial on the classes represented by the reduced K-homology fundamental classes [Sp(i)/U(i)] - 1, for $1 \le i \le n$.

Using the Weyl character formula, this immediately gives a description of the cyclic order:

$$c(Sp(n),k) = \gcd\left\{\frac{\left(\prod_{1 \le j < l \le i} (l-j)(2k+2i+2-(j+l))\right)\left(\prod_{1 \le j \le i} (k+i+1-j)\right)}{(2i-1)!(2i-3)!\dots 3!1!} : 1 \le i \le n\right\}.$$

These gcd's agree with those determined using the generating variety $(\mathbb{CP}^{2n-2})^{L^2}$.

9.4 The Tor Calculation for Spin(2n+1)

We now pay our debt to the proof of Theorem 6.2.1 by calculating $\operatorname{Tor}^{K.\Omega Spin(2n+1)}(\mathbb{Z}, \mathbb{Z}_{\tau})$; in the process we determine the cyclic order of $K^{\tau}Spin(2n+1)$. (The computations for the even *Spin* groups are similar and the resulting cyclic order is recorded in Theorem 6.2.2.) For the other simple groups, we were able to calculate the Tor group without knowing the map $K.\Omega G \rightarrow (K.*)_{\tau}$ and we determined, after the fact, the structure of this twisting map. The ring $K.\Omega Spin(2n + 1)$ is too complicated to permit this a-priori Tor calculation; we must first identify algebra generators of $K.\Omega Spin(2n + 1)$ and compute the twisting map. It happens that the K-cohomology of the Bott generating variety $V = Spin(2n + 1)/(Spin(2n - 1) \times_{\mathbb{Z}/2} Spin(2))$ admits a particularly simple representation-theoretic basis, and an evaluation dual basis maps to a set of algebra generators in $K.\Omega Spin(2n + 1)$. Once we know the twisted pushforwards of these algebra generators, the Tor computation becomes tractable.

The structure of the K-homology ring of $\Omega Spin(2n+1)$ was described by Clarke [7]:

$$K.\Omega Spin(2n+1) = \frac{\mathbb{Z}[\sigma_1, \sigma_2, \dots, \sigma_{n-1}, 2\sigma_n, 2\sigma_{n+1} + \sigma_n, \dots, 2\sigma_{2n-1} + \sigma_{2n-2}]}{(\rho_1, \dots, \rho_{n-1})},$$
$$\rho_k = \sigma_k^2 + \sum_{i=0}^{k-1} (-1)^{k-i} \sigma_i \sum_{j=k}^{2k-i-1} \binom{k-i-1}{j-k} (2\sigma_{j+1} + \sigma_j).$$

One can see why the a-priori Tor calculation is unlikely to be fruitful. The Kcohomology of the Bott generating variety is simply the quotient of the representation ring of $Spin(2n-1) \times_{\mathbb{Z}/2} Spin(2)$ by the image of the augmentation ideal of the representation ring of Spin(2n+1). Clarke writes this quotient in a convenient form:

$$KV = \mathbb{Z}[\mu, \gamma]/(\mu^n - 2\gamma - \mu\gamma, \gamma^2);$$

here $\mu = L - 1$ where L is the generating line bundle whose k-th power determines the twisting. Note that $\mu^{2n} = 0$ in this ring, and so $(\mu, \mu^2, \dots, \mu^{2n-1})$ is a basis for $K^{\cdot}(V) \otimes \mathbb{Q}$. Letting $(\sigma'_1, \dots, \sigma'_{2n-1})$ be the evaluation dual basis of $K_{\cdot}(V) \otimes \mathbb{Q}$, we see that

$$(\sigma'_1, \sigma'_2, \dots, \sigma'_{n-1}, 2\sigma'_n, 2\sigma'_{n+1} + \sigma'_n, \dots, 2\sigma'_{2n-1} + \sigma'_{2n-2})$$

is a basis for K.V; these elements map, respectively, to the given generators of $K.\Omega Spin(2n + 1)$. The twisting map $K.\Omega Spin(2n + 1) \rightarrow (K.*)_{\tau}$ takes a genera-

tor g to $\langle L^k, g \rangle \in \mathbb{Z}$. Because $\mu^{2n} = 0$, we have

$$\langle L^k, \sigma'_i \rangle = \langle (\mu+1)^k, \sigma'_i \rangle = \binom{k}{i},$$

and the images of our integral generators are respectively

$$\left(\binom{k}{1},\binom{k}{2},\ldots,\binom{k}{n-1},\binom{k}{n},\binom{k}{n},\binom{k}{n+1}+\binom{k}{n},\ldots,\binom{k}{2n-1}+\binom{k}{2n-2}\right).$$

We can now prove that Tor^{$K.\Omega Spin(2n+1)$} ($\mathbb{Z}, \mathbb{Z}_{\tau}$) is an exterior algebra on n-1 generators tensor a cyclic group. We first rewrite the above presentation of $K.\Omega Spin(2n+1)$ in a way that suggests a propitious choice of Tate resolution. Let (a_1, \ldots, a_{2n-1}) denote the given generators of $K.\Omega Spin(2n+1)$. For *i* sufficiently large, the relation ρ_i expresses the generator a_{2i} in lower terms; in particular

$$K.\Omega Spin(4n-1) = \frac{\mathbb{Z}[a_1, a_2, \dots, a_{2n-2}, a_{2n-1}, a_{2n+1}, a_{2n+3}, \dots, a_{4n-5}, a_{4n-3}]}{(\rho_1, \rho_2, \dots, \rho_{n-1})},$$

$$K.\Omega Spin(4n+1) = \frac{\mathbb{Z}[a_1, a_2, \dots, a_{2n-2}, a_{2n-1}, a_{2n+1}, a_{2n+3}, \dots, a_{4n-3}, a_{4n-1}]}{(\rho_1, \rho_2, \dots, \rho_{n-1})}.$$

The remaining relations can be written

$$\rho_i = 2a_{2i} + r_i a_{2i-1} + \dots,$$

with r_i odd and all unspecified monomials containing some a_j with j < 2i - 1, except for ρ_1 which is $2a_2 + a_1 - a_1^2$. If we can show that Tor over the subring $R_n = \mathbb{Z}[a_1, \ldots, a_{2n-2}]/(\rho_1, \ldots, \rho_{n-1})$ is exterior on n-2 generators, the desired result follows. Rather than presenting the general induction immediately, we discuss the first few cases explicitly.

The case n = 1 corresponding to Spin(3) requires no comment. The ring $K.\Omega Spin(7)$ is $\mathbb{Z}[a_1, a_2, a_3, a_5]/(2a_2+a_1-a_1^2)$. This is reminiscent of $K.\Omega G_2$ and indeed the Atiyah-Hirzebruch spectral sequence for the fibration $\Omega G_2 \rightarrow \Omega Spin(7) \rightarrow \Omega S^7$ collapses; there are no possible multiplicative extensions and so this confirms that $K.\Omega Spin(7)$ is $K.\Omega G_2$ adjoin a generator in degree 6. As in section 8.2, the Tor group in question is

$$\operatorname{Tor}^{R_2}(\mathbb{Z}, \mathbb{Z}_{\tau}) = \operatorname{Tor}^{\mathbb{Z}[a_1, a_2]/(2a_2 + a_1 - a_1^2)}(\mathbb{Z}, \mathbb{Z}_{\tau}) = \mathbb{Z}/g_{12}$$

(Note that the generator a_i of the subring R_n has image under the twisting map $c_i = \binom{k}{i}$ and as before we abbreviate $gcd\{c_1, c_2, \ldots, c_i\}$ by $g_{1..i}$.)

The relevant subring of $K.\Omega Spin(11)$ is

$$R_3 = \mathbb{Z}[a_1, a_2, a_3, a_4] / (\rho_1, 2a_4 + 3a_3 + (a_2 + 1)a_2 + (-2a_3 - a_2)a_1).$$

This presentation suggests the Tate resolution

Tor^{R₃} =
$$H(\mathbb{Z}\langle T_1, T_2, T_3, T_4\rangle \{S_1, S_2\};$$

 $dT_i = c_i, dS_1 = 2T_2 + (1 - c_1)T_1, dS_2 = 2T_4 + 3T_3 + (c_2 + 1)T_2 + (-2c_3 - c_2)T_1).$

The E_1 term of the spectral sequence associated to the filtration of this complex by S_2 is

$$egin{array}{cccc} \mathbb{Z}/g & \mathbb{Z}/g \oplus \mathbb{Z}/g & \mathbb{Z}/g \\ \mathbb{Z}/g & \mathbb{Z}/g \oplus \mathbb{Z}/g & \mathbb{Z}/g \\ \mathbb{Z}/g & \mathbb{Z}/g \end{array}$$

where $g = g_{1234}$ and the generator in degree (1, 1) is S_2 . At first blush the generators in degree (0, 1) have the form $t_3 = (g_{12}/g_{123})T_3 + O(2)$ and $t_4 = (g_{123}/g_{1234})T_4 + O(3)$, where the omitted terms contain only terms involving $(T_2 \text{ and } T_1)$ and $(T_3, T_2, \text{ and } T_1)$ respectively. In order to determine the differential on S_2 we need control over the T_3 term in the generator t_4 . The basic observation is that if there exists a cocycle t'_4 of the form $(g_{123}/g_{1234})T_4 + O(2)$, then some linear combination $t_4 + ct_3$ is cohomologous to t'_4 and so we may take the generators in degree (0, 1) to be t'_4 and t_3 . The existence of this cocycle is ensured by the fact that $(g_{123}/g_{1234})g_4$ is divisible by g_{12} , as is easily checked. The differential on S_2 is therefore $(2g_{1234}/g_{123})t'_4 + (3g_{123}/g_{12})t_3$. Because the greatest common divisor of $2g_{1234}/g_{123}$ and $3g_{123}/g_{12}$ is always 1, the torsion group is finally

$$\operatorname{Tor}^{R_3} = \mathbb{Z}/g_{1234} \langle x_4 \rangle;$$

here we can choose the generator x_4 to be $(g_{123}/g_{1234})T_4 + O(3)$.

The case of Spin(15) proceeds similarly. The relevant subring of $K.\Omega Spin(15)$ is

$$R_4 = \mathbb{Z}[a_1, a_2, a_3, a_4, a_5, a_6] / (\rho_1, \rho_2, 2a_6 + 5a_5 + 4a_4 + O(3)),$$

and we take the corresponding Tate resolution. Filtering by S_3 we have the spectral sequence (here condensed)

$$\mathbb{Z}/g\langle x_4, x_5, x_6 \rangle \qquad \mathbb{Z}/g\langle x_4, x_5, x_6 \rangle \qquad \mathbb{Z}/g\langle x_4, x_5, x_6 \rangle \qquad \cdots$$

The torsion g is $g_{1..6}$ and the generators in degree (0, 1) are

$$x_4 = (g_{123}/g_{1234})T_4 + O(3)$$

$$x_5 = (g_{1234}/g_{1..5})T_5 + O(4)$$

$$x_6 = (g_{1..5}/g_{1..6})T_6 + O(5).$$

It happens that $(g_{1234}/g_{1..5})g_5$ and $(g_{1..5}/g_{1..6})g_6$ are both divisible by g_{123} ; we can therefore adjust our generators so that they are

$$x_4 = (g_{123}/g_{1234})T_4 + O(3)$$

$$x_5 = (g_{1234}/g_{1..5})T_5 + O(3)$$

$$x_6 = (g_{1..5}/g_{1..6})T_6 + O(3).$$

The differential on S_3 is thus $(2g_{1..6}/g_{1..5})x_6 + (5g_{1..5}/g_{1234})x_5 + (4g_{1234}/g_{123})x_4$. Because $2g_{1..6}/g_{1..5}$ and $5g_{1..5}/g_{1234}$ are relatively prime, there exist constants z_1 and z_2 so that

if we set

$$y_6 = x_6 + z_1 x_4 = g_{1..5}/g_{1..6}T_6 + O(4)$$

$$y_5 = x_5 + z_2 x_4 = g_{1234}/g_{1..5}T_5 + O(4),$$

then $\{dS_3, y_6, y_5\}$ forms a basis for the degree (0, 1) group. Finally, then, the Tor group is

$$\operatorname{Tor}^{R_4} = \mathbb{Z}/g_{1..6}\langle y_5, y_6 \rangle$$

as desired.

The general case is now clear. Suppose we know that

Tor^{$$R_n$$} = $\mathbb{Z}/g_{1..(2n-2)}\langle x_{n+1},\ldots,x_{2n-2}\rangle$,

where $x_i = (g_{1..(i-1)}/g_{1..i})T_i + O(i-1)$. The ring R_{n+1} has two additional generators a_{2n-1} and a_{2n} and one additional relation ρ_n . Filter the appropriate Tate resolution by powers of S_n , then adjust the generators of the degree (0, 1) group in the spectral sequence so that the single generator x_{2n} involving T_{2n} does not contain any terms involving T_{2n-1} . This is possible because $g_{1..(2n-2)}$ divides $(g_{1..(2n-1)}/g_{1..(2n)})g_{2n}$. The differential of S_n then has the form

$$dS_n = (2g_{1..(2n)}/g_{1..(2n-1)})x_{2n} + (rg_{1..(2n-1)}/g_{1..(2n-2)})x_{2n-1} + \dots$$

As those two leading terms are relatively prime, this ensures that $\operatorname{Tor}^{R_{n+1}}$ again has the desired form. Note that in theory there could be multiplicative extensions in the filtration spectral sequence calculating the Tor group, but the above procedure gives a sufficiently explicit handle on the generating classes as to eliminate this possibility.

This completes the proof of Theorem 6.2.1 for the odd *Spin* groups and also establishes the odd *Spin* cyclic orders given in Theorem 6.2.2. The Tor calculation for the even *Spin* groups is analogous and the resulting cyclic orders are also given in Theorem 6.2.2.

9.5 Poincare-Dual Bases and the Cyclic Order of $K^{\tau}G$

We describe a general procedure for computing the cyclic order of $K^{\tau}G$ for any simple G and illustrate the method with the group G_2 .

Let V = G/H denote the Bott generating variety for ΩG ; recall that the Kcohomology of V is $R[H]/i^*I[G]$ where $i: H \to G$ is the inclusion. Pick a module basis $\{w_i\}$ for this ring and consider the Poincare-dual basis $\{Dw_i\}$ of K.V. The image of this basis in $K.\Omega G$, which we also denote by $\{Dw_i\}$, is a set of algebra generators for $K.\Omega G$. Note that for any set $\{y_i\}$ of algebra generators for $K.\Omega G$, the cyclic order of $K^{\tau}G$ is given by $gcd\{\tau_k(y_i) - \tau_0(y_i)\}$, where τ_k and τ_0 are respectively the twisted and untwisted maps from $K \Omega G$ to K *. In section 9.1.2 we saw that $\tau_k(Dw_i) = \langle L^k \cup w_i, [V] \rangle$ where L denotes the generating line bundle on V. Decompose w_i into a sum of irreducible representations $\sum v_{ij}$, and let $h(v_{ij})$ denote the highest weight corresponding to v_{ij} . The product $L^k \cup v_{ij}$ is again irreducible, with highest weight $kL+h(v_{ij})$, and Bott's theorem (9.1.1) therefore applies: the pushforward $\langle L^k \cup$ $|v_{ij}, [V]\rangle$ is either 0 or is (plus or minus) the dimension of the irreducible representation of G with highest weight $T(kL + h(v_{ij}) + \rho) - \rho$, where T reflects a weight into the fundamental Weyl chamber. This procedure expresses the cyclic order of $K^{\tau}G$ as the greatest common divisor of a finite set of differences of dimensions of irreducible representations of G.

Recall that the Bott generating variety for G_2 is $G_2/U(2)$ for the short-root inclusion of U(2). Let a and b denote the fundamental weights of G_2 corresponding to the 7 and 14 dimensional representations; in particular $R[G_2] = \mathbb{Z}[a, b]$. Similarly $R[U(2)] = \mathbb{Z}[f, t, t^{-1}]$, where f and t are respectively the standard representation and the determinant representation. The restriction map is

$$i^{*}(a) = f + f^{2}t^{-1} - 1 + ft^{-1}$$

$$i^{*}(b) = t + f^{3}t^{-1} - 2f + f^{2}t^{-1} + f^{3}t^{-2} - 2ft^{-1} + t^{-1}.$$





Let $s = t^{-1}$; the K-cohomology of the generating variety is then

$$K'V = \mathbb{Z}[f,s]/(f+f^2s-1+fs,1+f^3s^2-2fs+f^2s^2+f^3s^3-2fs^2+s^2).$$

(Note that the description of $K^{\cdot}V$ in Clarke [7] omits certain relations, as the ring given there is not finitely generated.) An integral basis for $K^{\cdot}V$ is then $\{1, s, s^2, f, fs, f^2\}$. These representations of U(2) are irreducible except for f^2 which splits as $(f^2 - t) + t$.

Consider the diagram of weights in Figure 9-2. The solid lines are the Weyl walls, the dotted lines describe the set of singular weights, and the seven highest weights h_i under consideration, namely $\{0, s, 2s, f, f + s, 2f, t\}$, are circled. Note that L = t and as such, for k > 0, the weight $kL+h_i$ is either singular or is already in the fundamental Weyl chamber. The basis for K.V is of course $\{D1, Ds, D(s^2), Df, D(fs), D(f^2)\}$ and we are interested in the differences $\tau_k(Dw) - \tau_0(Dw)$. Letting $\Gamma_{(n,m)}$ denote the dimension of the irreducible representation of G_2 with highest weight na + mb, the six differences are respectively

$$\begin{split} &\Gamma_{(0,k)} - \Gamma_{(0,0)} \\ &\Gamma_{(0,k-1)} - 0 \\ &\Gamma_{(0,k-2)} - 0 \\ &\Gamma_{(1,k)} - \Gamma_{(1,0)} \\ &\Gamma_{(1,k-1)} - 0 \\ &\Gamma_{(2,k)} + \Gamma_{(0,k+1)} - \Gamma_{(2,0)} - \Gamma_{(0,1)}. \end{split}$$

Applying the Weyl character formula, we arrive at the cyclic order

$$c(G_2, k) = \gcd\{k(422 + 585k + 400k^2 + 135k^3 + 18k^4)/120, k(2 + 15k + 40k^2 + 45k^3 + 18k^4)/120, k(2 - 15k + 40k^2 - 45k^3 + 18k^4)/120, k(601 + 660k + 350k^2 + 90k^3 + 9k^4)/30, k(16 + 60k + 80k^2 + 45k^3 + 9k^4)/30, k(2867 + 2550k + 1090k^2 + 225k^3 + 18k^4)/30\}.$$

Indeed, this agrees with the result from section 9.2.

Chapter 10

Twisted $Spin^c$ Bordism and the Twisted Index

The ordinary K-homology of a space X is entirely determined by the $Spin^c$ bordism of X; see [25]. This suggests that much of the structure in twisted K-homology ought to be visible in twisted $Spin^c$ bordism. In section 8 we saw that the cyclic order of the twisted K-homology of a group G is determined by a collection of relations of the form $\tau_k(x) - \tau_0(x) = 0$, where τ_j is the j-twisted map from $K.\Omega G$ to K.*. When the class $x \in K.\Omega G$ is represented as the image of the fundamental class of a $Spin^c$ manifold M, there is a natural $Spin^c$ manifold M(j) such that the fundamental class $[M(j)] \in MSpin^c *$ maps via the index to the element $\tau_j(x) \in K.*$. Moreover, there is an explicitly identifiable twisted $Spin^c$ nullbordism over G of M(k) - M(0). In short, the relations determining the cyclic order of twisted K-homology have realizations in twisted $Spin^c$ bordism. The construction of these nullbordisms is the focus of sections 10.1 and 10.2. Section 10.3 discusses the possibility of representing the exterior generators of the twisted K-homology of G by twisted $Spin^c$ manifolds.

10.1 A Cocycle Model for Twisted Spin^c Bordism

In order to describe twisted $Spin^c$ manifolds explicitly, we need a more geometric, less homotopy-theoretic description of twisted $Spin^c$ structures; in particular we present a cocycle model for twisted $Spin^c$ bordism. This model is presumably well known and in any case takes cues from the Hopkins-Singer philosophy of differential functions [26].

Recall that $Spin^c$ is the total space of a U(1)-principal bundle over SO. Correspondingly there is a principal bundle $BU(1) \to BSpin^c \to BSO$ which is classified by $\beta w_2 : BSO \to BBU(1)$, the integral Bockstein of the second Stiefel-Whitney class. A $Spin^c$ structure on an oriented manifold M is a lift to $BSpin^c$ of the classifying map $\nu : M \to BSO$ of the (stable) normal bundle of M. Such a lift is determined by a nullhomotopy of the composite $\beta w_2(\nu) : M \to BSO \to BBU(1)$. Specifying such a nullhomotopy is equivalent to choosing a 2-cochain c on M such that the coboundary of c is $\beta w_2(\nu(M))$; (note that we have chosen once and for all a 3-cocycle g representing the generator of $H^3(BBU(1);\mathbb{Z})$, and the condition on the cochain c is that $\delta c = \nu^*((\beta w_2)^*(g))$). Ordinary $Spin^c$ bordism of X is therefore equivalent to bordism of oriented manifolds M over X equipped with a 2-cochain c on M such that

$$\delta c = \beta w_2(\nu(M)).$$

The model for twisted $Spin^c$ bordism is similar. We first recall the homotopytheoretic definition of twisted $Spin^c$ bordism from section 7.1. Given a twisting map $\tau : X \to K(\mathbb{Z}, 3)$, we have a $K(\mathbb{Z}, 2)$ -principal bundle P on X and so an associated $BSpin^c$ bundle $Q = P \times_{K(\mathbb{Z}, 2)} BSpin^c$. More particularly we have a series of bundles

$$Q_n = P \times_{K(\mathbb{Z},2)} BSpin^c(n)$$

and universal vector bundles

$$UQ_n = (P \times_{K(\mathbb{Z},2)} ESpin^c(n)) \times_{Spin^c(n)} \mathbb{R}^n.$$

The corresponding Thom spectrum

$$Th(UQ) = P_+ \wedge_{K(\mathbb{Z},2)_+} MSpin^{\circ}$$

has as its homotopy groups the twisted $Spin^c$ bordism groups of X. The twisted index map to twisted K-homology is induced by the map id \wedge ind : $P_+ \wedge_{K(\mathbb{Z},2)_+} MSpin^c \rightarrow P_+ \wedge_{K(\mathbb{Z},2)_+} K$.

The principal bundle P and the associated $BSpin^{c}$ bundle Q are defined by the pullbacks



On the other hand BSO is precisely the quotient $* \times_{K(\mathbb{Z},2)} BSpin^c$, and the diagram

$$\begin{array}{c} Q \longrightarrow BSO \\ \downarrow \qquad \qquad \downarrow^{\beta w_2} \\ X \longrightarrow K(\mathbb{Z}, 3) \end{array}$$

is therefore a homotopy pullback. Twisted $Spin^c$ bordism is the homotopy of Th(UQ); a map from a sphere into Th(UQ) transverse to the zero section Q determines a manifold M equipped with a map $M \to Q$. This map $M \to Q$ specifies maps $i: M \to X$ and $\nu: M \to BSO$ (classifying the normal bundle of M) and a chosen homotopy between τi and $\beta w_2 \nu$. The choice of this homotopy is equivalent to the choice of a 2-cochain c with coboundary equal to the difference $\nu^*((\beta w_2)^*g) - i^*(\tau^*g)$, where g is as before a 3-cocycle representing the generator of the third cohomology of $K(\mathbb{Z}, 3)$. In summary, the τ -twisted $Spin^c$ bordism of X is bordism of oriented manifolds M equipped with a map $i: M \to X$ and a 2-cochain c such that

$$\delta c = \beta w_2(\nu(M)) - i^*(\tau),$$

where $\nu(M)$ is the stable normal bundle of M.

10.2 Twisted Nullbordism and the Geometry of the Cyclic Order

In section 8 we saw that the cyclic order of the twisted K-homology of G is the greatest common divisor of the collection of differences $\{\tau_k(x_i) - \tau_0(x_i)\}$, where $\{x_i\}$ is a set of algebra generators for $K.\Omega G$ and τ_j denotes the *j*-twisted map from $K.\Omega G$ to K.*. Frequently, these generators $\{x_i\}$ can be described as the images of the fundamental classes of $Spin^c$ manifolds M_i ; (for example, we gave such a description for SU(n+1), Sp(n), and G_2 in sections 9.2 and 9.3). In this case the manifolds M_i admit modified $Spin^c$ structures $M_i(j)$ and the index of $M_i(j)$ is $\tau_j(x_i) \in K.*$. Moreover, there is a twisted $Spin^c$ structure (over G) on $M_i \times I$ cobounding the difference $M_i(k) - M_i(0)$; the relations $\tau_k(x_i) - \tau_0(x_i) = 0$ determining the cyclic order of twisted K-homology therefore have realizations in twisted $Spin^c$ bordism.

Before constructing these twisted $Spin^c$ bordisms, we recall that a $Spin^c$ structure can be altered by a line bundle and we discuss how this alteration affects the pushforward of the fundamental class. A twisted $Spin^c$ manifold is, as before, an oriented manifold M together with a 2-cochain c such that $\delta c = \beta w_2(\nu(M)) - i^*(\tau)$. In the examples we consider, the underlying manifold M is almost complex and so has a canonical ordinary $Spin^c$ structure; in particular M comes equipped with a 2-cochain b such that $\delta b = \beta w_2(\nu(M))$. A twisted structure on M is then given by a choice of 2-cochain d such that $\delta d = -i^*(\tau)$. If the twisting class τ is zero on M, then the 'twisted' $Spin^c$ structure corresponding to the cochain b+d is of course ordinary, but it nevertheless differs from the $Spin^c$ structure determined by the original cochain b. We denote by M(d) this modification of the canonical $Spin^c$ structure on M by the 2-cocycle d; we also refer to the alteration as a modification by the corresponding line bundle L(d). Let $\pi : M \to *$ be the projection to a point; the pushforward in K-theory depends on the $Spin^c$ structure on M as follows:

$$\pi_!^{(M(d))}(1) = \pi_!^{(M)}(L(d)).$$

This relation follows more or less immediately from the fact that the Thom class defined by the $Spin^c$ structure on M(d) is L(d) tensor the Thom class defined by the structure on M; see [29]. In terms of our Rothenberg-Steenrod spectral sequence approach to twisted K-homology, this tells us that the twisted image $\tau_k([M]) =$ $\langle L(k), [M] \rangle$ of the fundamental class [M] is reinterpretable as the ordinary image $\tau_0([M(k)]) = \langle 1, [M(k)] \rangle$ of the fundamental class [M(k)].

We present twisted $Spin^c$ bordisms realizing the relations in the twisted Khomology of SU(n + 1). In section 9.2 we saw that the fundamental classes $\{[\mathbb{CP}^i]\}$ are algebra generators for $K.\Omega SU(n + 1)$, and the relations determining the cyclic order of $K^{\tau}SU(n + 1)$ are

 $\sim \sim \sim \sim \sim$

$$0 = \langle L^k, [\mathbb{C}\mathrm{P}^i] \rangle - 1 = \langle L^k, [\mathbb{C}\mathrm{P}^i] \rangle - \langle 1, [\mathbb{C}\mathrm{P}^i] \rangle$$

(These classes take values in the twisted K-theory of SU(n+1) via the inclusion $\mathbb{CP}^i \to \Sigma \mathbb{CP}^i \to \Sigma \mathbb{CP}^n \to SU(n+1)$, which is of course nullhomotopic.) By the above remarks we can rewrite the relation as

$$0 = \langle 1, [\mathbb{C}\mathrm{P}^{i}(k)] \rangle - \langle 1, [\mathbb{C}\mathrm{P}^{i}] \rangle = \langle 1, [\mathbb{C}\mathrm{P}^{i}(k) - \mathbb{C}\mathrm{P}^{i}] \rangle.$$

If we can produce a nullbordism of $\mathbb{CP}^i(k) - \mathbb{CP}^i$, we will have pulled the given relation back to twisted $Spin^c$ bordism.

Write $\Sigma \mathbb{C}P^i = ([-2,2] \times \mathbb{C}P^i)/(\{-2,2\} \times \mathbb{C}P^i \cup [-2,2] \times *)$ and consider the inclusion $i : \Sigma \mathbb{C}P^i \to \Sigma \mathbb{C}P^n \to SU(n+1)$. Choose a 3-cocycle representing the twisting τ on SU(n+1) such that $i^*(\tau)$ is k times the cocycle locally Poincare dual to the submanifold $\mathbb{C}P^{i-1} \times \{0\}$ in $\Sigma \mathbb{C}P^i$. The product $\mathbb{C}P^i \times [-1,1]$ has a canonical $Spin^c$ structure coming from the complex structure of $\mathbb{C}P^i$. There is a twisted structure on $\mathbb{C}P^i \times [-1,1]$ defined by the 2-cochain d that is k times the cochain locally Poincare dual to the submanifold $\mathbb{C}P^{i-1} \times [-1,0]$; denote this twisted structure by $(\mathbb{C}P^i \times [-1,1])(k|0)$. The coboundary of d is precisely $-i^*(\tau)$. Moreover, the cochain d

restricts to k times the generator of $H^2(\mathbb{CP}^i \times \{-1\})$ and to zero on $\mathbb{CP}^i \times \{1\}$. The difference $\mathbb{CP}^i(k) - \mathbb{CP}^i$ is therefore zero in $MSpin^{c,\tau}SU(n+1)$, as desired. Notice that the same argument shows that $\mathbb{CP}^i(l+k) - \mathbb{CP}^i(l)$ is null for any l, which implies that $\binom{l+k+i}{i} - \binom{l+i}{i}$ is zero in $K^{\tau}SU(n+1)$. In fact, for any sequence of integers $\{l_i\}$, $1 \leq i \leq n$, the gcd of the set $\{\binom{l_i+k+i}{i} - \binom{l_i+i}{i}\}$ is again the cyclic order of $K^{\tau}SU(n+1)$.

Whenever algebra generators of $K.\Omega G$ are represented as the fundamental classes of $Spin^c$ manifolds, the same argument produces nullbordisms in $MSpin^{c,\tau}G$ realizing the cyclic order of $K^{\tau}G$; we forgo details. Note though that in general the twisting cochain d will no longer be locally Poincare dual to a submanifold but merely to an appropriate singular chain.

10.3 Representing the Exterior Generators of Twisted K-Homology

We would like to represent the algebra generators of $K^{\tau}G$ as the fundamental classes of twisted $Spin^{c}$ manifolds over G. Here we merely suggest an approach for further investigation, taking clues from the structure of the Rothenberg-Steenrod spectral sequence; in the process we produce a candidate representative for the generator of $K_{1}^{\tau}SU(3)$. Finding representatives in general will require a more thorough investigation of $MSpin^{c,\tau}G$ and of the associated map to $K^{\tau}G$.

The structure of the ordinary $Spin^c$ bordism group is governed by $Spin^c$ characteristic numbers; we briefly recall how to compute these invariants. In section 10.1 we considered the principal bundle $BU(1) \rightarrow BSpin^c \rightarrow BSO$ classified by βw_2 : $BSO \rightarrow BBU(1)$. There is another principal bundle $B\mathbb{Z}/2 \rightarrow BSpin^c \rightarrow BSO \times$ BU(1) classified by $(w_2 \times r)$: $BSO \times BU(1) \rightarrow BB\mathbb{Z}/2$, where r is the nontrivial map $BU(1) \rightarrow BB\mathbb{Z}/2$. This latter bundle is usually more convenient for computations of $Spin^c$ characteristic classes. The relationship between the two bundles is encoded in the matrix

Indeed this diagram shows that the total spaces of the two fibrations are the same.

Following Anderson, Brown, and Peterson [2], Stong [45] showed that a $Spin^c$ manifold M is zero in $Spin^c$ bordism if and only if all of its rational and mod 2 characteristic numbers vanish. The map

$$(\pi \times \lambda)^* : H^*(BSO \times BU(1); \mathbb{Q}) \to H^*(BSpin^c; \mathbb{Q})$$

is an isomorphism and

$$(\pi \times \lambda)^* : H^*(BSO \times BU(1); \mathbb{Z}/2) \to H^*(BSpin^c; \mathbb{Z}/2)$$

is an epimorphism. In particular a 2*n*-dimensional $Spin^c$ manifold M is nullbordant if all the characteristic classes of the underlying oriented manifold vanish and the single $Spin^c$ characteristic number $\langle \lambda(M)^n, [M]_H \rangle$ is zero. The characteristic class λ depends on the $Spin^c$ structure on M as follows. Let M(d) denote as in the last section the modification of the $Spin^c$ structure on M by the line bundle or 2-cocycle d. The class $\lambda(M(d))$ is then $\lambda(M) + 2d$, as is easily checked by noting that the composite $BU(1) \to BSpin^c \xrightarrow{\lambda} BU(1)$ is multiplication by 2.

 $\sim \sim \sim \sim \sim$

We produce a candidate twisted $Spin^c$ representative for the exterior generator of $K^{\tau}SU(3)$ by investigating the corresponding class in the E^2 term of the Rothenberg-

Steenrod spectral sequence. For simplicity we assume the twisting class k is odd; the even case is entirely analogous.

In section 8.1 we saw that the generator of $K_1^{\tau}SU(3)$ is represented at the E^2 term of the Rothenberg-Steenrod spectral sequence by $x_2 - \frac{k+3}{2}x_1$; here x_2 and x_1 are elements of $\operatorname{Tor}_1^{K,\Omega SU(3)}(\mathbb{Z},\mathbb{Z}_{\tau})$, therefore of the E^1 term of the spectral sequence, and their differentials are given by

$$d^{1}x_{2} = \langle 1, [\mathbb{CP}^{2}(k)] \rangle - 1 = \langle 1, [\mathbb{CP}^{2}(k) - \mathbb{CP}^{2}(0)] \rangle$$
$$d^{1}x_{1} = \langle 1, [\mathbb{CP}^{1}(k)] \rangle - 1 = \langle 1, [\mathbb{CP}^{1}(k) - \mathbb{CP}^{1}(0)] \rangle$$

In section 10.2 we found a twisted $Spin^c$ bordism $X_2 = (\mathbb{CP}^2 \times I)(k|0)$ whose boundary has index

$$\operatorname{ind}(\partial X_2) = d^1 x_2.$$

Because of this index property, we consider X_2 a geometric representative of the algebraic class x_2 . Note that the bordism X_2 is over $\Sigma \mathbb{CP}^2$ and therefore over SU(3).

Similarly, we have a bordism $\widetilde{X_1} = (\mathbb{CP}^1 \times I)(k|0)$ whose boundary has index ind $(\partial \widetilde{X_1}) = d^1x_1$. Given our selection of X_2 , the manifold $\widetilde{X_1}$ is a poor choice for a geometric representative of x_1 ; we would like to have a five-dimensional bordism X_1 , still living over $\Sigma \mathbb{CP}^2$, with the same index property as $\widetilde{X_1}$. A natural choice for the underlying oriented bordism is $P(\nu + 1) \times I$, where $P(\nu + 1)$ is the projectivization of the sum of a trivial bundle and the normal bundle of \mathbb{CP}^1 in \mathbb{CP}^2 ; this projectivization is a resolution of the Thom space of the normal bundle and as such the bordism maps to $\mathbb{CP}^2 \times I \subset \Sigma \mathbb{CP}^2$. There is moreover a twisted $Spin^c$ structure on this bordism, denoted $X_1 = (P(\nu + 1) \times I)(k|0)$ and produced as in section 10.2, such that

$$\operatorname{ind}(\partial X_1) = d^1 x_1.$$

The linear combination $C = X_2 - \frac{k+3}{2}X_1$ wants to be an element of $MSpin^{c,\tau}\Sigma\mathbb{CP}^2$ mapping to the exterior generator of $K^{\tau}SU(3)$. The trouble of course is that C is not a closed manifold and so does not properly represent an element of $MSpin^{c,\tau}\Sigma\mathbb{CP}^2$. Note though that the map $\partial C \to \Sigma \mathbb{CP}^2$ is nullhomotopic by a nullhomotopy on which the twisting class is zero. Suppose there is a nullbordism W of ∂C in $MSpin^c*$; then the union $W \cup_{\partial C} C$ is a closed twisted $Spin^c$ manifold over $\Sigma \mathbb{CP}^2$, as desired.

The boundary of C is

$$\partial C = (\mathbb{C}P^2(k) - \mathbb{C}P^2) - \frac{k+3}{2}(P(\nu+1)(k) - P(\nu+1)).$$

All the SO-characteristic numbers of ∂C certainly vanish. The cohomology ring of $P(\nu + 1)$ is $H'(P(\nu + 1)) = \mathbb{Z}[y, x]/(y^2, x^2 + yx)$, where y is the first Chern class of the tautological bundle on the base \mathbb{CP}^1 and x is the first Chern class of the fibrewise tautological bundle on the total space. The tangential $Spin^c$ characteristic class of $P(\nu + 1)(k)$ is

$$\lambda(T(P(\nu+1)(k))) = \lambda(T_{\text{horiz}}) + \lambda(T_{\text{vert}}) = -(2+2k)y - 2x,$$

and the associated characteristic number is

$$\langle \lambda(T(P(\nu+1)(k)))^2, [P(\nu+1)]_H \rangle = 8k+4.$$

Similarly the characteristic number for $\mathbb{CP}^2(k)$ is

$$\langle \lambda(T(\mathbb{CP}^2(k)))^2, [\mathbb{CP}^2]_H \rangle = 4k^2 + 12k + 9.$$

The vanishing of the $Spin^c$ characteristic number for ∂C follows:

$$\langle \lambda(T(\partial C))^2, [\partial C]_H] \rangle = 4k^2 + 12k + 9 - 9 - \frac{k+3}{2}(8k+4-4) = 0.$$

Picking any $Spin^c$ nullbordism W of ∂C , the five-dimensional twisted $Spin^c$ manifold $W \cup_{\partial C} C$ should represent the generator of $K_1^{\tau}SU(3)$.

Bibliography

- J. F. Adams. Primitive elements in the K-theory of BSU. Quart. J. Math. Oxford Ser. (2), 27(106):253-262, 1976.
- [2] D. W. Anderson, E. H. Brown, Jr., and F. P. Peterson. The structure of the Spin cobordism ring. Ann. of Math. (2), 86:271–298, 1967.
- [3] Michael Atiyah. K-theory past and present. In Sitzungsberichte der Berliner Mathematischen Gesellschaft, pages 411–417. Berliner Math. Gesellschaft, Berlin, 2001.
- [4] J. E. Bergner. Three models for the homotopy theory of homotopy theories. Preprint (2005), math.AT/0504334.
- [5] Raoul Bott. Homogeneous vector bundles. Ann. of Math. (2), 66:203-248, 1957.
- [6] Raoul Bott. The space of loops on a Lie group. Michigan Math. J., 5:35-61, 1958.
- [7] Francis Clarke. On the K-theory of the loop space of a Lie group. Proc. Cambridge Philos. Soc., 76:1–20, 1974.
- [8] R. L. Cohen, J. D. S. Jones, and G. B. Segal. Floer's infinite-dimensional Morse theory and homotopy theory. In *The Floer memorial volume*, volume 133 of *Progr. Math.*, pages 297–325. Birkhäuser, Basel, 1995.
- [9] C. L. Douglas and M. J. Hopkins. Symplectic Floer homotopy theory. Work in progress.

- [10] C. L. Douglas and C. Manolescu. Seiberg-Witten-Floer homotopy theory. Work in progress.
- [11] P. W. Duckworth. The K-theory Pontrjagin rings for the loop spaces on the exceptional Lie groups. Quart. J. Math. Oxford Ser. (2), 35(139):253-262, 1984.
- [12] Gerald Dunn. Lax operad actions and coherence for monoidal N-categories, A_{∞} rings and modules. Theory Appl. Categ., 3:No. 4, 50–84 (electronic), 1997.
- [13] W. G. Dwyer, P. S. Hirschhorn, and D. M. Kan. Model categories and more general abstract homotopy theory. Preprint.
- [14] W. G. Dwyer and D. M. Kan. Simplicial localizations of categories. J. Pure Appl. Algebra, 17(3):267–284, 1980.
- [15] W. G. Dwyer and D. M. Kan. Calculating simplicial localizations. J. Pure Appl. Algebra, 18(1):17–35, 1980.
- [16] W. G. Dwyer and D. M. Kan. A classification theorem for diagrams of simplicial sets. *Topology*, 23(2):139–155, 1984.
- [17] A. D. Elmendorf. The Grassmannian geometry of spectra. J. Pure Appl. Algebra, 54(1):37–94, 1988.
- [18] A. D. Elmendorf and M. A. Mandell. Rings, modules, and algebras in infinite loop space theory. Preprint (2004), math.KT/0403403.
- [19] D. S. Freed, M. J. Hopkins, and C. Teleman. Twisted K-theory and loop group representations. Preprint (2003), math.AT/0312155.
- [20] William Fulton and Joe Harris. Representation theory, volume 129 of Graduate Texts in Mathematics. Springer-Verlag, New York, 1991. A first course, Readings in Mathematics.
- [21] M. Furuta. Prespectrum with parametrized universe (draft). Preprint (2004).

- [22] Philip S. Hirschhorn. Model categories and their localizations, volume 99 of Mathematical Surveys and Monographs. American Mathematical Society, Providence, RI, 2003.
- [23] A. Hirschowitz and C. Simpson. Descente pour les n-champs. Preprint (1998), math.AG/9807049.
- [24] M. J. Hopkins. Thesis, Northwestern University, 1984.
- [25] Michael J. Hopkins and Mark A. Hovey. Spin cobordism determines real Ktheory. Math. Z., 210(2):181–196, 1992.
- [26] Hopkins M. J. and I. M. Singer. Quadratic functions in geometry, topology, and M-theory. Preprint (2002), math.AT/0211216.
- [27] Nicolaas H. Kuiper. The homotopy type of the unitary group of Hilbert space. Topology, 3:19–30, 1965.
- [28] Miguel L. Laplaza. Coherence for distributivity. In Coherence in categories, pages 29–65. Lecture Notes in Math., Vol. 281. Springer, Berlin, 1972.
- [29] H. Blaine Lawson, Jr. and Marie-Louise Michelsohn. Spin geometry, volume 38 of Princeton Mathematical Series. Princeton University Press, Princeton, NJ, 1989.
- [30] L. G. Lewis, Jr., J. P. May, M. Steinberger, and J. E. McClure. Equivariant stable homotopy theory, volume 1213 of Lecture Notes in Mathematics. Springer-Verlag, Berlin, 1986.
- [31] J. Lurie. On ∞ -topoi. Preprint (2003), math.CT/0306109.
- [32] J. Lurie. On ∞ -topoi (revised version). Preprint (2005).
- [33] Mark Mahowald. Ring spectra which are Thom complexes. Duke Math. J., 46(3):549-559, 1979.

- [34] M. A. Mandell, J. P. May, S. Schwede, and B. Shipley. Model categories of diagram spectra. Proc. London Math. Soc. (3), 82(2):441-512, 2001.
- [35] V. Mathai, R. B. Melrose, and I. M. Singer. The index of projective families of elliptic operators. Preprint (2002), math.DG/0206002.
- [36] J. P. May. E_∞ ring spaces and E_∞ ring spectra. Springer-Verlag, Berlin, 1977.
 With contributions by Frank Quinn, Nigel Ray, and Jørgen Tornehave, Lecture Notes in Mathematics, Vol. 577.
- [37] J. P. May and J. Sigurdsson. Parametrized Homotopy Theory. 2005.
- [38] Jouko Mickelsson. Gerbes, (twisted) K-theory, and the supersymmetric WZW model. In Infinite dimensional groups and manifolds, volume 5 of IRMA Lect. Math. Theor. Phys., pages 93–107. de Gruyter, Berlin, 2004.
- [39] S. A. Mitchell. The Bott filtration of a loop group. In Algebraic topology, Barcelona, 1986, volume 1298 of Lecture Notes in Math., pages 215-226.
 Springer, Berlin, 1987.
- [40] I. Moerdijk. Introduction to the language of stacks and gerbes. Preprint (2002), math.AT/0212266.
- [41] Andrew Pressley and Graeme Segal. Loop groups. Oxford Mathematical Monographs. The Clarendon Press Oxford University Press, New York, 1986. Oxford Science Publications.
- [42] Charles Rezk. A model for the homotopy theory of homotopy theory. Trans.
 Amer. Math. Soc., 353(3):973-1007 (electronic), 2001.
- [43] Graeme Segal. Classifying spaces and spectral sequences. Inst. Hautes Études Sci. Publ. Math., (34):105-112, 1968.
- [44] Graeme Segal and George Wilson. Loop groups and equations of KdV type. Inst. Hautes Études Sci. Publ. Math., (61):5-65, 1985.

- [45] Robert E. Stong. Notes on cobordism theory. Mathematical notes. Princeton University Press, Princeton, N.J., 1968.
- [46] John Tate. Homology of Noetherian rings and local rings. Illinois J. Math., 1:14-27, 1957.